

# WestminsterResearch

http://www.westminster.ac.uk/westminsterresearch

A Qualitative Study of UK Community Health Practitioners' Perceptions of their Personal and Professional Development after Training for and Practicing as Health & Social Care Innovators Fanning, A.

This is an electronic version of a PhD thesis awarded by the University of Westminster. © Mrs Agnes Fanning, 2017.

The WestminsterResearch online digital archive at the University of Westminster aims to make the research output of the University available to a wider audience. Copyright and Moral Rights remain with the authors and/or copyright owners.

Whilst further distribution of specific materials from within this archive is forbidden, you may freely distribute the URL of WestminsterResearch: (<a href="http://westminsterresearch.wmin.ac.uk/">(http://westminsterresearch.wmin.ac.uk/</a>).

In case of abuse or copyright appearing without permission e-mail repository@westminster.ac.uk

A Qualitative Study of UK Community Health Practitioners' Perceptions				
of their Personal and Professional Development after Training for and				
Practicing as Health & Social Care Innovators				
Agnes Fanning				
A thesis submitted in partial fulfilment of the requirements of the University of Westminster for the Professional Doctorate Award in Health & Social Science				
September 2017				

#### **Abstract**

This qualitative study explores the personal and professional journey of a group of healthcare professionals who undertook higher education (HE) or training in social innovation and put that learning into practice. Social Innovation is a relatively new concept in the field of health, and research is, in the main, confined to the structural and organisational aspects of innovation. Studies looking at the personal development of the innovators are rare, and no study has researched first-hand the experiences of new healthcare innovators as they learn how to set up and manage their own projects. Twenty-six community health professionals were interviewed. The participants were selected because they had either attended a HE programme or independent training on social innovation and were interested in setting up their own innovation in their clinical practice setting.

Individual interviews were achieved using Skype which proved to be an effective data collection method and allowed for a geographically dispersed sample. Thematic Analysis allowed several key themes to emerge from the data: the importance of personal resilience; increase in confidence; how levels of self- efficacy played a key role in their success; learning to shift from working in glorious isolation to seeking help from influential others. Improved technical skills and becoming better organised were also powerful factors. However, one finding proved pivotal to their success - finding themselves. The majority talked of discovering the 'real me' as a result of their learning, mixing with likeminded others and the first-hand experience of the struggle of developing a project, often in the face of opposition. For many, the positive changes transferred to their personal lives.

The findings suggest that many community health professionals have an inner drive to improve their clinical practice, but do not always know the best way to do this without formalised help. The educational input enhanced their learning and also impacted on their personal development enabling them to proceed with their innovations.

These findings are supported by research in the broader field of industry indicating that, whatever the context, there is a commonality of spirit, an ability to persevere and overcome adversity among innovators. These findings are therefore generalisable to others contemplating innovative projects in health and social care settings. In addition, the health and social professional curricula will benefit from including the subject of innovation within their educational programmes and subsequently staff and managers who work with innovative practitioners will also benefit from working with innovative professionals.

# **Table of Contents**

Acknowledgements	7
Communications Arising	8
Author's Declaration	9
Definitions	10
Chapter 1 Introduction	11
1.1 Background	
1.2 A Brief History of Social Innovation	
1.3. Social Enterprise	
1.4 An outline of the thesis	
Chapter 2 Literature Review	
2.1 The global context of social innovation	
2.2 The impact of Social Innovation	
2.3 History of Innovation in the NHS	
2.4 Direction of Innovation in the NHS	
2.5 The future of innovation within healthcare	
2.6 Innovation in Care Closer to Home	
2.7 Intrinsic and external factors in relation to innovation	
2.8 Setting up an innovative project	
2.9 From theory to developing a skills base - Enablers & Barriers	
2.10 Skills for successful innovation	
2.11 Educational Support for Innovators	
2.12 Diffusion of Innovation	
2.13 Adoption of Innovation	
2.14 Innovation and change management	
2.15 The experience of social innovators	
2.16 The gap between policy, practice and staff training	
2.17 Community Healthcare Innovation Education/Training in the UK	
2.18 A gap in the literature	
2.19 The Aim:	43
Chapter 3 Methods	45
3.1 A qualitative research approach	45
3.2 Other research approaches considered but rejected	46
3.3 Data Collection Methods	
3.4 Sampling and Recruitment	48
3.4.1 Purposive sampling	49
3.4.2 Opportunistic sampling	49
3.5 Recruitment	51
3.6 Data collection	53
3.6.1 Interviews	54
3.7 Focus Groups – difficulties encountered	55
3.8 The move to Skype interviewing	56

3.9 Authenticity	of response	57
3.10 Face to face	interviews	59
3.11 Data Analys	sis	59
3.12 Data manag	gement	60
3.13. Transcribing	g	61
3.14 Coding		61
3.15 Patterns & T	Themes	63
3.16 Themes and	d sub-themes	63
3.17 Reflexivity a	and the Researcher's position	64
3.18 Ethical Issue	es	65
3.19 Confidential	lity	66
3.20 Informed co	onsent	66
•	urance	
	dity and reliability of the data	
3.22 Summary		69
Chapter 4	Findings part one	70
<del>-</del>	0 P	
4.1.1 Four prin	ncipal themes emerged from the data:	71
	etter is possible - a perceived need to raise standards above and b	
		-
4.2 .1 A perceiv	ved need to save time and resources	72
4.2.2 Improving	g clinical outcomes	74
4.2.3 Poor com	nmunication of risk	75
4.2.4 Lack of Sp	pecialist Services	76
4.2.5 Lack of in	magination from organisations to understand the need for change.	77
4.2.6 Maintaini	ing excellence in clinical practice	78
4.3 Theme 2 -Swi	imming against the stream - the challenges faced when setting o	ut to improve
clinical practice a	and patient experience	79
4.3.1 Organisat	tional lack of recognising existing problems	79
4.3.2 Lack of vi	ision – to try for better	81
4.3.3 Frustration	on – this is how we do things around here	83
4.3.4 Self- Dou	bt - do I really understand the problem?	84
4.4 Theme 3 - Car	n't do it alone - recognition of the need for further training and s	upport86
4.4.1 Identifyin	ng a lack of budgeting/financial skills	86
4.4.2 How educ	cation influenced motivation and perseverance	87
	ce in their own clinical skills	
4.4.4 Lack of co	onfidence in knowing how to make changes happen	91
4.5 Theme 4 – Th	ne real me - finding a professional home	92
4.5.1 Permission	on to think and develop the 'self'	93
	ef – beyond work	
4.6 Summary of f	findings of the first stage of the journey	96
Chapter 5	Findings part two	98
•	part two: the complex road to becoming an innovator	
•	ne Real Me in Action - Early Involvement of others	

3.2 .1 NO IIIaII	(or woman) is an Island	99
5.2.2 Seeking	support from further afield	101
5.2.3 Working	Differently	102
5.2.4 Increase	d Confidence in own professional judgement	103
5.2.5 Personal	Development	105
5.3 Summary of	the 'the real me' in action	106
5.4 Theme 6 - D	evelopment of Real Me &The Impact of Learning	106
5.4.1 Skills in f	inance, business plans, negotiating	.107
5.4.2 Working	strategically	109
_	pportunities – finding new courage	
5.4.4 Not frigh	tened of failure	110
5.5 Summary of	'the real me' & the impact of learning	111
Chapter 6	Innovation in Action	112
6.1 Summary of	the chart information	113
6.2 Four Case St	udies - two successful ones and two which did not meet their original goals	114
6.3 Case One: Page 1	atient Support Network	114
6.3.1 Setting u	p a patient networking project	115
6.4 Case Two: E	nd of Life Care - Learning Disabilities & Families	118
6.4.1 Stages of	Development for End of Life Care in Learning Disability Sector	119
6.5 Case Three: I	Development of an alcohol consumption awareness game for the over 65s	122
6.6 Case Four: Th	ne development of sexual health promotion material	124
6.7 Summary		.127
Chapter 7	Discussion	127
•	Discussionf key findings	
7.1 Discussion of		.128
7.1 Discussion of 7.2 Exploration of	key findings	.128 .128
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is	key findingsof skills required to set up an innovative project	. <b>128</b> . <b>128</b> nd
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond curren ideas for impro	key findings	.128 .128 nd
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond curren ideas for impro	key findings	.128 .128 nd
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impre 7.2.2 Can't do	key findings	.128 .128 nd :
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impre 7.2.2 Can't do involvement of	ref key findings	.128 .128 nd : : 129
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond current ideas for impre 7.2.2 Can't do involvement of 7.2.3 Change of	r key findings  of skills required to set up an innovative project  possible – a perceived need to raise standards of health care delivery above a standard	.128 .128 nd .129 .131
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impre 7.2.2 Can't do involvement of 7.2.3 Change of working differ	ref skills required to set up an innovative project	.128 .128 nd .129 .131 and .133
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond current ideas for impropression of 7.2.2 Can't do involvement of 7.2.3 Change of working different 7.3 To look at the 7.10 between the following different 7.3 To look at the 7.2 Exploration of 7.2 Exp	by skills required to set up an innovative project  possible – a perceived need to raise standards of health care delivery above a standards of health care delivery above a standards of health care delivery above and practice / Swimming against the stream – challenges faced when setting out overnent  it alone – recognition of the need for further training and support, Early fothers – No man (or woman) is an island, Seeking support  of Perspective in recognising alternative ways of approaching clinical practice a sently across the sector	.128 .128 nd .129 .131 and .133
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impro 7.2.2 Can't do involvement of 7.2.3 Change of working differ 7.3 To look at the third objective of 7.1.2 Can't do working differ 7.3 To look at the third objective of 7.2.3 Change of working differ 7.3 To look at the third objective of 7.2.3 Change of working differ 7.3 To look at the third objective of 7.3 To look at the 7.3 To look at the third objective of 7.3 To look at the 1.3 T	ref skills required to set up an innovative project	.128 .128 nd .129 .131 and .133
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impro 7.2.2 Can't do involvement of 7.2.3 Change of working differor 7.3 To look at the third objective of and skills which 7.4 The real metals.	of skills required to set up an innovative project  possible – a perceived need to raise standards of health care delivery above a standard of health ca	.128 nd .129 .131 and .133 he
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond current ideas for improper 7.2.2 Can't do involvement of 7.2.3 Change of working differ 7.3 To look at the third objective of and skills which 7.4 The real metals.	of skills required to set up an innovative project	.128 .128 .129 .131 and .133 he .138 .139
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impro 7.2.2 Can't do involvement of 7.2.3 Change of working differ 7.3 To look at the third objective of and skills which 7.4 The real me 7.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.2.2 E	of skills required to set up an innovative project	.128 nd .129 .131 and .133 he .138 .146
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impro 7.2.2 Can't do involvement of 7.2.3 Change of working differ 7.3 To look at the third objective of and skills which 7.4 The real me 7.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.2.2 E	of skills required to set up an innovative project	.128 nd .129 .131 and .133 he .138 .146
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impro 7.2.2 Can't do involvement of 7.2.3 Change of working differ 7.3 To look at the third objective of and skills which 7.4 The real me 7.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.5 Recommend 7.6 Limitations of 1.2.2 Exploration of 1.2.2 E	of skills required to set up an innovative project	.128 nd .129 .131 and .133 he .138 .143 .146
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond current ideas for improfered for involvement of 7.2.2 Can't do involvement of 7.2.3 Change of working differed for 7.3 To look at the third objective of and skills which 7.4 The real meters of 7.5 Recommend 7.6 Limitations of 7.7 Conclusion	of skills required to set up an innovative project	.128 .128 .129 .131 and .133 he .138 .143 .146 .147
7.1 Discussion of 7.2 Exploration of 7.2.1 Better is beyond currer ideas for impro 7.2.2 Can't do involvement of 7.2.3 Change of working differ 7.3 To look at the third objective of and skills which 7.4 The real meter 7.5 Recommend 7.6 Limitations of 7.7 Conclusion Chapter 8 8.1 The research	riskills required to set up an innovative project	.128 nd .129 .131 and .133 he .138 .143 .146 .147

# Student ID W1317169

8.4 Professional I		
8.5 Personal Development		152
References		152
Appendix 1	Details of Projects	180
Appendix 2	Information Sheet for Participants of the Study	183
Appendix 3	Consent Form for Participants	184
Appendix 4	Questions for participants	185
Appendix 5	Additional Questions	187
Appendix 6	Consent Form for Transcriber	188
Appendix 7	Ethics approval - Bucks New University	189
Appendix 8	Ethics Approval - Westminster University	190

# **Acknowledgements**

I would never have been able to complete my Professional Doctorate if I did not have the constant unfailing love and support of my husband John and my children Siobhan, Ciara and Daniel. Our lives have been on hold whilst I have been studying so I am looking forward to spending quality time with them all again.

I would like to thank Buckinghamshire New University for supporting me to study the Professional Doctorate and for Westminster University for accepting my application to enrol onto the programme. I cannot thank my supervisors Dr Patricia Maitland and Dr Tina Cartwright enough for all of the guidance and support they have given to me during my time at Westminster University. Both have always been supportive even when I am sure that inwardly they were despairing of me.

Thank you to all of the participants who selflessly gave so freely of their time to take part in this research. The data collected from their interviews has been truly inspirational and I cannot wait to share the findings with a wider audience. I hope that my interpretation of the data has truly captured all the great life changing work that they are all doing to deliver quality care for patients in the community.

Finally thank you to the reader for taking time to read this thesis. I hope you enjoy reading it and that you too are inspired by the innovative work that is currently happening in the community.

# Communications Arising

# Publication

Fanning, A., (2013) Social Innovation and Enterprise, chapter in *Community and Public Health Nursing (5<sup>th</sup> edition)*, Sines D., Aldridge-Bent, S., Fanning, A., Farrelly, P., Potter, K., Wright, J. Wiley, Oxford

# Author's Declaration

I declare that all the information contained in this thesis is my own work.

#### **Definitions**

**Inductive data analysis.** When the meaning of the data is generated from the raw material and theories are developed. The data is not analysis by any predetermined framework.

**Skype interview.** A face to face talk between two people using Skype computer software.

**Innovation.** An idea, object, behaviour or object that is perceived as new by its audience.

**Social Enterprise.** Social enterprise as a phrase came into being to describe commercial organizations which had as their remit social and environmental change.

**Social Innovation.** Social Innovation describes a wide range of activities across the workplace with a philosophy that encourages the joining together of individuals with an innovative concept to propel social change through transformative forces (Fanning 2013: 258 in Sines et al 2013). Social Innovation also refers to new concepts, strategies, ideas and organisations that meet social needs (Oxford Dictionary, 2011).

**Thematic analysis**. An inductive approach to qualitative data analysis that is generic so can be used in a range of studies and is not linked to a particular theoretical perspective. It is a process which first *codes* and labels the material from across all the interviews then identifies broader *themes* from the codes. The themes as the basis for the data interpretation.

**Diffusion of innovation.** A theory that seeks to explain how, why, and at what rate new ideas and technology spread

**Rate of adoption.** The relative speed with which an innovation is adopted by members of a social system. The rate of adoption is a numerical indicator of the steepness of the adoption curve for an innovation.

## Chapter 1

#### Introduction

#### 1.1 Background

This section introduces the background to the study in three key areas. It then provides an overview of the thesis: 1) The origin of the research focus – a social Innovation programme for community health practitioners. 2) A brief history of social innovation. 3) A brief history of social enterprise. This is followed by an outline of the thesis.

The interest in this subject began with the researcher developing and teaching on a Post Graduate programme on social innovation for community practitioners at a University in the South East of England. The concept of social innovation was not familiar to many of the healthcare professionals who had enrolled on the programme but they had expressed interest in the programme because they had ideas about improving practice and were unsure how to put their ideas into action. The programme was a place to start. One of the key initial attractions to the programme, they claimed, was the unique way that it was marketed. Although all students had to be a qualified community health professional, recruitment did not focus on traditional academic level entry, but rather it asked questions such as, 'If you wanted to change anything what would it be?' and 'What is the bravest thing you have ever done?' The programme attracted students from across the country.

It was delivered through action learning sets and experiential learning in two-three day blocks over a period of nine months. This enabled students to build on their knowledge and put their ideas into practice, such as setting up a new initiative or project in their workplace. Key topics covered in the programme were: Reflection, lateral thinking, personal and professional values, creativity, innovation, understanding and reading accounts, economics of business, developing a business plan, implementing a project, dealing with failure, setting up a small business/social enterprise. Established external innovators (mainly in the field of health and social care) were invited to speak to the students, demonstrating that innovation is a real concept which can make a positive difference to patients and clients.

It is important to note that this current study is not an evaluation of the programme, the researcher is not concerned with the students' view of the programme itself, but is an attempt to better understand the impact of the learning experience on the

participants and how it impacted on their behaviour when implementing their own social innovation programme. As the researcher watched individuals grow in confidence both personally and professionally during the programme, it seemed to be a worthy research goal. The researcher had also heard other practitioners speaking at conferences and read about innovative projects that encouraged further investigation into this world of innovators in the community health workforce. Hence the idea for this research came about.

## 1.2 A Brief History of Social Innovation

Social Innovation describes a wide range of activities across the workplace with a philosophy that encourages the joining together of individuals through an innovative concept to propel social change via transformative forces (Fanning 2013: 258 in Sines et al 2013). Social Innovation also refers to new concepts, strategies, ideas and organisations that meet social needs (Oxford Dictionary, 2011). Social innovation is not a new concept and has been around since the beginning of the seventeenth century, but McSherry & Douglas (2011) recognise that it is relatively new in the health field with the Department of Health investing money to promote innovation in the work place in 2009 (DH, 2009) and since then NHS policies have been promoting innovation (DH, 2010, 2011, 2012). Not only in the health field, but nationally there appears to have been a resurgence of social innovation in the second half of the twenty first century with a number of social entrepreneurs who are already waiting and ready to be acknowledged and supported (Clark, 2009). This is particularly evident in the health setting as community health services are being transformed and new ways of working will require support at senior levels to encourage new ideas and new ways of working to be implemented.

The aftermath of the French revolution saw the entry of the term social innovation into the English vocabulary and it has been used extensively in ensuing decades since then (Social Innovation 2012). In Burke's (1790:64) words, 'Innovation is revolution – and revolution is innovation', confirming that to innovate is not to reform. Outside of the health service evidence of this is seen in social innovations such as the introduction of The Open University and The Big Issue. Both of these innovations

have benefited communities by providing education in non-traditional ways and providing work and an income for the homeless in communities across the UK. It is almost certain that such innovations as the formation of the Franciscan Order in 1209 would suggest that St. Francis of Assisi would be one of the first social entrepreneurs (Bornstein, 2010) as it was an innovative concept to create an environment that met the social, as well as the spiritual needs of people. In 1952, Glidden & Powell (cited in Harris, Bennet & Ross, 2013: 5) spoke about Muriel Powell who was head of the London Teaching Hospital Matrons Association, her approach to patients was to address them as individuals and not refer to their bed number or medical condition. Today this seems a natural approach, but in 1952 this was an innovative way of carrying out holistic patient care. In more recent times, Mulgan (2012) gives further examples of successful innovation – Which Magazine, Healthline and International Alert. These ideas were the brainstorms of individuals which, with support, have grown into very successful global businesses and would come under the auspices of a social enterprise. Health professionals can learn from these innovators as skills used to promote innovation are transferrable to any situation, not just global enterprises and currently there is an emergence of social innovations, particularly in technology, being developed by health professionals (Jiwa et al, 2014, Doyle et al 2014, Fiedler et al, 2014, Sarhan, 2014). Social Innovation is about taking new ideas that meet unmet needs (Mulgan, 2007), of which there are many in community health services, but the process from the initial idea to the implementation of the concept may take some time and collaboration is required to assist in the many boundaries and challenges that may be met along the way (Simpson, 2009). These challenges may come from a variety of sources, such as colleagues, peers, managers and patients themselves. A related term is Social Enterprise (SE) which can be confused with Social Innovation (SI), but SE is different in that a social enterprise is a business that is trading for social and environmental purposes not simply its own and they put their profits back into benefitting the community, whereas Social Innovation refers to new concepts and ideas and is not necessarily linked to profits.

# 1.3. Social Enterprise

Social enterprises in the UK have been in existence since a workers' co-operative was set up in Rochdale in 1844. The idea was to provide quality food in response to what was considered exploitative factory conditions. Gradually these co-operatives developed across the country and in the 1990s the term 'social enterprise' came into being noting that many organisations had been using the power of business to bring about social and environmental change, but there was not an overarching name for this, as such 'Social Enterprise' was established (Social Enterprise, 2010). According to the Small business Survey in 2010 it was claimed that 5.7% of all small and medium-sized enterprises were social enterprises with only a minor percentage health related. The survey identified that there were 68,000 social enterprises in the UK, but Findlay (2012) from the Social Enterprise Mark Company disputed these figures stating that a number of the enterprises had not met the legal requirements to become a social enterprise.

A similar concern was voiced by Social Enterprises UK (2012) which are independent of the state, who were worried that businesses may 'hijack' the term social enterprise when they do not meet the philosophy of a social enterprise. However a further study was carried out in 2013 and suggests that at that time there were approximately 70,000 social enterprises in the UK which contributes £18.5 billion and employing up to one million people (Social Enterprise, 2013). Those businesses whose stated aim to change the world for the better through trading to tackle social problems, improve communities and people's life chances for the environment have the underpinning ethos of a social enterprise as they reinvest their profits back into the business or local community (Bornstein, 2012). So when they profit, society profits. In brief, a social enterprise is a business that is trading for social and environmental purposes not simply its own efficiency. Small social enterprises can remain connected to large organisations, such as the NHS, without actually being embedded in the chain of management within the organisation (Walsh, 2014) and evidence of this is discovered in the research.

## 1.4 An outline of the thesis

This thesis is made up of eight chapters. Following this introduction, *Chapter 2* reviews the literature on social innovation internationally, nationally and then within the National Health Service. Global innovations and innovations within the United Kingdom will be presented and the role diffusion plays in implementing innovative projects will be discussed. Since the publication of *Our Health, Our Care, Our Say (DH, 2006)* the concept of innovation has been encouraged within the NHS. Numerous small scale successful projects have been set up, but there seems little evidence of these being benchmarked across the sector. There is a tendency for them to remain as individual projects and awareness of these projects are only known in a local context. The NHS is a large institution which relies on managers to support their workforce but there is evidence that lack of support on occasions and other organisational barriers have prevented some innovative projects being implemented (Carr,2010). Staff may be confronted with challenges from a variety of sources, such as colleagues, peers, managers and patients themselves which will be discussed in chapters 2 and 7.

- 1.4.1 Chapter 3 will address the research approach and methods of the study which was carried out during the period of 2012 -2015. It is divided into six sections: the first outlines the rationale for choosing a qualitative approach for this study. The second section addresses the sampling and recruitment strategies and methods. The third discusses the data collection process, issues arising and how they were overcome. The fourth outlines the specific analytical approach and procedures undertaken, and how they are based on Braun & Clarke's (2006) guidelines. The fifth section focusses on the ethical concerns of the study and the trustworthiness of the findings. Finally, the issues around quality assurance and research governances are discussed.
- 1.4.2 Chapter 4 presents the first phase of the findings which explore the emerging themes underpinning the individuals' initial desire to step forward and make changes within their organisation for the benefit of their patients and the barriers they faced in doing so. It also identifies some of the participants' personal characteristics which, in part, help drive them forward and their early experiences of working with senior management which went on to influence their later work.

- 1.4.3 Chapter 5 presents the second part of the findings which demonstrates the personal and professional development reported by the participants as they move forward with their own projects. The change of perception regarding the participants' self-perception shaped this chapter which focussed on 'the real me'. Participants of the study reported some profound personal changes as they attempted to operationalise what they had learned on their programmes.
- 1.4.4 Chapter 6 presents an overview of all of the projects that the participants were involved in. Details of four case studies will be presented. Two of the case studies were successful projects and two of the case studies did not progress and the rationale for the success/failure of the projects will be discussed.
- 1.4.5 Chapter 7 discusses the findings and is divided into four sections. Firstly confidence and self-efficacy in the knowledge that the participants knew their specialist subject area was evident. Some had an innate confidence from the onset and the additional understanding of practicing innovatively enhanced their confidence. Others only recognised their confidence when their change in clinical practice demonstrated that they were working more effectively and patient satisfaction was increasing. As their self-belief developed it unravelled a resilience which became a distinguishable factor in the findings. The resilience discovered encouraged the participants to work differently experiencing change in the workplace, thus working differently was another key factor in the findings. Fourthly, how post qualifying education and learning relevant skills to set up an innovative project has enhanced their clinical practice will also be discussed. What additional skills did educational programmes give to participants enabling them to pursue their innovations effectively? Examining everyday discourse used in the nursing and health educational literature identified common attributes, antecedents and the consequence of post qualifying education (Gallagher, 2006). As a result of the findings in this research study the relevance of post qualifying education for health professionals is discussed. This chapter will revisit the findings and will present the recommendations, limitations of the study and finally the conclusions.
- 1.4.6 Chapter 8 is a reflective account of the researcher's personal journey carrying out this research study. Reflexivity is revisited and discussed as it is central to the theoretical approach adopted. The chapter closes by sharing the personal and

professional development of the researcher of which similar traits seen in the participants are also evident in the characteristic of the researcher.

## Chapter 2 Literature Review

This chapter will look at literature on innovation from a global perspective. It will look at the history of innovation in the NHS and proceed to review Department of Health policies in relation to innovation today. Influences of the diffusion of innovation on community health professionals and the organisations in which they work will be reviewed and the enablers and inhibitors to implementing innovation will be investigated. The bulk of the literature on Social Innovation is written in general terms about why it is a good idea and how to operationalise it, little has been written about the impact becoming an innovator has on individual practitioners. Therefore, given that the focus of this research is the personal and professional development of UK healthcare professionals some who have after undertaken a post graduate course in social innovation and others who have attended programmes where innovation is explored, the impact of continuing professional development for community health professionals will be reviewed. While not focused specifically on innovation, this continuing professional development literature offers general insights into the impact of study on mature students and has value here. Finally the contribution to original knowledge that this study offers is set out.

## 2.1 The global context of social innovation

Social Innovation is a globally accepted concept relating to the development and implementation of new ways of thinking and working within a wide range of professions in order to promote the wellbeing of society, and the global flow of knowledge, skills and ideas has been a defining feature of human progress (Shamsuzzoha et al, 2013:1). The World Economic Forum is one of its founders, aided by The Social Innovation Exchange (SIE), an international organisation formed in 2008 and located in Australia. Both these organisations have the aim of supporting and inspiring social innovation practitioners worldwide. The SIE which works with a number of social care organisations researching ideas and the practical experiences of existing social innovation projects, offers the following definition of Social Innovation:

'Innovative activities and services that are motivated by the goal of meeting a social need and that are predominately developed and diffused through organisations whose primary purposes are social' (Mulgan, 2007: 15).

Further commentary on the topic comes from The Theoretical, Empirical and Policy Foundations for Building Social Innovation in Europe (TEPSIE), (2014:10) who suggest that social innovation is an opportunity to address the challenges that face public institutions such as the NHS, as innovation is an opportunity to modernise and improve the service. They also suggest that there is emerging literature focused on learning from frugal approaches to innovation, which is about improvising solutions to problems using scarce resources, based on a high understanding of local needs.

The two way flow of ideas between emerging and more developed areas of the world encourages innovation across economies. The perception that innovation is a oneway traffic and originates from a developed part of the world is not always the case. 'Innovation blowback' or 'reverse innovation' as a concept is important to understand when reviewing innovation from a global perspective as the innovation may have originated from a developing country prior to spreading to the 'developed world' (Shamsuzzoha et al, 2013:1). One such example is when Procter & Gamble developed a cough medicine in Mexico to treat the local population. It was a success and subsequently it was introduced into Latin America and eventually the United States. There are numerous other examples such as the cardiac machine that was developed in India and is now sold to the United Kingdom where it is located in businesses across the UK. However, Harris et al (2016) suggest that care should be taken when using this term as it suggests a 'cultural arrogance' since 'reverse innovation' resonates differently with each person and is dependent on the context. It could be seen as patronising, particularly if the learning is coming from a low-income country, hence suggesting that this term be replaced with 'learning from others'.

Kaya et al (2015) attests that innovation is the underlying principle of health care services across the world, resulting in significant health improvements for patients, but this is rarely recognised as innovation. In support of Kaya's (2015) findings, literature from outside of the United Kingdom (Welch 2008, Suhonen & Paasivaara 2011) relating to innovation in nursing identifies that it is the knowledge, confidence

and belief in an idea that has enabled projects to be implemented, rather than just the good idea on its own. In addition, several global perspective research studies have also been carried out on social innovation (Goldenberg et al. 2009, Murray et al, 2010, Goldsmith, 2010, Gardener et al, 2007), and one of the key findings was the obvious drive and passion that the practitioners had for their individual innovations. The Department of Health Social Enterprise Investment Fund (DH 2009a, 2012) confirmed this too, claiming that enabling staff to develop their ideas has resulted in better motivated staff who have ownership of their idea, giving them more freedom to be creative with regard to the health needs of their population. There is also a wealth of literature from Canada and USA relating to innovation in health care: the overall findings show that the methods of communication and the culture of the organisation are essential if an innovation is to proceed successfully (Torrie et al, 2009, Dreidger et al, 2010, Lin et al, 2016). In the UK O'Leary (2011) reports the success of a training programme for care home healthcare professionals. enabling them to monitor and review at an early stage patients who are at risk of malnutrition, which prevents hospital admission. This is just one example of many evidence based projects that promote innovation in the NHS (QNI, 2014, Dean, 2011, Adams et al, 2011, Butterworth, 2013, Kendall-Raynor, 2013, Davis, 2011) but since Lord Darzi's policy on promoting innovation in the NHS (2009,) this evidence based information is continuing.

## 2.2 The impact of Social Innovation

There is an increasing amount of literature on the impact of innovation within the NHS (Trueland, 2014, Davis, 2011, Fitzgerald, 2003, Kendall-Raynor 2013, Brooks et al, 2011, Smit et al, 2013, Lennan et al, 2014, Andrews et al, 2013, Adams et al, 2006), with much of this literature based on technical innovations that are supporting patients to manage their own medical condition (DH,2011b, DH,2011f, Jiwa et al, 2013, Sarhan, 2014, Singh et al, 2010, Gardner et al, 2007, Skinner, 2010). The literature highlights that one of the major factors to be considered when judging the effectiveness of any social innovation intervention is the relative advantages of the innovation over existing practice. The advantages will not be the same for all innovations as some will emphasise the economic benefits whereas others,

particularly 'in the third world', may focus on quality of life (Goldsmith, 2010, Holland et al, 2012, Shamsuzzoha et al, 2013).

Examples of new initiatives that are aiding people living in Africa are: 1) supplying portable water pumps to assist sub-Saharan smallholder farmers to grow crops out of season, 2) a cardio-pad for people who live in rural areas and do not have access to medical services. It is a computer tablet which diagnoses heart disease. It recognises people who are at risk of heart disease or hypertension and they can be treated without having to travel up to 900 kilometres to visit a cardiac specialist, 3) growing sweet potatoes which contain levels of beta carotene six times higher than that contained in a regular sweet potato, to help fight childhood blindness. In Africa it is estimated that 28% of children have Vitamin A deficiency (Holland et al, 2012).

The literature indicates that the measures of success are locally developed and assessed. For example, a reduction in childhood blindness and lower mortality rates for heart disease as mentioned above. Overall the literature shows that as a general concept social innovation is regarded as successful and as a practice it has endured (Bessant et al, 2012, Mulgan, 2012).

#### Reflection:

It is interesting to note that so much can be learnt from the third world initiatives. There has been a tendency to suggest that it is the UK and other developed countries that are the leaders in innovation, particularly in health. I found it fascinating to learn that the UK have taken on initiatives from third world countries and the literature has proved that where there is a desperate need for change as a means of benefitting society then every avenue needs to be explored. I would like to see this philosophical attitude adopted by staff in the UK health service.

# 2.3 History of Innovation in the NHS

Innovation has been constant within the NHS since its formation in 1948 and this has been mostly in the field of research and technology. It is mainly evidenced by the introduction of new drugs, imaging systems and medical devices. However some commentators suggest that 'new software packages or internet sites may be designed to prevent innovations from disrupting the status quo' (Block, 2013:47). Although innovation in health care is sorely needed, most changes bring, at minimum, the need for revised workflow and education, or at maximum, a complete

redesign of longstanding processes (Gosselin et al, 2015). Either way, change creates the need for health professionals to unlearn old practices and procedures which often results in anxiety, uncertainty, and distress for practitioners, confirming Obholzer's (1994) thoughts on increased anxiety.

In 2010 the Department of Health carried out an evaluation of 26 social enterprises across the health and social care sector, demonstrating that social enterprise was a catalyst for delivering services in an innovative and improved way. However, the report looked at social enterprise from an organisational point of view only, as does Obholzer (1994); neither looked at the individual perspective of the innovator. Two studies that did begin to address this are Bettencourt (2010) and Gibson & Kelly (2010) who stress that innovation in practice will only happen if an organisation has a clear vision which frequently includes a rethinking of health care roles and a change in the way people are allowed to practice.

The need for innovation is indeed urgent as indicated by Bessant, Kunne & Moslen (2012) with increased concerns over the economy of the health care system in the UK, a view which is also supported by Jiwa et al (2012:5) claiming that 'healthcare systems in the early 21st century are facing a crisis'. Over the last decade UK governments have had to manage the increasing demands on the National Health Service, and have recognised that providers will need to deliver their services differently - more effectively and at less cost. Therefore, the significance of social enterprise and innovation within health care has grown in importance. Lord Darzi's NHS Next Stage Review (2009) was aimed at stimulating innovation in the NHS working with the Health Innovation Council to act as 'quardian' for innovation and to provide guidance and strategic advice (NHS, 2009). He promoted innovation across the NHS, claiming that an innovative culture cannot be led from the centre and would be best placed in localities ((NHS, 2009). Additional government funding of £220 million was divided equally across the ten strategic health authorities (localities) with the emphasis placed on health improvement and patient empowerment and engagement (NHS, 2009).

A year later the White Paper (DH,2010), which promotes innovation and supports the Big Society agenda (Govt.agency, 2010) in two ways, recommended firstly that the bureaucratic and hierarchical management model be challenged by devolving power

to local organisations in order to design the best responses to public health issues. Secondly, it suggested putting the patient at the heart of the process with clinicians, nurses and allied health professionals playing an essential mediation role in working with patients, communities, GPs and local authorities to deliver improved health and public health outcomes.

It is noteworthy, that like the international research, the literature on the NHS promotion of innovation paid little attention to the personal development and impact of new processes on individual practitioners.

The Queen's Nursing Institute who have been supporting nurse-led projects since 1990 have seen a massive expansion of projects that are improving patient outcomes within the NHS since the emphasis on innovation has increased (QNI 2010, 2012, 2014). The outcome of such nurse-led projects evidenced in the QNI impact reports (2011, 2012, 2014) underpin The NHS Institute for Innovation and the Improvement and National Innovation Centre (2011) claims, that most innovation actually comes from staff working close to the patient and not, as many people suggest, that the innovation comes from policy makers and senior physicians.

The intention of the NHS is to raise the scale of innovation in the UK and, to support this, they have developed an innovation team to act as internal expert consultants (NHS, 2015). However, the NHS policies that are promoting innovation look at this from an operational perspective, not giving attention to additional education or training that might be required to support the staff working in the NHS. Since 2006 there has been a drive within the NHS to promote social enterprise (DH, 2006) and with the publication of *Our Health, Our Care, Our Say* (DH, 2006) there is evidence of more nurse-led innovations being set up (QNI, 2014, Dean, 2011, Adams et al. 2011, Butterworth, 2013, Kendall-Raynor, 2013, O'Leary, 2011, Davis 2011). This list is not exhaustive as many small scale innovations are in place, but the findings have not yet been published. The impact that has been demonstrated in the literature to date has proved there are better outcomes for the patient and greater autonomy for the practitioners. This is positive news, and Davis' (2011:6) article demonstrated how a 'chronic illness case-management service for community nursing achieved a total saving over a nine month period of more than £400,000, with an average cost reduction of almost £1,500 per patient, as well as greater satisfaction'. O'Leary

(2011) spoke about an oral nutritional supplement prescribing review which focused on patients who were potentially at risk of malnutrition. Community staff and care home staff were trained to screen, treat, monitor and review patients at risk of malnutrition. With the implementation of this initiative it is estimated that the saving to the clinical commissioning group would be up to £200,000. Another example discussed by O'Leary (2011) was a review of the transport services for patients requiring haemodialysis, whereby patients were grouped according to their postcode and where possible their day for dialysis changed to fall in line with their postcode group, giving rise to an estimated saving of up to £164,000 for the clinical commissioning group. These examples are not to be disregarded, but what is not known about these innovative projects is what the drivers were that made these innovations succeed. These are excellent examples of good practice but it would be interesting to know whether any of the clinicians behind these innovations had received education or training in how to set up a social enterprise or develop a successful innovation. The literature does not give background educational information about these individuals, only details of the clinical area in which they worked.

## 2.4 Direction of Innovation in the NHS

As this evidence of innovation continues to grow, Sir David Nicholson, Chief Executive of NHS England, said that 'Innovation must become core business for the NHS (DH, 2012a) suggesting that NHS commissioning boards must promote innovation. This statement builds upon Lord Darzi's views put forward in the report 'High Quality Care for All' (DH, 2008) where he presented a vision for the future of the NHS with a key aspect of the report promoting an escalation of innovation in the NHS. This is followed in 2009 by the NHS Constitution (DH, 2009) which promised a commitment to innovation. This commitment involved the Health Service Management Centre (HSMC) recommending ways that innovation could be supported, which involved; factoring time for leaders and managers to attend training and to visit areas of innovative practice, providing support to build on previous experience of what does and does not work, valuing innovators through incentives, using techniques to share knowledge and to nurture individuals who are demonstrating innovation (Skinner, 2010). The DH policies generally make broad

statements about promoting innovation such as 'Increasing access to healthcare, maintaining wellness and reducing the cost of treatment' (Jiwa, 2013), but in this time of low morale within the workforce policies should be seen to be actively promoting innovation and directing staff to ways of implementing innovation into their everyday practice, which unfortunately the directives from government are not doing. Currently the NHS is relying on individuals to drive innovation forward, rather than encouraging all staff to become innovative in their ways of thinking, which is contrary to what the Department of Health is suggesting.

Another factor to consider when reviewing the Department of Health policies is that the drive towards innovation does veer towards those in higher positions such as consultants who are in a position to negotiate at a strategic level, rather than the frontline workforce of nurses and allied health professionals. *The Five Year Forward View* (DH, 2014) reaffirms this as it talks about new forms of integrated care being the blueprint of the NHS as a whole. This cannot happen from frontline staff alone; it has to become a directive from senior management and all staff should work together to implement these changes. *The Five Year Forward View* (DH, 2014) also refers to technical innovation and large-scale projects which require input from senior clinicians and technical staff and not necessarily front line staff. NHS England (2017) currently sends out monthly bulletins informing health trusts what is happening in the world of technology and how it is impacting on patients and workload. It is an excellent way of informing staff of what progress is being made in relation to patient technology, but it would be extremely helpful if there was additional information for staff on how to use the technology.

A further means of promoting innovation is the joining forces of academics and clinicians to form an Academic Health Science Network (AHSN); the aim being to develop innovation in localities. Once again, there is an expectation that new initiatives that result from the AHSN are disseminated across all levels of the workforce.

Front line staff working in a specific area, such as diabetes may be involved in the innovation as it requires a multi-disciplinary approach. Also patients who are at the interface of nurse-led care have spoken out in favour of innovative practitioners (QNI, 2014). Contrary to this view Gibson & Kelly (2010) deliberate whether it is right

to expect frontline staff to change practice, whilst balancing all of the responsibilities of their clinical roles. Disputing Gibson & Kelly's (2010) view, Lansley (2012b) promotes the notion that innovation should come from front-line health care professionals, 'putting the patient first' and challenging the top-down, target driven culture that underlies the current NHS system. The Department of Health (DH. 2009a: 43) claimed that it would 'align its entire work programme to support NHS organisations to meet the challenges ahead'. It is obvious then, that community health professionals will require knowledge and resilience to be able to deal with the challenges that they will be confronted with in attempting to implement any new initiative. This approach is also reinforced by Lansley's controversial, yet insightful view that 'the National Health Service requires healthy competition, which potentially will result in innovation' (Lansley, 2012). Research in this area is therefore required to inform managers and participants of the effectiveness of the current mode of delivery, and without further research and additional knowledge and education in becoming an innovator, innovations will become vulnerable to the vagaries of policy changes.

## Reflection:

Emphasis is on the importance of vision and it is obvious from literature that practitioners may have vision but they do not always have the relevant skills and knowledge to do anything about this. It is distressing to read that in many cases, unless a manager shares this vision innovations may not develop. The message by Nicholson in 2012 saying that innovation must be at the core of health business has not been emphasised enough to managers who are frequently either the enablers or the inhibitors of an innovation.

It has been frustrating reading the literature as there is a lot of emphasis on innovation in the health service, particularly since the publication of Our Health, Our Care, Our Say (DH, 2006) yet this message is not being recognised by the front line staff who frequently see this as just another change. There remains a gap in the literature regarding how to implement an innovative project and how to get heard. O'Leary (2011) talks about gaps in knowledge, but does not delve into any explanation how to fill this gap. I also anticipated that there would be more literature on the evaluations of projects which has not been evident. These are two areas where further research could be carried out.

## 2.5 The future of innovation within healthcare

The health policy at the time of the coalition government (2010-2015) saw major health reforms, mainly driven by the health minister Andrew Lansley whereby he

promoted innovation despite the rigidity and bureaucracy within the NHS. He wanted to remove the numerous layers of senior management and devolve the power to the general practitioners. At the same time The Darzi Report (2010) was recommending care closer to home and increased patient choice which placed community practitioners central to driving innovation forward by providing an increased service to patients in the community and also promoting health promotion and prevention of hospital admissions. It says that more innovation should come from the front-line of nursing and other allied health professionals. This would enable innovative services to be developed that 'put the patient first' and challenge the top-down, target driven culture that underlies the low morale found in so many parts of the NHS. Yet, Lansley (2012) claims that what the NHS requires is a healthy competition, claiming this will result in innovation. Lansley's views are not welcomed by all though, as the Kings Fund published a review in 2015 which claims that Lansley was forcing the NHS to reform itself, ignoring traditional hierarchical roles and traditions which in essence was a disaster waiting to happen. However, they suggest that Jeremy Hunt, the current health minister should not veer away from new reforms, but make sure any new reforms are implemented and executed correctly. Additional suggestions include placing less emphasis on regulation and more on supporting leaders and staff to improve care. In essence this is not dissimilar to Lansley's policy as both health ministers speak about bottom-up changes to the NHS with an emphasis on quality of care.

## 2.6 Innovation in Care Closer to Home

Staff working in the NHS have witnessed a political drive to deliver health care closer to home (DH, 2010) and at less cost. With the move of health care services being delivered in Primary Care, traditional practices must therefore be challenged to make way for innovative approaches to care (Sines et al 2009). Established community health care delivery is defined as an ongoing process of improving people's lives and alleviating the underlying socioeconomic conditions that contribute to poor health (WHO, 2012). Key professionals involved in promoting this care would include district nurses, health visitors, school nurses, community children's nurses, community mental health nurses, learning disability nurses and allied health professionals such as occupational therapists, speech therapists and

physiotherapists and members of these wider teams. With the decline in health care staff numbers, in particular district nurses where there has been a decrease of over 40% in the past decade (QNI, 2013), the reduction in numbers has caused widespread concern regarding the capacity to deliver quality care. In June 2013, for example, over 20% of district nursing courses in England did not have any students enrolled on their programmes (QNI, 2013). However, at the same time with the merging of organisations which deliver the care, staff are being encouraged to be creative in their practice, yet remain within the boundaries of their professionalism. Pesut (2013) suggests that if health care professionals are provided with innovative models of service delivery and resources this will assist them to think more creatively, enabling them to work more effectively (reducing costs while maintaining or improving the quality of care) within the health care system.

#### 2.7 Intrinsic and external factors in relation to innovation

Although challenging, it is thought that working with innovative models of care can be achieved, but will require a radical shake up of established ways of working and additional staff training within the NHS and wider field of social care (Lansley, 2012c, Goodwin et al, 2013). With the current changes of transforming community services and the Five Year Forward View (NHS, 2014) many community health professionals will be in a position to present new ways of working, but it will be essential to have management support to drive any new initiative forward. A systematic review carried out by Chaudoir, Dugan and Barr in 2013 suggests that effective leadership cannot be disregarded and the culture and the emphasis placed on evidence-based practice within an organisation will play a crucial role in whether innovation is prioritised within an organisation.

Overall, the underlying thrust of Social innovation is to encourage the participation of an entrepreneur to work with individuals and communities in the promotion of well-being not only *for* the community but also *with* them (Hubert, 2010), identifying new ways of working and participating with the local community for a positive social outcome. It is not meant to be a replacement for services already in place, but should be seen as supportive in introducing innovation to an already existing service that may not be reaching its full potential, but applying an idea to produce something new and useful' (Paterson et al, 2009). Murray et al (2010) however points out that a

successful social innovation project should take account of a number of changes such as organisational, financial and any relationships with stakeholders and territories. Considering this viewpoint, no solution has been provided as to how innovators can develop these additional skills in finance, communicating with stakeholders and commissioning services. In recognising a number of constructs that represent aspects of any innovation Chaudoir et al (2013) state that the social, political context must be clear and the scope and purpose of any stakeholder involvement, patient benefit and management support must be understood if the innovation is to be implemented successfully. They also mention that the innovation should be supported by policy and have the relevant funds to see the innovation through. Nevertheless, innovations may not always have these structures in place, which may be why there may be difficulty for some in getting their innovations supported by their organisation.

A key factor in support of innovation as suggested by O'Leary (2011) is to target learning where there are gaps in knowledge by staff wanting to be involved in an innovation in the clinical practice setting. However, O'Leary does seem to suggest that this is referring to clinical knowledge and not the wider skills required, such as financial acumen to set up a project. When implementing innovation in practice Smit et al (2013) stress the importance of characteristics and attitude, claiming that a positive attitude is a major attribute required, also suggesting that self-efficacy as a construct plays an important role in promoting innovation.

There also appears to be some cynicism regarding nurse innovators, as views such as Cheater's (2009) claim there is little evidence available on the impact that nurse-led innovations have had on community services. Nonetheless, the examples mentioned in an earlier paragraph demonstrate the worth and value of nurse-led innovation but there is a need for further evaluations to look at how innovation has impacted on the service provision and also on the individual health professional implementing the innovation. Also what skill set and knowledge was required to set up the innovation and how did the practitioners acquire these skills?

# 2.8 Setting up an innovative project

The traditional perception of individuals involved in social innovation (often called social entrepreneurs) is that they may be a little eccentric, often a loner, but passionate about making a change to society aiming to systematically transform an unfair societal position (Martin and Osberg, 2015). Previously mentioned is the conception of The Big Issue which was founded by John Bird & Gordon Roddick (1991) as a means of offering homeless people the opportunity to earn a legitimate income. They are examples of individuals who had a strong belief in their idea. Their way of achieving the fulfilment this passion was to observe people and learn from them and they had the ability to visualise a better life for the homeless. This particular innovation project has gone on to improve the lives of many homeless people, easing a social problem and is still going strong 20 years on. What was it that made this such a success? It was an observed gap in social provision and the belief that they could make a positive difference to society. Bird & Roddick filled this gap in social provision but what is not known is what additional support was required to set this up. How was their business case presented? Where did the funding come from? There remains a gap in the market regarding the skill set and support mechanisms in place for innovators in the health and social care sector of community provision.

# 2.9 From theory to developing a skills base - Enablers & Barriers

The philosophy of social innovation encourages the joining together of individuals by an innovative concept (Fanning, in Sines et al, 2013). It is an expectation that all practitioners and team members will be encouraged to work within this concept to enable the 'idea' to become a reality. Notwithstanding these facts there remains a requirement for better understanding of how teams within an organisation can either support or hinder an innovation (Strating and Nieboer, 2010), therefore as the transformation of the social innovation takes place it may change the way people relate to each other. Individuals will need to work through this process to make this happen, whilst recognising that taking a notion through from *idea* (concept) to *actual* (practical application) suggests that an element of risk is probable as innovation and

risk are inextricably linked according to Fleming and Kinder(2015). Risk can be determined by the social structural factors of the organisation and individual characteristics. However, according to Robinson (2014), there are a number of characteristics evident in entrepreneurs that are essential for an idea to succeed, these being, tenacity, passion, tolerance of ambiguity, vision, self-belief, flexibility and rule breaking which are essential to tolerate risk. He suggests that tenacity is the most crucial element as it deals with the notion of 'failing' so the innovators need to have the strength to learn from this. It will almost definitely require a change to the 4Ws – Who, What, When or Where of service delivery (NHS, 2009). Nevertheless, even with this understanding of taking risks it is estimated that approximately 50% of innovation projects within the public sector fail (Tidd and Bessant, 2009). There appears to be little evidence about how management accept risk.

There are concerns in this unstable economic climate where jobs are being lost and constant spending cuts are threatened that individuals might consider that developing any innovation is a risk (Cheater, 2009). Changes in the way services are delivered must be managed diplomatically as potentially these may cause some disagreement amongst teams and management. Nevertheless, in view of Lansley's (2010) recommendation to promote innovation and integrate practice, education and research this provides an opportunity for community practitioners to be vocal in their attempt to promote this new way of working as they plan, develop and put their ideas into practice (Bornstein, 2007:22). In doing so, organisations such as community trusts, independent service providers and policy makers become aware of the benefits of working with social innovators, thus breaking down many of the barriers that exist today between public sector organisations and social enterprises. To this end Innovation should become a way of life at all levels of clinical practice as suggested by Nicholson (2015), but in some instances this might prove difficult as clinicians may not be willing to change their behaviour and the organisations themselves may not want to change, suggesting that 'the combination of these factors could potentially inhibit learning and generating ideas' (Gopalasamy 2010: 5). However, if a culture of leadership practice, whereby having a stake in a venture increases, both organisational and individual commitment to the 'idea' (Raelin, 2010) could potentially promote learning and the generation of new ideas across organisations. Yet, traditionally, the public sector as an organisation tends to be

averse to risk (OECD, 2015), possibly because innovation implies a certain amount of experimentation and inherent challenges associated with innovations in public health services which requires an 'understanding of the art and science around the actual innovation and the organisational context where it occurs' (Joseph, 2015:172). It is therefore advisable to pay attention to those health professionals who are actively trying to get their voices heard in driving innovation forward. Yet, Davis (2011) has commented that nurse managers are in a unique position to support staff and that this vantage point enables them to identify challenges and how to deal with them so that they can drive efficiency and innovation forward.

## Reflection:

I was surprised to read that many practitioners did not want to rock the boat and be seen as trouble makers which potentially has meant that some innovations have not been taken forward. On the other hand the transformation agenda for community services is creating a vulnerability amongst the workforce which has seen a 'business as usual' attitude rather than looking at new ways of working. Telehealth is an interesting concept and has been given a lot of recognition. Yet, in the wider context, as technology is advancing at a rapid speed it could be argued that it is inevitable that technology would be used in the health service and as such is not innovative but is an essential means of progression.

Communication is a theme that runs throughout most nursing and allied health professional programmes yet many of the barriers identified in the literature have come about due to poor communication. It was not surprising to note that some managers may be wary of their staff having more knowledge in certain areas and thus feel more threatened. It is essential therefore that managers are given tools to assist them in assisting others in reaching their potential.

#### 2.10 Skills for successful innovation

Reportedly, innovators typically have a *vision* of a better service provision and have that inner drive and passion to act on it whilst there is a need for the change (Brookes et al, 2011, Nieboer et al, 2012, Pekkarinen et al, 2011). Their main focus is to improve quality of care for their patient/client. Drucker (1957:45) was a great follower of social innovation and supports social innovation, claiming that we require social innovation more than we require technical innovation and that it is a part of a theory of social change (Bornstein, 2007:27). However, the amount of technical innovation that has occurred within the NHS has seen greater interest and investment than those innovations not involving technology (Hellstrom et al, 2002, Oborn et al, 2011, Bacigalupe et al, 2013, Jiwa et al, 2013). Telehealth has seen a

surge of interest as more patients and staff are using this to monitor patient care and medicine management (Sarhan, 2014, NHS, 2017) and Bacigalupe et al (2013: 248) discusses the emerging information and communication technologies (ICT) as new ways of archiving and searching for health information.

Rogers (1962) in a seminal piece of work recognises that innovators are individuals who have an idea and are motivated to become a change agent and not afraid to take risks. As an activity social innovation was often regarded with negative views, and frequently was considered to lead to risks with negative consequences, such as investing in an idea which does not succeed, resulting in a financial loss (Godin, 2015). This view is supported by a study carried out in 2010 by the Centre for Innovation in Health Management, signifying the importance of choosing an innovation carefully in order for organisations to benefit, stressing that it is not just the innovation that needs to be considered, but how it is actually managed (Gopalasamy, 2010). However, other thoughts were that 'Great events may, and do spring from the most trifling causes' (Godin 2013). Consideration has therefore been given to the additional skills and knowledge required to plan and implement an innovation in the community setting. Consequently this research study is the first to address the lack of innovation incorporated within educational programmes for community health professionals.

# 2.11 Educational Support for Innovators

The underpinning philosophy of any educational programme addressing innovation will provide the skills mentioned above and should be designed to enable changes to happen by equipping students with the knowledge, confidence and skills to promote change and innovation in their practice areas. A recent programme developed by NHS England (2015) focusses on the knowledge- based skills mentioned above and is initially targeting doctors in training who are developing clinical innovation. Dependent on the success with this group of people it will extend to allied health professionals. Other programmes such as the University of Oxford Management in Medicine programme and the NHS Leadership programme do not cover the entrepreneurial skills required to set up a project, but rather provide management skills to support such an initiative.

Promises are being made to support innovation in the clinical setting and as recently as 2015 Sir Bruce Keogh launched a Clinician Entrepreneurs programme at the Health and Care Innovation EXPO. The programme will be the first national level training programme for clinician entrepreneurs and will become a global beacon creating a 'brain gain' effect, attracting and retaining the brightest talent for the NHS. Protected time for innovation as part of professional clinical training and the creation of substantive clinician entrepreneur posts across the whole healthcare landscape will enable medics, nurses, AHP and scientists to directly influence the fast-changing environment they work in.

Education generally for post qualifying health professionals has identified similar concerns whether it is looking at promoting innovation or developing competencies within a particular clinical area. A study carried out by Weglicki et al (2015) looking at the continuing professional development needs of nursing and allied health professionals responsible for prescribing found that the key themes were: personal anxiety and lack of confidence in their own competence, external barriers that prevented them working effectively, lack of support and recognising their personal way of learning. A further study by Baldacchino (2011) looked at the perceived impact on qualified nurses of teaching spirituality and the findings showed that the students had a heightened self-awareness which also impacted on their personal development. These findings were also proved by Pool et al (2015) when they looked at strategies for continuing professional development amongst nurses. Holland & Lauder's study (2012:61) suggests that the delivery of care is dependent on the knowledge and skills of the workforce and that in developing more supportive learning environments this was leading to increased innovative practice. They stress that correct leadership and commitment are requirements for a successful transformation of nursing care.

The literature reviewed supports post qualifying education and particularly in view of recent investigations in Mid Staffordshire Trust recommendations are to increase the focus of nurse education to improve the theory-practice gap. Leonard et al (2016:150) comment that nursing practice is homogenous, yet it is multi-faceted and requires a diversity of skills and knowledge. Education is knowledge and knowledge strengthens the power to promote confidence in health professionals so that innovation in practice will be encouraged throughout the workforce.

#### 2.12 Diffusion of Innovation

Diffusion of innovations is a theoretical model, with roots in rural sociology. It is a multidisciplinary theory of planned social change (Goss, 1979) and it addresses how the structure and function of a social system changes (Rogers,1995, cited in Haider & Kreps, 2004).

"To get the bad customs of a country changed and new ones, though better, introduced, it is necessary first to remove the prejudices of the people, enlighten their ignorance, and convince them that their interests will be promoted by the proposed changes; and this is not the work of a day" Benjamin Franklin (1781).

One of the key factors of diffusion is about removing prejudices so that innovators can move forward. Doyle et al (2014) carried out a literature review which explored the use of Rogers' Diffusion of Innovation model as a framework to guide implementation of mobile devices into nursing curricula, which they found supported the planning, implementation and adoption of these technologies. Numerous other researchers (Leggott et al, 2015, Block, 2013, Romero et al, 2011, Hayes et al 2014, Vedel et al, 2013, Schleien et al, 2010, Andrews et al, 2013, Geibert, 2010, Lin et al, 2016, Denis et al, 2002, Jordan et al, 2014, Haider et al, 2004, Jahanmir & Lages, 2015) have adopted Rogers Diffusion of Innovation theory to support their introduction and implementation of new ideas in practice. Rogers' (1995) five key factors of successful diffusion are outlined below:

- Innovators are those individuals who have an idea and are motivated to become a change agent and they are not afraid to take risks.
- Early adopters are frequently people in senior leadership positions who can
  make things happen so can work with the critical mass to approve a change.
  They may take on board the original idea from the innovator and make their
  own adoption decisions
- Early majority are slower than the early adopters but are willing to make a
  change. They are generally members of a team and not the leader of a team.
   These make up about one third of all members of a system.

- Late majority view innovation with wariness and require a lot of encouragement to take the innovation on board. They tend to accept the innovation after everyone else around them has accepted it
- Laggards tend to be averse to change and like to work within the safety net of familiarity and tradition. They are the last to adopt an innovation.

Caution should be taken when using these five characteristics to diffuse an innovation as innovators may be so enthusiastic about an idea that they overrule any other consideration and over adopt.

#### Reflection:

It was interesting for me to discover that studies by Weglicki et al (2015) and Pool et al (2015) highlight that students on post qualifying programmes have low self-esteem and lack confidence, yet many were in senior clinical positions. Is this a cultural phenomenon? Would the same findings be discovered if a similar research was carried out on non-health care professionals? Learning about diffusion of innovation was also interesting and retrospectively as a past community health professional and also in my current role I have been able to categorise people and recognise their traits as described by Rogers (1995).

However there are synergies between the use of diffusion in third world countries and the concepts for diffusion in health care in the UK. As Blaut (1993: 388, cited in McMaster & Wastell, 2005) proposes, diffusion is about making sense of the world and the commonality across the world is the explanation of 'progress'. He suggests that there are core axioms, the centre being progressive and innovative and the periphery being traditional and stagnant. The innovation will commence at the centre and merge into the periphery, as 'air flows into vacuum'. Within the UK Health care system the nurse or allied health professional will be at the centre and influence those managers and team members who are on the periphery.

#### 2.13 Adoption of Innovation

In a longitudinal qualitative study looking at diffusion of a collaborative care model in primary care, carried out by Vedel et al in 2013, they stress the importance of

diffusion being communicated across all parties, ensuring that a critical mass of those involved adopt the innovation for it to have any chance of success. Their study identified that adopting innovation took time and it was the nurses that attained a critical mass faster than other members of the primary care team. Those that were early adopters demonstrated positive traits and the laggards generally found it very hard to break old habits and practise in a different way.

The cultural dimension of innovation looks at how to motivate and encourage people to innovate. While pay does not seem to be a significant motivator of innovation in the public sector, other extrinsic rewards such as recognition, career advancement, special assignments and competitions may play a role. Intrinsic factors such as observing the value and impact of one's work and learning new skills can also be a strong incentive (Blackwell, 2014, Lambrou et al 2010).

There is also the possibility that an adopter may want to make the change to gain social status (Roger1995, 2003). This may be due to wanting to 'fit in' with the innovators and not necessarily that they are in full agreement with the innovation. However within health the researcher would argue that the patient's best interest is central to an innovation being adapted and it would be an added bonus if, coincidentally, the successful innovation led to promotion for the individual.

In December 2011 the Department of Health published an executive summary in response to their invitation calling for evidence and ideas about how the adoption and diffusion of innovations can be accelerated across the NHS. The response claimed that it is a sign of strength to make a commitment to building an economy driven by entrepreneurs who are creating innovative health solutions, facing the challenges of social cohesion, finding solutions to poverty, climate change and sustainability. It means building an innovation agenda that is truly holistic and recognises and supports social innovation and social entrepreneurs as a part of an integrated strategy (DH 2011)

#### 2.14 Innovation and change management

"There is nothing more difficult to plan, more doubtful of success, nor more dangerous to manage than the creation of a new order of things ..... Whenever his enemies have occasion to attack the innovator they do so with the passion of

partisans, while the others defend him sluggishly so that the innovator and his party alike are vulnerable" Niccolo Machiavelli, The Prince (1523)

Introducing an idea may be accepted but may also take some time even if it is known to have advantages (Rogers, 1983, 1995, 2003). Those working within the NHS over the past twenty years and beyond will have worked through change processes on a number of occasions and therefore will have noted that much of the literature around change management within the NHS is linked to innovation. They are, however, two distinctly different concepts. Change Management incorporates organisational tools that can be utilised to help individuals make successful personal transitions resulting in the adoption and realisation of change (Creasey, 2007). Whereas innovation is an idea, service or product that is new or applied differently, which significantly improves the quality of health and care (DH, 2010: 9). The distinction between traditional change management and social innovation can be a grey area as the two are often aligned, but they differ in at least one key quality. This being that change management is not necessarily aiming to improve society, but rather focuses on ways of developing a workforce or systems for the good of an organisation; while social enterprises have the improvement of society as their primary goal. A study by Redfern and Harris (2011) claims that innovation in nursing requires the support of the managers and that nurses need to be skilled in driving innovation forward.

Nonetheless, the importance of supportive management and training were emphasised as key if innovation is going to be encouraged and to succeed (Block, 2013). Paradoxically government is promoting innovation and this is an ideal opportunity for healthcare professionals to make a stand and display some of their original and possibly unconventional tactics, yet often when innovation is introduced within an organisation it is turned down because senior managers are not comfortable with what might be perceived as rebellious behaviour. If innovation is to flourish, then support mechanisms are required to maintain and develop these initiatives (DH, 2011) which at times can be risky. This is a view that Mullahy & Boling (2008) support as they remark that, to be entrepreneurs, professionals must be prepared to accept and also to be able to say 'No' when they are advocating for their patients, in some instances this may mean that risks will be taken. Alongside this they suggest that perseverance and optimism are attributes required to confidently take risks.

Douglas et al (2016:3) wrote about the diffusion of a management innovation and stated that this is frequently driven by dissatisfaction with the inability to maintain the status quo and suggested that the overall success will depend on whether the innovation fits with the plans of the organisation. Frequently restructuring of an organisation is required and when looking at improving performance they introduced a top-down approach which was practised across the whole of the organisation but did not result in all staff being early adopters. There were various stages of adopting the innovation, with some laggards unable to accept the changes. These individuals were given more time to gain an understanding, eventually buying into the change.

The NHS is noted for working to procedures and processes which can make change difficult to manage. Nevertheless Carr & Clarke (2010) assert that Nurse Managers are pivotal in improving services and promoting innovation. They identify the importance of organisational learning as a means of sharing knowledge and applying and transforming practice to improve patient care. It is acknowledged that there is a link between learning and change and refers to 'double-loop' learning where individuals' assumptions and beliefs are tested. Carr & Clarke (2010) stress the significance of exchanging knowledge and an effective manager will acknowledge that frequently knowledge is not disseminated and remains the possession of one individual in an organisation. This is not conducive to promoting innovation and they conclude that organisations facilitating learning at both an individual and organisational level improve the health and social needs of their population and service innovation achievements.

McEwan et al (2010) stress the significance that empowerment has on organisational change and recognise that change takes time and does not always follow a linear pattern. Through the process of change individuals may encounter personal growth; frequently this results in an increase in confidence and the ability to challenge new ideas as they are presented. However, Joseph (2015:172) suggests that innovation requires organisational commitment to allow employees to inquire and question organisational practices. Nurses are seen to be able to promote innovation by providing creative solutions, thus making a difference to patients and the organisation (ICN, 2009).

#### Reflection:

There is a gap in the literature on the topic of innovation and it was not surprising to find that innovation as a topic did not form part of change management literature.

## 2.15 The experience of social innovators

With health policy driving changes to the way the community workforce provide their services there is limited evidence to determine how community practitioners are being prepared for their role as innovators, either at the point of initial post registration training or later in their career. The impact of such demands will require practitioners to address the government's drive to bring care closer to home through demonstration of innovative practice such as the PACE project (Post-acute Care Enablement) which is in operation at the Royal Free NHS Foundation Trust in London. This project aims to provide complex care in the patients' home, focussing on the community provision and the ability to lead the safe and appropriate transfer of patients from acute bed-based models of care to community settings, which had traditionally been poorly delivered. This required a collaborative partnership approach with health and social care to deliver cost and quality outcomes as part of a 'supply-chain' model of reform (Hails, Steadman, Walls, 2010). It is a crossorganisational service model that offers medically stable hospital patients the opportunity to be cared for at home through the provision of integrated acute, community and social care. The aim is to reduce costs, and provide more comfort for the patient and their families. The evaluation of the project proved that PACE patients spend an average of three fewer days in hospital and consistently report an 85% level of satisfaction with the care they receive from the PACE team (Hails et al, 2010:3). This model has now been replicated in other regions, for example Edinburgh and Bromley. This team recognised that they were challenged to gain skills unfamiliar to them as they had no previous training in management or operationalising projects. (Hails et al, 2010:81). This current research will review this aspect to see if education or training in innovation assists practitioners who want to carry out projects such as this, without the difficulties that this team experienced.

The introduction of the NHS Institute for Innovation and Improvement (2005) has given immense assistance to practitioners wanting to support the transformation of

the NHS through innovation (Butterworth, 2013). The researcher argues that if there was more support, education and awareness of how to execute an innovative idea then this would further support the transformation of the community service agenda (DH, 2011, 2015).

For many people working within a bureaucratic system, such as the National Health Service, displays of 'thinking outside the box' have not been encouraged; conversely, employees have been expected to follow the rules and guidelines which are often target driven (Keogh, 2015). Even though patient safety has, in theory, always been high on the government agenda (National Patient Safety Agency, 2010) up until very recently even reporting incidents of poor practice have been left unreported because of an attitude and fear that individual jobs could be jeopardised. Yet, since recognising the need for a policy on 'whistleblowing' this has encouraged a 'climate of openness' (DH, 2010). Whilst one of the main reasons for introducing this policy was to safeguard the public by staff having 'permission' to report bad practice it also gave staff working within the NHS a platform to be heard. There has been a culture of fear within the NHS which leaves staff not wanting to put their head above the parapet. It transpired in the Baby P case that there were grounds for such concern as the consultant paediatrician who reported poor record keeping and understaffing had employment laws used to gag her (Butler, 2011).

## 2.16 The gap between policy, practice and staff training

NHS policies are clearly promoting innovation within clinical practice (DH, 2010, 2015) but these potential opportunities will only come to fruition if there is support available from managers as it is essential for a member of staff to be heard if they want changes to happen. The Department for Innovation, Universities & Skill's (DIUS) White Paper 'Innovation Nation' (2008) provides an ambitious new direction for UK innovation policy. However, current evidence indicates that organisations need to understand how to identify the characteristics and behaviours of innovative people and consequently, how to promote and encourage innovative working within organisations.

## 2.17 Community Healthcare Innovation Education/Training in the UK

It is an expectation that innovation in nursing is pivotal to the delivery and maintenance of evidence based care whether it is in a hospital setting or a patient's home (DH, 2008, Apekey et al 2011, Ross et al, 2011). Therefore to influence change across the healthcare sector, evidence of how community health professionals have impacted on service delivery must be demonstrated. A number of the innovations in the 'Transforming Health & Social Care' (DH, 2009b) document have looked at whole organisations developing innovation and change in their community. However, there appears to be little research available which investigates how community health professionals are prepared for the role and the impact of their innovative practice on the individual client groups. One of the few is a study carried out by Ross et al in 2011, which looked at the professional and personal features, mechanisms on innovation and outcomes in nursing which claimed a link between a supportive employing organisation and individual personal characteristics as being influencing factors when working together to implement an innovation. It reports that innovation in nursing requires the support of the managers and that nurses need to be skilled in driving innovation forward. Trueland (2014) reports that the one stop clinic delivered in the Forth Valley Royal Hospital is successful because since the conception the service has had the full support of the multi-disciplinary team and is not reliant on just one individual to lead the service. The importance of knowledge brokering as a means of disseminating best practice, stresses that there must be links between clinical practice, research and policy makers (Gerrish et al 2011). Knowledge acquired by the practitioners in the study is focussed on evidence based practice as a result of studying clinical courses. However, frequently, nurses and allied health professionals are directed towards management courses to further their leadership skills, and often these have a tendency to focus on motivating staff and giving training on working effectively in teams (Frankel, 2008). Crucially, however, these leadership and management programmes do not tackle the issue of innovation in sufficient depth to enable health professionals to have a noticeable impact on practice. These thoughts are replicated by Frymire (2006:11) who argues that one of the biggest challenges is hiring people with the brainpower (both natural and trained) to be able to think creatively. The view of Cash (2000) is that there is conflict between education, practice and management, leaving health professionals to make

a stand in 'arguing for the virtues of good clinical and educational practice' (Cash, 2000:p.258). Hence researching the effect and impact of the inclusion of innovation as a topic within nursing and allied health professional programmes, and how this would support and aid community health practitioners, service users and the organisations concerned, is of paramount importance.

## 2.18 A gap in the literature

The research reviewed presented research on technical innovation and innovative practices led by medical practitioners. A selection of literature identified small scale practical innovations within clinical practice but there appears to be no published research which looks at what effect the process of being an innovator has on the individuals concerned or the impact that education has had on how they drive the innovation forward. It is clear from the literature reviewed, that a crucial feature of successful healthcare delivery is the human factor - systems will only go so far in the delivery of competent care. However, a preliminary overview of literature demonstrated that despite much being written about social innovation in its many forms, there is a lack of research showing what support is in place for those individuals wanting to promote innovation in the workplace, together with lack of research identifying where innovation is included within health professional education. There is a gap in the research literature which this study aims to go some way to address.

Therefore, with a view to capturing this information the aims and objectives of this study are:

#### 2.19 The Aim:

To undertake a qualitative investigation of the personal and professional development of a group of community healthcare professionals who, after training in social innovation, instigated and managed their own innovative community healthcare initiatives.

## **Objectives**

- To explore and identify the personal qualities, knowledge and skills required
  of health care innovators when setting up and managing a social innovation
  project
- To identify the difficulties encountered within individuals themselves and the external factors within their organisations that have impacted on the development of their project
- 3. To look at the impact of education and training on the progress of these participants.
- 4. To inform potential innovators about what they can expect to encounter as they proceed with their own project

#### Reflection:

The literature accessed to date has seen an increase of attention given to innovation in the workplace, but it has also shown that much of this is lip service and no additional resources have been allocated to enable innovations to be implemented. Additional resources, either financial or manpower, are not always required but there is no evidence shown that shows 'time to think and reflect' is given to support innovation in the workplace. No literature has been accessed that tackles the issue of personal and professional development of health and social care innovators either and I am hopeful that the participants involved in this study will give some insight into what are the key drivers, enablers, inhibitors and also the impact, if any, education has had on them as clinical practitioners trying to introduce an innovation into their practice area.

## Chapter 3 Methods

This chapter sets out an overview of the research process carried out during the period of 2012 – 2015. It is divided into six sections: the first outlines the rationale for choosing a qualitative approach for this study. The second section addresses the sampling and recruitment strategies and methods. The third discusses the data collection process, issues arising and how they were overcome. The fourth outlines the specific analytical approach and procedures undertaken, and how they are based on Braun & Clarke's (2006) guidelines. The fifth section focusses on the ethical concerns of the study and the trustworthiness of the findings described and established. Finally, the issues around quality assurance and research governances are discussed.

## 3.1 A qualitative research approach

This is a qualitative research study. The aims of qualitative research are to seek answers to questions about the 'what', 'how' or 'why' of a phenomenon, rather than questions about 'how many' or 'how much' (Green and Thorogood, 2004:5). The purpose of this study is to understand the personal and professional journey undertaken by a cohort of community health professionals after completing a Post Graduate course with the aim of developing their social innovation knowledge and skills. This study will elicit information about the key drivers observed in these professionals that enabled them to be proactive in promoting new healthcare initiatives in their workplace. The journey described here covers what was considered to be important to these individuals when seeking to improve their organisation's service delivery; how they try to achieve their goals, and how the course impacted on their ability to set up an innovative project within their professional domain. A 'critical realist' research approach has been adopted. The work of Ritchie et al (2013) has been influential in selecting this approach because of their long record of research within healthcare. Their approach is that of being broadly 'critical realists' (Robson, 2002, Bhaskar, 1978, cited in Ritchie et al, 2009). Ritchie (2013: 21) states that 'ontologically, we see reality as something that exists independently of those who observe it but is only accessible through the perceptions and interpretations of individuals'. We recognise the critical importance

of participants' own interpretations of the issues researched and believe that their varying vantage points will yield different types of understanding. The reality of healthcare delivery in all its forms exists independently, but it is through the words of the research subjects that we have a unique access to it and, in this case, their desire to instigate change and improvement in professional practice.

Ritchie (2013) further defines this position 'as being within a broadly interpretative framework which emphasises the importance of understanding people's perspectives in the context of the conditions and circumstances of their lives' (Ritchie, 2013:22). This sums up very neatly the aim of this study which is to explore how the thinking of the professionals has been shaped by the variety of circumstances that they are working in. The aim here will be to identify common themes related to attitude and interpretation of their personal experiences in setting up an innovative project within their clinical practice. There is, according to Ritchie (2013:22), 'a strong requirement for any interpretation to be heavily grounded in and supported by the data'. This level of openness to the thoughts and idea of the participants required an inductive approach to the data analysis (the meaning of the material emerges from the participants themselves rather than from a predetermined theory) which has been achieved, using the guidelines for data analysis of Braun and Clarke (2006). This study is not, therefore, embedded within any major theoretical research tradition, but rather aims to achieve 'empathic neutrality' (Ritchie et al. 2008:22) with the researcher doing her best to be unbiased, and neutral in her data collection.

# 3.2 Other research approaches considered but rejected

This section will look at other qualitative research approaches that were considered but rejected the first being grounded theory which is frequently used within the field of health and social science; indeed, it had its origins in a study of dying patients in a hospital setting (Glaser & Strauss, 1965). It was contemplated, then rejected because researchers develop inductive theoretical analysis from their data as they go along and continue to collate further data, collecting and analysing data simultaneously (Silverman, 2011). This protracted method of data collection was not considered necessary here as 'one off' interviews were all that was considered practicable and would be sufficient. It is not the aim of this study to create a new

theory of social innovation from the data but rather, as the topic of the personal journey and experiences of social innovators is little researched, simply carry out a descriptive study which may in time lead to a larger piece of research in this area.

An ethnographic research approach was also considered because the researcher was previously a District Nurse, therefore arguably she could be referred to as a 'native' rather than a 'stranger' to the participants, thus potentially enabling the researcher to prioritise and rationalise the findings as they are seen through the eyes of the insider. However, the 'stranger' element of the role would require a conscious effort to enable naïve questions to be asked and to analyse social life from a theoretical perspective (Green & Thorogood, 2004:136). Ethnography in the truest sense is carried out 'in the field' and the focus of this study did not revolve around field observation so for that reason this approach was discarded.

Narrative analysis which focuses on the stories that participants share involving aspects of their lives (Robson, 2011) was also considered. While at first glance this was appealing as this research explores the participants' journey towards becoming successful innovators, in essence there was a story to be told. However, although the data received from the interviews were comprehensive and demonstrated individual personal development, the structure of the interviews did not encourage 'story telling' (Robson 2011), but rather a synopsis of how education, training and becoming an innovator impacted on them, thus falling short of a true narrative approach. Narrative approaches have been used to provide rich and detailed accounts of the social formations shaping subjective experiences of health and well-being (Riessman, 2007), but the use of narrative analysis would detract from the structured approach required for this study, so this approach was rejected also.

#### 3.3 Data Collection Methods

Initially focus groups were considered to be the most effective means of gathering data. However, due to the difficulties experienced in gathering focus groups together this method was discarded and preparations had to be made for the unexpected (Krueger & Casey, 2009). Qualitative research can often use only one interview method but as Deakin & Wakefield (2013:2) observe 'it is increasingly likely that one or more types of interview are employed'. For that reason, alternative methods of

interviewing were considered as the participants came from different locations and the prospect of travelling to Scotland, Wales, Northern Ireland and across numerous counties in England for individual interviews had also to be justified in the light of possible poor response rates. Resources, finances and time factors had to be taken into consideration also. Conducting face to face interviews for those participants who had agreed to participate would be too time consuming and expensive due to the geographical distances, a view supported by Deakin & Wakefield (2013, Sedgwick & Spiers, 2009). Therefore alternative qualitative approaches were appraised and the eventual mechanism of gathering data was semi-structured interviews in the form of individual Skype interviews. Skype interviewing has for some time been accepted as an effective way of managing verbal exchanges and is considered to be a 'conversation with a structure and a purpose' (Cohen et al, 2007). This notion of it being a 'conversation' is apt, as this is how the respondents clearly viewed it. Rich, free flowing data was obtained from each participant which was why semi-structured interviewing was considered to be more effective with these busy, but very enthusiastic, participants rather than a survey questionnaire which relies largely on closed questions (Robson, 2004, Silverman, 2011), not enabling them to speak freely about their experiences. Having met several of these participants as students or as presenters at conferences, it was apparent that their characteristics showed them to generally be people with a passion for their work and a willingness to share their innovative ideas to a wider audience. Because the participants by nature were willing to veer from traditional practices, they were eager to participate in Skype interviewing, thus supporting the views of Edward & Holland (2013:95), who state that 'there have been dramatic changes in communication technology and qualitative interviewing must adapt if it is to survive'.

#### 3.4 Sampling and Recruitment

This section details the two sampling methods used: *purposeful* and *opportunistic* or *emergent* sampling, how the participants were recruited and their demographic details. A purposive sample of twenty community health professionals from regions across the United Kingdom (UK) were selected, together with an additional six participants whose contribution was achieved using convenience and opportunistic sampling as the study progressed. The sample frame for the majority of the

participants was the enrolment data set for the Post Graduate Diploma in Social Innovation course they had studied.

The intention of this study was to talk with people who could produce rich, well informed material about their personal experiences when setting up and managing innovative projects in clinical practice. The participants were selected on the grounds of preparing to start their own project following the course, or their extensive involvement with an existing project of their own. The samples identified sufficient diversity, which according to Patton (1990) yields two kinds of findings; a) samples of a high quality and b) identifying shared patterns across the data which emerged out of heterogeneity (Patton, 1990: 172).

## 3.4.1 Purposive sampling

The researcher had developed and taught on the Post Graduate Diploma in Social Innovation at a university in the South of England and watching the students develop professionally, and, most importantly, personally was the catalyst that prompted this research. These students were an ideal sample for the purpose because of their diversity and ease of contact, therefore the purposive sample was identified and the researcher approached two cohorts of students. It was straightforward to contact this group as the researcher was able to acquire all of their email contacts from the university administrator. Qualitative inquiry generally focuses on small samples which have been purposefully selected (Patton, 1990:169) so this was an ideal sample for the purpose. Forty people in the initial phase of the study were contacted to ask if they would be willing to participate and twenty seven responses were received. This resulted in twenty recruits. Seven of the initial responders did not respond to subsequent emails so the researcher did not pursue them further. As the fifty percent response rate included participants from across the UK and from a range of professional backgrounds, the researcher reasoned that this was a sufficient sample to proceed with the research (Blaikie, 2011).

# 3.4.2 Opportunistic sampling

The additional six respondents were approached after hearing them speak at various conferences or after reading about their projects in the professional literature. The

inclusion of the opportunistic respondents captured a broad range of innovation project experience and reduced any potential for bias by not only including the researcher's own students but also participants who had studied elsewhere or participants who had successfully implemented an innovative project in their clinical practice. Initially, there had been no intention of seeking more participants other than the students from the researcher's university, but having heard the presentations of the additional participants the researcher was of the opinion that their experience would add strength to this study. Including these extra participants did not incur any further expenditure and they were very willing to participate, so all that was required was additional time to interview them, which would enlarge the data set (Roller & Lavrakas, 2015, Patton, 1990).

The first two additional participants were identified at a conference in London in 2014 where they spoke about setting up a social enterprise in their General Practice surgery. The project was a lymphedema service offered to the local community which prevented long waiting lists for hospital lymphedema clinics. The researcher approached them and asked if they would be willing to participate in this research study which they were happy to be involved in. Another two participants were approached at a conference in 2015. One practitioner had developed a General Practice and Community Development programme which was targeted at general practice nurses and district nurses; this had improved the service provision for communities in a rural community. The second practitioner was a community matron responsible for care homes in her local area who was concerned about the high infection rate among patients. She developed a training session for staff and since her input there has been a 70% reduction of infection related admissions to hospital

A further two participants were approached after the researcher read about their innovative projects. One project was 'Community at Risk (CART) Tool'. A prediction tool to assess vulnerable patients in the community was developed and those predicted to be highest at risk of adverse outcomes were offered individualized care pathways. This saw a reduction of emergency admissions to hospital. The second project was a Community Cardiac Rehabilitation Women's Walk Group; the aim of this project was to improve the physical activity levels of women referred to the cardiac rehabilitation team. The patients' quality of life indicators were improved and their blood pressure was maintained at an optimal level. Both of these projects were

of interest to the researcher, mainly because they were simple, cost effective and there was evidence of improved patient outcomes.

Permission was sought from the six participants and they were sent the same information as the purposive sample, which included information about the research and a consent form (Appendices 2 & 3). They all agreed to take part in the study and Skype interviews were arranged for two of the participants and four of the participants agreed to a face to face interview where they were provided with the same information (Appendices 2 & 3) prior to their individual interviews.

#### 3.5 Recruitment

The twenty respondents were given an information sheet about the study (Appendix 2), and a consent form with information sheet regarding confidentiality and storing of information along with detailed instructions should they decide to opt out of the study at a later date (Appendix 3). Personal demographic was collected and information about the type of innovative projects the participants are engaged with was also collected to provide a detailed picture of the various community needs that they are attempting to address. Agreement to partake in the study was granted (no one refused) from them all prior to being interviewed.

The sample included male and female participants but it was intriguing and noteworthy that the respondents were predominantly white British females with only two male participants. This narrow demographic spread is surprising. Although three quarters of the NHS workforce are women, the percentage of males participating in the study is a little below expectation. In addition, considering the NHS workforce is ethnically diverse, it is curious that only two participants were not British. It may be a reflection of research findings on ethnic diversity across the NHS and the proportionately unbalanced representation of all groups throughout the NHS hierarchy (Rao, 2014). This is an area that may warrant further research in relation to who takes up places on social innovation courses. The group had a mean age of 43yrs (table 1).

Table 1

Occupation	Number of Participants
Health Visitor	5
District Nurse	6
Specialist Nurse	5
Community Matron	2
Learning Disability Nurse	2
Mental Health Nurse	2
Dietitian	1
Midwife	1
Occupational Therapist	1
School Nurse	1
NHS Agenda For Change Banding	Grade of Participants
Band 5	0
Band 6	18
Band 7	6
Band 8+	2
Age	Age of Participants
25-29	0
30-39	8
40-49	10
50-59	6
60+	2
Location	Location of Clinical Practice
South West England	6
South East England	7
North West England	6
North East England	5
Scotland	1

The geographic locations of the participants' clinical practice spanned the UK.

Participants were selected from the following groups:

- Twenty participants who had enrolled onto a course at the researcher's area of work
- Four participants were selected after hearing them speaking at conferences about their innovation
- Two participants were selected after reading about an innovative project in a nursing journal

The details in Table 1 show that most of the participants have a background in nursing, with just two from allied health professional disciplines which reflect the community nursing and allied health workforce. Allied health professionals make up six percent of the workforce whereas nurses make up sixty percent (Dorning & Bardsley, 2014, Unison, 2015). These details are representative of the current workforce. Christie & Co. (2015) claim that 21% are between the ages of 30-39, 30% are between the ages of 40-49, 17% between the ages of 50-54 and 20% being over 55 years of age.

None of the participants, with the exception of the two who had set up their own social enterprise, were in positions above a Band 7 according to the NHS pay 'Agenda for Change' grading. All registered nurses upon qualification are awarded a Band 5. Allied health professionals normally are awarded a Band 6 upon qualification. This is due to the fact that allied health professionals have always been a graduate profession, whereas nurse education has only been an all graduate profession since 2009.

The purposeful sample consisted of students who had been recruited nationally onto the post graduate diploma in social innovation which had enrolled students from across the UK and been delivered in two locations, one in the South and the second in the North of England. The conferences and literature attracted delegates and publications from across the UK. The wide diversity of the participants enabled the researcher to consider how innovation impacts on the various populations across the country.

#### 3.6 Data collection

This section will cover the method of data collection employed: Skype interviews and face to face interviews which enabled the exploration of unanticipated issues as they emerged (Ritchie and Lewis, 2009:47).

#### 3.6.1 Interviews

The interview guide was developed in three stages. Stage one, the exploration stage, involved brainstorming with work colleagues, phrasing and rephrasing the questions and working with peers to sequence and estimate time for each question (Krueger & Casey, 2009). Stage two, the interview schedule, was then tested on students who had undertaken a leadership module at the researcher's university. their feedback informed the next stage of the interview development. Stage three: suggestions from peers and students were collated and a pilot interview schedule was developed which was tested on a third cohort who had not yet completed the course on social innovation. They were half way through the course so they were able to give relevant feedback and suggestions. One example was to suggest that a question was rephrased so that it was an open ended question. The pilot question originally read – 'Did you benefit from the course you attended?' The amended version reads 'Having been on a course how has your practice changed?' The suggestions and feedback were reviewed and the questionnaire was refined to enable the participants to explore their own personal journeys. The rephrasing of the questions also enabled the questionnaire (Appendix 4) to be used on respondents who had attended any course on innovation, not just those who had been studying at the researcher's University.

The purposive sample interviews were undertaken from September 2014 to June 2015 and the opportunistic sampling occurred over a period of fifteen months from September 2014 to December 2015. The participants came from various allied health professional backgrounds and twenty two of them had studied some form of educational course that promoted innovation in practice. Two of the participants had never been on any educational course yet had been successful in promoting innovation in their clinical practice. Twenty of the participants had been students on a course at the researcher's university; a further two participants had been on an NHS leadership programme which had promoted innovation, two had been on training days which looked at enhancing clinical practice, and the remaining two participants had not attended any such course, programme or training days.

Table 2

The Education/Training in innovation the participants had received.			
Type of Education/Training	Length of Education/ Training	Number of participants	
Accredited course/ programme at M level	12 months	20	
Non accredited course/programme	12 days across 12 months	2	
Study day	2 days	2	
No education/training in innovation		2	

## 3.7 Focus Groups – difficulties encountered

The purpose of arranging focus groups was a way of gathering similar minded people together to discuss factors and determine feelings, views and perceptions pertaining to the study (Cresswell, 2011, Krueger & Casey, 2009, Silverman, 2011). This means of data collection is frequently used when a researcher is looking for a diverse and comprehensive range of ideas or feelings that people have about a specific topic (Krueger & Casey, 2009:19).

It has already been established that the participants came from various regions across the UK so focus groups were arranged in two major cities, London in the South and Manchester in the North. The first focus group in London had an anticipated eight participants, however only two participants attended. They had changed their working schedule to attend so individual face to face interviews were undertaken as two participants were not sufficient to run a focus group. Each participant waited in an adjoining office whilst the other was being interviewed. The location was in a basement office in the surgery of one of the participants. It was spacious and airy and the atmosphere was calm, promoting a relaxed environment (Ritchie & Rallis, 2009). The participants had already signed the permission to participate in the study and the confidentiality form so the interviews commenced and were recorded with the researcher sitting at one side of a coffee table and the interviewee sitting opposite. These seating arrangements promoted a positive interaction between the participant and the researcher while the researcher was asking questions (Ritchie & Rallis, 2009). Unknown to the researcher there was a railway line next to the surgery and ten minutes into the first interview a train passed by shaking the room and interfering with the recording. Each time a train went by the interview was paused. The two interviews lasted 105 minutes, due to these interruptions.

The second focus group in Manchester was arranged for a week later, but this was cancelled at the last minute due to several participants being unable to attend. This was a dilemma that looked difficult to resolve. Knowing the positive reaction received from the requests to partake in the study, the researcher was fully aware that the reasons for poor attendance and cancellations were not due to apathy, but to work commitments. This was one of the concerns held by Krueger & Casey (2009) who mentioned that logistics was a key concern when planning focus groups.

## 3.8 The move to Skype interviewing

Clearly another method had to be considered. Qualitative research frequently uses only one interview method, but as Deakin & Wakefield (2013:2) observe "it is increasingly likely that one or more types of interview are employed". For that reason, alternative methods of interviewing were considered. The prospect of travelling to Scotland, Wales, Northern Ireland and across numerous counties in England for individual interviews was difficult to justify in the light of possible poor showings. Resources and time factors also had to be taken into consideration. Therefore with the increasing availability of digital technology (Weller, 2015) Skype interviewing was chosen as a way of gathering data from the participants from the purposive sample and two participants from the opportunistic sample also agreed to be interviewed via Skype.

All remaining participants were approached giving them the opportunity to take part in, or decline, a Skype and video recording. Although O'Connor (2008, cited in Deaken & Wakefield, 2013:3) implies that the non–verbal cues that help contextualize the interviewee in a face to face interview could decrease the interviewer effect in Skype interviewing.

On one occasion the video recording was not compatible with the participant's computer so only an audio interview took place. This did not seem to detract from the interview, possibly because the participant was known to the researcher, so a familiarity already existed and the interview continued ensuring the participant was

not disadvantaged by the technical failure. Nevertheless, this sort of incident was identified as a concern by Sullivan (2012) claiming that technology may not always be reliable so a back-up plan should be available. This was not required, but telephone details of each participant were available and the researcher had the iphone voice memo recording device ready should it be required.

Taking all of these factors into account the 'benefits of using Skype as a method of data collection...., definitely outweighs the drawbacks' (Sullivan, 2012: 59), whilst also proving to be the most cost effective approach owing to the limited resources available for this research study. The interviews encouraged the participants to draw on their current clinical expertise and the interviews came alive with the participants' enthusiasm and expert knowledge. As a result each interview lasted at least 45 minutes with one interview lasting one hour and fifteen minutes. From the researcher's perspective it prompted the participants to share their personal development and it reduced the anxiety of travelling, taking time away from work and there were no cost implications; these views were supported by the participants also and had previously been experienced by Hanna (2012) in his own doctoral research studies.

To date, as a research method, Skype interviewing has not been used extensively in research but there is now a consensus of opinion that it offers researchers a novel interview method as a means of collecting qualitative data, according to Deakin & Wakefield (2013, cited in Janghorban, Roudsari, Taghipour 2014:1). Since 2000 there has been an increase of internet use by 36.2% and the use of Skype as a research method is far more viable now and due to the significant amount of internet usage such as broadband it is far more accessible (Deakin & Wakefield, 2013).

#### 3.9 Authenticity of response

Qualitative research is sometimes questioned as to whether participants are actually being genuine in their responses: are they presenting an authentic self? (Sullivan, 2012). At one level this is impossible to know, but the researcher has a duty to design an interview experience that is as stress free and private as possible, thereby raising the likelihood that the participant has no reason to be anything other than themselves and honest with the interviewer with no anticipation of repercussions.

Skype helped with this goal; the participants were interviewed at times that fitted their schedule, at a location chosen by themselves away from any influencing factors in their work environment. In fact one participant requested to be interviewed on a Saturday morning in the relaxed domain of their home. When the interview commenced it was apparent that she was wearing pyjamas and drinking tea. This was the 'authentic self' so contraindicates what Sullivan has been suggesting. At the same time Sullivan (2012) purports that Skype mimics face to face interviews, which enhanced confidence in commencing the Skype interviews. Bargh et al (2002; Ellison et al, 2006: 418) suggest:

"In comparison to face-to-face interactions, Internet interactions allow individuals to better express aspects of their true selves – aspects of themselves that they wanted to express but felt unable to".

A factor evident during the Skype interviews was the generosity of the participants' time. Even though the majority of the interviews lasted no longer than one hour the participants were all eager to share their experience and time was not a factor. The researcher concluded each interview, with many participants offering to participate in further Skype interviews if further information was required at a later stage. This was actually taken up, as a further four questions were asked of the participants later in the study after the key themes had been identified (Appendix 5)

Contrary to the traditional face to face interviewing which is seen as the 'gold standard' method of interviewing, Madge and O'Connor (2004) proclaim that Skype interviewing may promote reflection, especially when approaching sensitive topics. These experiences cannot always be anticipated, but Deakin and Wakefield (2013, Edwards and Holland 2013, cited in Weller 2015:28) suggest that the physical distance between the researcher and participant might make some feel more at ease when discussing sensitive issues. The researcher experienced this during a number of the Skype interviews. An excerpt from one of the participant's interview stated:

"As an Occupational Therapist I would look at how things were being done and say to myself I wouldn't do it that way, but I never had the confidence to challenge colleagues' practice. I look back and think if I had, the patients would possibly have been discharged earlier".

Seitz's (2015, cited in Weller, 2015:28) view is that Skype interviewing presents an emotional barrier but this was not the experience of the researcher as many of the participants were relaxed and disclosed sensitive issues which added a richness to the data obtained. One extreme example was that of a participant being 'emotionally bullied' by the organisation as a result of her determination to change practice. She volunteered this information freely and mentioned that the interview had enabled her to reflect on her responses and see things from the organisation's point of view.

#### 3.10 Face to face interviews

The remaining four participants of the opportunistic sample agreed to face to face interviews which were carried out in a quiet convenient location. These were recorded with each participant's permission and notes were also taken. The interviews were structured in line with the Skype interviews so that there was consistency in gathering data. The nature of the questions enabled the participants to respond freely (Rossman & Rallis, 2012).

## 3.11 Data Analysis

Thematic analysis was recognised as a suitable approach for the type of data expected to be generated by the methods employed and this study design. First introduced in the 1970s, it was used in a variety of ways but inconsistently (Braun & Clarke, 2014). In 1998 Boyatzis wrote a key text on 'coding and theme development' that laid down specifications and guidelines which was the catalyst that prompted 'thematic analysis' to prosper, particularly in the health and social science field. One of the attractions of thematic analysis is that it is theoretically flexible, not linked to any particular epistemological or ontological framework. Nevertheless, the underpinning epistemology is an essentialist/realist approach whereby motivation, experience and meaning is explored in a meaningful but straightforward way, contextualising and giving the participants 'a voice' to each individual story (Sobh & Perry, 2006).

Thematic analysis identifies, analyses and reports patterns (themes) within data. It minimally organises and describes a data set in rich detail, Braun & Clarke (2006) argue that there have been various approaches to using thematic analysis which

may confuse a new researcher, so they developed a flexible, yet structured process to collecting and analysing data which was relevant to this particular research study. It enables the researcher to revisit responses and build on them, suggesting that this should be seen as a foundational method for qualitative analysis (Braun & Clarke, 2006: 78).

This version collates qualitative data to seek themes which are then used for coding to identify patterns across the dataset in relation to the research question. The question of what 'level' patterns are sought, and what interpretations are made of those patterns, are left to the researcher to provide a robust, systematic framework for coding. It offers a theoretically flexible approach; whereas others (e.g., Boyatzis, 1998; Guest et al, 2012; Joffe, 2011) position thematic analysis implicitly or explicitly within more realist/post positivist paradigms (Braun & Clark, 2014).

Braun & Clark's (2006:5) six–phase process for thematic analysis was used. This involved the researcher becoming immersed with the data in order to generate the initial codes, followed by searching for major and sub themes, reviewing the themes and defining and naming them.

## 3.12 Data management

Each case was numbered and initialled to maintain confidentiality. A key was developed which matched the name with a case number and was stored securely and separately from the transcripts. The researcher began transcribing the first two interview recordings but the recording was very faint and it was interrupted by the noise of the train. It was difficult to decipher the interviews as one of the participants spoke softly so, added to the sound interference, the recording was jeopardised. However, notes had been taken at the time of the interview and were used as support to familiarise myself with the material. Agreement was sought from both participants to ask if their recording could be sent to a technical advisor as it was difficult to retrieve the complete content of the interview and they both agreed to this action. The transcripts were then sent to the two participants to ensure that the information was accurate, both responded positively, but also added further information that they had not thought of at the time of the interview or that may have

been misinterpreted when transcribed. This reinforces the importance of reviewing the data with the participants involved in the study.

## 3.13. Transcribing

To capture all of the data in a timely way a professional transcriber was employed to transcribe the remaining interviews. Ensuring confidentiality was maintained as the transcriber signed a confidentiality agreement (Appendix 8).

The purposive sample interviews were arranged over a period of six months. Knowing that the interviews were detailed, displaying the participants' sometimes flamboyant characteristics, the interviews were sent to the transcriber approximately four at a time. As a validity check the transcripts were sent to the participants to confirm accuracy. This enabled the findings to be turned around without delay.

The transcriptions were verbatim, each utterance, hesitation and casual phrase was included in the returned transcripts as is considered good practice (Green & Thorogood, 2004). Whilst reading the transcripts, notes were made of any 'word or brief phrase that captures the essence of why you think a particular bit of data may be useful' as recommended by Braun & Clarke (2013:207). This assisted with the initial coding as key phrases were highlighted such as "I feel more confident", "I am a knowledgeable practitioner", "we learn from failure". Such comments and similar statements were common throughout the responses.

#### 3.14 Coding

The analysis was data driven and this process of coding was challenging as many of the codes seemed to stand in isolation and, on occasions, did not seem to slot into any specific established theme that had been identified from the coding process. One such phrase used by a Health Visitor (participant 4) "I think innovation is an incredibly lonely road, and it's like walking down a street listening to your own footsteps" was so profound but did not sit with any other code. When other statements were made that stood out but were felt to be of importance they were given a separate code. As the codes became more apparent some of the 'one off' statements merged into a particular theme.

#### Table 3

#### Codes & Themes

**Codes:** Professional development, thinking outside the box, taking measured risk, improve services, maintain excellence, lack of vision, value for money, communication, lack of recognition, career aspirations, confidence in my ability, career aspirations



#### Theme 1: Better is possible – a perceived need to save time & resources

**Codes:** Self-belief, lack of vision, confidence, frustration, self-doubt, perseverance, barriers, lack of support, change, challenges



# Theme 2: Swimming against the stream – the challenges faced when setting out ideas for improvement

**Codes:** Lack of confidence, barriers, uncertainty, involve others, tenacity, education, perseverance, networking, motivation, flexibility, doing things differently, opportunities



## Theme 3: Can't do it alone – recognition of the need for further training and support

**Codes:** Confidence in myself, self-belief, take action, personal development, fun, expertise, caring, intuition, networking, professional development, resilience, working in isolation



## Theme 4: The Real Me – finding a professional home

**Codes:** Opportunities, opened up avenues, involve others early, recognition from others, clear direction, confidence in myself, confidence working with others, caring, motivation, taking risks, determination



#### Theme 5: The Real Me in Action - Early Involvement of others

**Codes:** Others believing in me, having fun, embracing failure, education, working strategically, networking, thinking bigger, support from others, knowledgeable practitioner, professional skills personal skills, life experience, leadership & management, perseverance



## Theme 6: Development of the Real Me - The impact of learning

Using Braun & Clarke's (2006) six phases of analysis the researcher familiarised herself with the data and then commenced to code them. As the data became familiar it became easier to code them into particular themes. Having identified & streamlined the codes sub-themes were then defined to develop major themes. These frequently encompassed codes that interlinked across the themes (Table 3).

The themes were subsequently reviewed and specifics of each theme were refined and linked to generate propositions and associations (Grbich, 1999, Braun & Clarke, 2006 in Silverman, 2011: 274-275), resulting in a logical division of themes. The themes were then reviewed to ensure that they met the overall aim of this research study.

## 3.15 Patterns & Themes

Braun & Clarke, (2006) claim that analysing data in this way will lose some depth and complexity of data, but this did not seem to detract or reduce quality of this data, in fact it generated a rich material. Each interview was read over three times and at each reading a new or more refined version of an original theme would emerge.

Once the key themes were identified the data set was reviewed again ensuring that all the key messages had been identified and noted.

#### 3.16 Themes and sub-themes

Many items of interest emerged. On occasions a single statement was made by one participant and was not raised by others, so did not fall under the premise of being a theme, nevertheless, a number of 'single participant' issues were identified and stored. However, many emergent topics and issues arose which were shared across several cases. To reduce the large number of individual comments and to make them meaningful broader categories and major themes were developed.

Some themes were obvious right away, such as 'developing confidence' while others required more consideration. Interestingly, Braun & Clarke (2006) suggest that as data is revisited it can become boring and frustrating, but the researcher would not agree as she found the whole process compelling.

To work through the various themes and sub themes to create connections a mind map was used and gradually key patterns were defined (Table 4). This gave the analysis more structure and enabled a clear link between the main theme and the sub themes.

The mind map identified key gaps in the interview material elicited such as why were participants not supported in moving their innovations forward, what were the drivers and characteristics that enabled them to pursue their innovations and what additional skills and knowledge were required to effectively get their innovations acknowledged. Therefore it was decided that the same respondents would be contacted again and further questions asked (Appendix 5) No one refused the invitation and people were generous with their time. The established process of transcription and member feedback was undertaken. This material was included in the mind map and did indeed enhance and elaborate the findings.

Table 4

Key Patterns				
Confidence	Individual	Thinking outside the box		
Resilience	Motivation	Strategy		
Knowledge	Embrace Failure	Finance		
Self -Belief	Academic Credibility	Improved patient outcomes		

## 3.17 Reflexivity and the Researcher's position

As course leader for the Post Graduate Diploma in Social Innovation on which the participants were enrolled, and knowing them as students, the researcher's position has been considered in terms of being an 'insider' or 'outsider' researcher. Both positions have implications for the reliability and validity of the research data. An insider researcher is one whose research is conducted within a population of which they are also a member, they may share an identity, language, and experiential base (Asselin, 2003). Two key advantages of this position can be an already established level of trust and the speedy establishment of communications. However, issues of preconceived assumptions about the meaning of the data or the motives of participants also need to be considered.

Conversely, an 'outsider' researcher refers to one who has little or no connection with the group studied. The advantage here is one of naivety and being uncontaminated by local knowledge; however, trust can take longer to build and subtleties within the context may go unobserved. A further position is offered by Corbin Dwyer (2009), relating to the notion of occupying 'the space between'. She argues that the researcher need not sit at one position but rather be part of each. that insider or outsider is too simplistic. This more fluid position is what was achieved here. The researcher's professional background includes having been a community health professional (District Nurse) thus having a first-hand understanding of the world that many of the participants come from. In addition, being involved with the course, the researcher can be said to have insider knowledge and some understanding of the contexts the participants operate within. But the researcher has no knowledge of undertaking such a course, or of setting up an innovative project at work. In some ways this is an ideal position as it offers sufficient familiarity of the world and work of the subjects, but with enough distance to enjoy the advantages of the naive questioner.

#### 3.18 Ethical Issues

Ethical approval was obtained from the researcher's place of employment (Appendix 7) and from the University of Westminster (Appendix 8) as it is obligatory that the research meets the requirements of both higher education institutions (Blaikie, 2011; 27). Also it was essential that the Nursing & Midwifery Code (NMC, 2015) was upheld throughout the research process. The Code acts as a framework to promote professional standards of practice and behavior in all aspects of work through practicing effectively and prioritizing people, preserving safety and promoting professionalism and trust. This was evident by ensuring that the initial focus groups had been set up in a safe environment and subsequent skype interviews were carried out in the safety of the individual participant's home. The research was structured and participants were informed of the various stages of the research process and the needs and availability of the participants were always prioritized ensuring that the trust and the professionalism of the researcher was constant throughout the research process. Pawson's (2003) TAPUPAS framework (Table 5)

was also used as a guide providing a transparent process resulting in a clear audit trail.

## 3.19 Confidentiality

In line with the NMC Code (2015) confidentiality was maintained at all times and participants were assured that their replies would remain confidential and their interviews would remain anonymous. They would be able to withdraw from the study should they wish to, without penalty. Although the questions being asked were not deemed to be sensitive, the potential of sensitive issues arising had to be considered, therefore a gentle approach was used throughout the interviewing, allowing participants to process the questions and give timely responses. One participant became emotional, not by the questions, but in answering the questions as she realised how far her personal journey had been and she was in awe of herself about what she had achieved. The interview continued with an increased passion and enthusiasm, adding to the richness of data being gathered.

#### 3.20 Informed consent

Participants were given an information sheet (Appendix 2) and a consent form (Appendix 3) to sign. The researcher took full responsibility to ensure that the participants were able to fully comprehend and interpret the content of the information sheet (Ritchie & Lewis, 2009).

Care was taken to ensure that no participating organisation could be identified and permission to use direct quotes was gained. Participants were free to say they did not want their direct quotes used, however no participant refused. Participants were also informed that any identification of unsafe practice revealed by them cannot be ignored and would have to be reported to the relevant professional body. These issues were all stated on the consent form (Appendix 3). The face to face interviews were i-recorded on a digital tape recorder and notes were also taken to aid prompting where required. The Skype interviews were all recorded directly onto the personal computer of the researcher which was password protected and always stored securely in a locked drawer.

At this stage the raw data remained the intellectual property of each participant in that they could add/delete/alter any of the data. The feedback was often humorous and many added facts that they had forgotten when first interviewed. Nothing was deleted, which was encouraging to know that the transcription had been carried out accurately. Participants were informed that the data from the face to face and Skype interviews have been stored in a safely secured password protected personal computer. All participants were informed that the transcripts would be destroyed following the successful completion and submission of the research study.

# 3.21 Quality Assurance

This section will address the data management practice followed.

## 3.21.1 The validity and reliability of the data

To ensure the validity of the data the transcribed material from each interview was sent back to the participants for verification or 'member checking' (Porter, 2007, Robson, 2011). This gave all participants the opportunity to revisit their responses and to clarify any misinterpretations perceived by the researcher. It also gave the participants ownership of their 'voice' and the opportunity to continue their involvement in the study (Braun & Clarke, 2006). Reliability can be established through transparency and openness by the researcher about their theoretical perspective and by allowing interviewees to comment on interpretations drawn from interview analysis as suggested by Newton (2010). Hammersley (1992, cited in Richie & Lewis, 2004: 276) 'argues that it is not always possible to ascertain whether the accounts are true and accurate as there is no independent, reliable access to reality'. Therefore reliance is on feedback from the participants.

#### Table 5

#### **TAPUPAS**

Transparency: Is the process of knowledge generation open to outside scrutiny?

All research evidence has adhered to quality and ethical procedures, ensuring a clear audit trail is available.

Accuracy: Are the claims made based on relevant and appropriate information?

All claims are based on relevant and appropriate information that has been acquired from participants involved in the study.

Purposivity: Are the methods used fit for purpose?

A variety of research methods were considered, however thematic analysis was deemed an appropriate method for this research study.

Utility: Are the knowledge claims appropriate to the needs of the practitioner?

The knowledge gained from the research is beneficial to all community health professionals and is transferable for all health professionals to learn from.

Propriety: Has the research been conducted ethically and legally?

All research governance and ethical processes have been followed in line with the procedures of both Universities.

Accessibility: Is the research presented in a style that is accessible to the practitioner?

The research findings will be written in a format that health practitioners will be able to understand. Participants will be sent an executive summary of the findings on completion of the research.

Specificity: Does the knowledge generated reach source-specific standards?

The knowledge generated is evidence based and can be presented at conferences where managers can be exposed to the findings and there is sufficient evidence to potentially influence future educational programmes for community health professionals.

(Proctor, 2007: 85)

A set of criteria was followed which was developed by Pawson (2003), see table 5. It not only looks at the validity of the research, but accessibility and ethics. The acronym **TAPUPAS** as described by Pawson (2003) sets out the criteria and this framework was used as a reaffirmation that accurate processes had been followed as outlined above.

## 3.22 Summary

The original plan to run a series of small focus groups across the United Kingdom proved to be more difficult than anticipated. Individual interviews at the respondents' location were considered, but due to the geographical distances this was excluded. Having been influenced by Sullivan's (2013) opinion on the success of using Skype interviews, serious thought was given to this method. Installing additional audio/video equipment to the computer meant that it would be possible to view the participants during the Skype interviews so Skype interviews were chosen as an appropriate method, which proved to be an excellent decision. Whilst no research method is without its flaws, the researcher was aware that many of the participants were known to her and to each other, and took account of Feig's (2006) view that emotions are the 'hot buttons' of behaviour and may influence behaviour and could be challenging, particularly having insight into how important improving practice was to the participants. By using Skype and face to face interviews it enabled each participant to have an equal voice and their individual responses to be scrutinised.

Since the research study commenced, a considerable amount of literature has been published on the subject of innovation in the health service and there has been renewed interest in this subject (Davis, 2011, Kendall-Raynor, 2013, Butterworth, 2013, DH, 2015, Sarhan, 2014) as a result there were opportunities to approach other community health professionals and involve them in this research study, therefore opportunistic as well as purposive samples were interviewed.

Skype interviewing has proved to be an appropriate method in acquiring valuable information and the researcher would promote the use of Skype interviewing as an effective way of gathering data.

## Chapter 4 Findings part one

"Knowing yourself is the beginning of all wisdom" (Aristotle)

#### 4.1 Introduction

This chapter reports on the material gathered from the 26 research subjects and explores the early days of their involvement with social innovation. As the aim of the research is not to evaluate a course, no distinction has been made between those participants who undertook the post graduate course outlined in an earlier chapter, and those who undertook other types of training linked to social innovation. It could be regarded that attending training was the way that they became involved with the research, becoming one of the sample, but here it simply provides a context against which their subsequent actions, attitudes and values can be understood. The changes within themselves that they express are at times seen as 'before and after' the educational experiences, but for these individuals it was putting their new knowledge, as a result of the education, into action that has appeared to have had the lasting impact.

The long and difficult journey towards becoming a socially innovative practitioner began for these participants as a slow burn stretching over months or even years. It became apparent very quickly when reading the transcripts that the participants were describing something active, not an event, but a series of processes. These included the process of implementing change within their organisations but also, perhaps more importantly, that an internal process had been ignited somewhere along the line. Their personal development had been progressive, with one event leading to another; they had needed to have gone through an experience before being able to move on, and therefore the notion of a *journey* is the first theme to surface. With this established, the other themes fit into place.

As information emerged from the data it became clear that that their journeys were diverse and wide-ranging. The fuel for the burn varied, but common themes emerged across their wide range of clinical experiences. This chapter will explore the main themes and supporting sub-themes, which underpinned the individuals' desire to step forward and make changes within their organisation for the benefit of their

patients as well as the barriers they faced in doing so. It will also identify some of the participants' personal characteristics which, in part, help drive them forward and their early experiences of working with senior management which went on to influence their later work.

## 4.1.1 Four principal themes emerged from the data:

#### Table 6

Key Themes, incorporating sub themes:

Better is possible – a perceived need to raise standards of health care delivery above and beyond current practice: - Saving time & resources, improving clinical outcomes, poor communication of risk, lack of specialist services, lack of imagination from organisations to understand the need for change, maintaining excellence in clinical practice

Swimming against the stream – the challenges faced when setting out ideas for improvement: - Organisational lack of recognising existing problems, lack of vision, frustration, self-doubt

**Can't do it alone – recognition of the need for further training and support:** - identifying a lack of budgeting/financial skills, how education influenced motivation and perseverance, confidence in own clinical skills, lack of confidence in knowing how to make change happen

The real me – finding a professional home: - Permission to think and develop the 'self', self-belief

Many of the participants reported that although not dangerous, certainly sub-optimal professional practice was witnessed. However, what was key to these participants is that it was generally not acknowledged as a problem by their organisation. They were working in environments which, in many cases, adhered to the minimum quality standards policy required and they felt that there were areas that could be improved upon to enhance the care of their patients and client groups. Many participants held the view that their organisations could work with them if they wished to implement the suggested changes, but frequently their innovations were ignored and the ideas were shelved.

Most of the participants had been in clinical practice for over twenty years and had seen numerous changes and cut backs in NHS resources during that time. Yet, they seemed to feel that against a backdrop of uncertainty with regard to provision of care and change of job roles, this uncertainty was in fact a driver demanding intervention. Rather than focusing on problems resulting from the uncertainty and instability of the

organisations, the participants saw this as an opportunity to introduce solutions towards improving services. Necessity, they really did think, was the mother of invention. As health professionals, reflective practice is an essential way of working. It is core to any health professional's practice and through reflection this group have uncovered a number of factors that have either hindered or enhanced their ability to act therapeutically with their patients. Some of their key reflective observations are outlined in the themes below. Interestingly, they did not report observing gross misconduct or high levels of risky care, but they became concerned by the unquestioned way sub-optimal routines and ineffective models of delivery went unchallenged. For some it was the opportunity to make a difference through making changes by shifting care from the more expensive secondary facilities to help closer to the patient's home.

4.2 Theme 1 - Better is possible - a perceived need to raise standards above and beyond current practice

This theme is broad and complex and consists of sub-themes which together paint a clear picture of a need to raise standards in some aspects of patient care. The themes are a mix of practical issues and entrenched attitudes, some institutional and some local. They are: 1) A perceived need to save time and resources, 2) Improving clinical outcomes, 3) Poor communication of risk, 4) Lack of Specialist Services, 5) Lack of imagination from managers in understanding the need for change, 6) maintaining experience in clinical practice.

#### 4.2 .1 A perceived need to save time and resources

Nationally, set against the ongoing economic crisis within the NHS, there is a demand for increased local efficiency whilst at the same time reducing clinical posts. It is particularly noticeable when looking at integration across services within local communities. This is a challenge against a background of staff cuts, rising caseloads and less time for patient need. Service changes, high staff sickness rates, vacancies left unfilled and uncertainty surrounding reduction in posts was causing low staff morale and lack of motivation within community teams. Unlike the hospital setting where teams, which are usually inter-professional, are predominantly located on one ward, the community teams were dispersed and not always located physically in the

same building which created additional difficulties in promoting a positive working environment. The occupational therapist, dietitian and specialist nurses participating in the study were particularly isolated as their core team usually consisted of themselves (specialist nurses) and in the case of the occupational therapist and the dietitian there was maybe one other team member who in most instances was their manager.

Throughout this turmoil and uncertainty, many participants were witnessing inappropriate use of resources. One such example provided by a participant was that of a duplication of services from the District Nursing Service and the Multiple Sclerosis (MS) Specialist Nurse, where both were visiting patients and duplicating advice (sometimes conflicting). The participant did not feel that this was a good experience for the patient as it frequently left them questioning which advice they should adhere to. The Specialist Nurse (participant 8) developed a plan whereby the Specialist team visited patients weekly and would only involve the District Nursing service if their skill set was required. This resulted in a reduced case-load for the District Nursing service and an improved patient experience.

"This small change in service delivery has totally transformed the provision of care for MS care closer to home, there has been enhanced patient satisfaction and it reduced costs substantially. It enhanced the MS nurse role and the GP's were extremely satisfied" (Specialist nurse, participant 8)

The core community teams, such as District Nurses and Health Visitors were under pressure to deliver more for less so they had to begin to look at how services could continue to be delivered without compromising patient care. This required thinking differently and looking at what services they had and how to use them more effectively. District Nurses in particular were being asked to take on new patients, increasing their caseload on a daily basis. Unlike in a ward, where staff can say 'there are no more beds', when caring for patients in the home it is not an issue about bed occupancy, but the staff resources available to visit them at home. The District Nurse participants were recognising that their caseloads were beyond full capacity and were reaching unsafe levels of practice. This presented ethical dilemmas as they did not have any policy whereby they could 'close their books' so had to take on the additional patients. It was at this point that they began to consider

delivering their services more effectively, without compromising quality of service provision.

A further example is given by a Community Matron (participant 20) who wanted to improve the referral experience for patients by consolidating existing points of access into a unified point of access.

"I want to stop referrals being made to clinicians whilst they are visiting other patients as it is not always sent to the appropriate clinician and it is not professional practice to be receiving such calls whilst visiting patients. Currently duplicate referrals are being made to different services within Primary Care which causes confusion and also wastes time" (participant 20)

It was evident that no additional resources were going to be added to their workforce to ease this burden, but quality of care had to be maintained so something within the current structure had to develop.

# 4.2.2 Improving clinical outcomes

The participants also wanted to address what is necessary in terms of outcomes that are meaningful to patients and their communities. Amidst the confusion of service provision and limited staff resources it was imperative that the patient remained at the centre of any decision being made. The primary motivation expressed by all the participants was the desire to improve clinical outcomes as well as the patient experience. Some of the participants recognised that their client group were disadvantaged by the very nature of their physical disability, such as having a learning disability or physically not being able to access a leg ulcer clinic. These participants who were at the front line of service delivery were at the receiving end of anecdotal statements such as "Why are there so many people visiting me asking the same questions?" or "Why can't I be seen at a clinic near me?" Whilst this was not concrete evidence it gave the participants the opportunity to investigate some of these concerns.

One such example was where patients suffering from epilepsy were waiting months to see a neurologist and during this time their anxiety levels would increase. A Specialist Nurse (participant 16) recognised this and set up a nurse led clinic

alongside the consultant neurologist outpatient clinic. This reduced waiting lists, prevented a duplication of assessments and also meant that she could seek advice from the neurologist if it was needed.

"Since developing this service we have seen a drop in accident and emergency admissions and the neurologists accept our expertise and have noted that their clinics are dealing with the more complex neurological conditions now. The patients are less anxious as they are seen quicker and their medications can be amended accordingly as I am a Nurse Prescriber also".

A Specialist Respiratory Nurse (participant 17) working in a rural location implied that it was impractical for her to travel long distances for potentially a ten minute consultation. However her caseload consisted of mainly people over the age of 70 years who were unable to travel to her monthly clinic. She spoke with the local bus company and spoke to her GPs and commissioners and her project has been funded for 12 months. The bus company will pick up patients at agreed stops and bring them to the clinic and bring them back home again. She commented:

"I am already seeing improved clinical outcomes with my patients because they arrive at my clinic relaxed, not breathless, and also energised as they have been socialising on the bus coming in"

#### 4.2.3 Poor communication of risk

The sharing of knowledge is essential to the success of any organisation and with reference to the NHS and other patient facing organisations sharing of knowledge will also impact on patient care. Sharing of best practice is encouraged across all sectors within the NHS but despite this expectation the participants were not experiencing this. They found that senior managers, in many cases, were not aware of some of the resource conditions which resulted in hazardous consequences for care standards. Frequently community trusts are managed by non-clinicians who do not fully comprehend the complexities and skills required when visiting patients in their homes. One example of this is where a band 4 health care assistant had been allocated to visit patients that should have been seen by a qualified district nurse who would have had the relevant skills to assess patients at risk. The District Nurse (participant 1) who was concerned about this said:

"We have no team leader so the manager of the community service is overseeing our team, she only seems interested in looking at numbers of patients visited and not which member of the team visits the individual patients or what medical condition the patient has" (participant 1)

Working as an autonomous lone worker can have its risks as practitioners may not always be part of the communication chain and on occasions referrals will go direct to the manager and not the clinical teams. However all the participants were very aware of the Queen's Nurse Campaign: 'Right nurse, Right skills', so when they allocate staff to visit patients this knowledge can prevent potential problems. It seemed to these participants that managers frequently were not in a position to defuse difficult situations as they arose, due to the seeming lack of leadership within some of these organisations. In addition, at times, there was little recognition around the feelings of stress and negativity that was being felt by their employees as a result of being under managed.

This perceived lack of risk awareness, had a knock on effect for any staff member who voiced concerns or promoted a new way of working. A number of these participants had wanted to take the next step and develop their service, but encountered weak management support and, most importantly, felt they had no one to turn to when the pressure was on, so frequently their ideas had to be put on hold. When communication about risk was entered into with senior staff, participants were, at times, met with criticism rather than support.

"The organisation I worked for was very risk averse — I was offered another job elsewhere so I jumped ship and I feel that the people I am working with now accept the fact that I don't take things lying down any more. I am inclined to quietly reflect and then quietly professionally challenge what I am being asked to do and why. I have been able to negotiate some things to say 'I don't think this suggestion is a good one, I would rather we did this based on the experiences I have had in the past"

#### 4.2.4 Lack of Specialist Services

One specialist wound care nurse explained that she was not supported to provide a leg ulcer clinic to both North and South areas of her borough. This left patients in the

South missing out on a weekly leg ulcer clinic as they were, due to their condition, unable to travel by public transport to the North of the area and they were unable to afford a taxi, therefore their care was compromised with the potential of worsening their leg ulcer. The lack of parity across services due to staff reductions and geography caused worry and concern for the specialist nurse, she said:

"No one knows what is happening behind closed doors (the patient's home) if there is not quality monitoring and communication between services across locations"

Similar examples were given by other participants claiming that interruption of services meant that a patient could be without any contact (other than emergency call out services) particularly between the hours of 18.00 and 08.00 leaving a gap in access to services, particularly relevant to specialist services. This interruption of services leads to unsafe practice within organisations and poor patient experience, which was particularly evident in end of life situations. Patients may have requested to die at home but often, particularly towards the last weeks of life, if no community support is available the patient will be admitted to hospital against their wishes. A participant who is a learning disability nurse (participant 18) working with patients receiving end of life care, felt very strongly about this gap of service provision as it is stressful and frightening being at this stage of an illness, but it is magnified if either the patient or a family member has a learning disability.

"Despite patients' preferred place of care being at home there are still an unacceptable number of unnecessary out of hours admissions to hospice or hospital of patients at end of life who have expressed a wish to die at home, this makes them even more distressed" (Specialist Nurse)

4.2.5 Lack of imagination from organisations to understand the need for change

A difficulty often encountered was lack of vision from within their employer organisation. Each participant highlighted an aspect of their work that could have been enhanced if people were able to look beyond the 'normal' service delivery. Parts of the NHS in particular were experienced by these care workers as entrenched within old systems and processes making creating and sustaining any vision harder to achieve. Three of the participants recognised that they were constrained and valuable time was taken away from enabling them to use their

creative skills so looked outside the NHS where there were more opportunities to be creative, interestingly, instilling a love of their work once again. Two set up a social enterprise together providing a unique lymphedema service and another set up a social enterprise providing support for looked after children. Both social enterprises continue to thrive and have drawn on other creative innovative practitioners to join their workforce.

"It wasn't just about doing a service differently, it is also about us as leaders still being clinical and leading a social enterprise. So, we have introduced yoga for patients, and that was through negotiating with public health and having funding through a service level agreement, so far the feedback has been good" (Lymphedema Specialist Nurse)

#### The health visitor said:

"I am now working the way holistic care can be given – I work in an integrated service where the children are supported by the most appropriate professional"

Innovative practitioners, by the very nature of looking at doing things differently, are proactive when looking at solutions to improving service delivery. Many found that the reactionary approach of large organisations such as the NHS was stifling and left them feeling devalued and frustrated at the lack of vision and short sightedness. Observing poor management and lack of recognition that delivering a service in a different way, yet in line with the organisation's vision, would enhance service and inevitably patient satisfaction was a major concern for the participants. One such example came from families who were able to access the NF1 support service which was introduced by one of the participants after recognising that many NF1 patients and families were feeling isolated; a number of comments were made from families who said their lives were improved through communicating with families who have children with the same condition.

#### 4.2.6 Maintaining excellence in clinical practice

In contrast to large organisations such as the NHS, which can at times adopt a mindset of achieving day-to-day mundane (but necessary) targets, these participants were striving for excellence. However, they were witnessing complacency and, occasionally, professional jealousy by colleagues who were suspicious of anyone challenging the status quo. Yet many felt that they had no choice but to work with colleagues who displayed characteristics of apathy and social acceptance, lacking in enthusiasm by just wanting to 'get the job done'.

"I try to get the team on board to share my vision, but this is not always possible, so I have to work to the best of my ability with those who are willing to work differently" (District Nurse)

4.3 Theme 2 -Swimming against the stream - the challenges faced when setting out to improve clinical practice and patient experience

This section is composed of a group of sub-themes which describe the different challenges, professional and personal, faced by these 'would be' innovators when they tried to implement their ideas. While the themes identify separate issues they are characterised by a sense of doing protracted battle with established practice and finally the impact of the struggle on the participants. The sub-themes are:

1)Organisational lack of recognition of the existence of a problem, 2) Lack of vision to try for better, 3) Frustration – this is how we do things around here, and 4) Self-doubt – have I really understood the problem?

## 4.3.1 Organisational lack of recognising existing problems

The majority of the participants said that they observed a lack of insight by the organisations to consider the need for change to happen. There was an acceptance of current practice which did not consider any change of service delivery being required. The participants had the relevant knowledge and skills that could be called upon to implement such a change, but their experience and skill set was, in general, not acknowledged by their organisations. This was commonly thought to be because the organisations had not received any complaint about their service so they did not see the need to do things differently. One example was that given by a School Nurse who was writing a book on sexual health which was written in terminology frequently used by teenagers. She had also planned workshops in the format of competitions, which enabled teenagers to speak freely in a safe environment where they would not be intimidated or judged. Local clubs or halls offered their space to run the

workshops, which is quite different to the traditional school nursing service delivery where the School Nurse visits the school and offers 'drop in' sessions. The School Nurse said that:

"In general, I don't think that people understand enough about contraception and sexual health, either as healthcare professionals, teachers or students – and they should to maintain their health as not everything is covered by the school teachers and this could be supportive material for the school nursing service."

Her employer organisation did not view her project as a public health initiative and claimed that her project was already being covered in schools, they did not accept that it was aimed at those marginalised groups who often have dropped out of education so would not be getting this information. Unfortunately the book has not been approved as a resource by the NHS so an educational minister from the Caribbean, who was known to the School Nurse, was contacted and her book is now used as a resource in the Caribbean and comments to date have proved that it has been accepted by the school and students positively. The workshops have not been implemented at the moment as the staff will need training in facilitating such workshops.

At least fifty percent of participants said that the reason their organisations were not supportive was because they were unsure of what the expected outcomes of the proposed initiatives would be and in some cases they did not understand the rationale for implementing a particular idea. The proposed outcome or goal for the school nursing initiative mentioned above would be to reduce the amount of sexually transmitted diseases and also reduce teenage pregnancies and promote wellbeing in adolescents. However, the organisations were, in her view, short sighted and did not have a long term vision of the benefits of such a service. A way of monitoring this would be to develop a database of adolescents acquiring a sexually transmitted disease (STD) and also look at the number of unplanned teenage pregnancies that had occurred over a period of time (eg,12/24 months) and subsequently pilot the project and monitor these same conditions over a further period of time. She has suggested this to her manager and at the time of the research interview this was being considered. She was pleased at this response as she feels that if the statistics

show a high rate of pregnancies and STDs then her book may be considered at a later date.

## 4.3.2 Lack of vision – to try for better

Lack of vision on the part of their employer was claimed to have been experienced by many of the participants, particularly those working in larger organisations such as the NHS. However, an example of the desired visionary thinking was related by a Specialist Nurse who works with patients suffering with prostate cancer. She explained that her service was unable to deliver the government targets of offering an initial out-patient clinic within 2 weeks of diagnosis because they were still seeing patients who had been on the system for years as they would have out-patient appointments 'for life'. Fortunately, she worked with a department with vision and together they developed a remote follow up service which she said was the first of its kind in the UK. This then released appointment times for newly diagnosed patients to be seen. This service was also evaluated positively by patients receiving this service and it is hoped that the service will be viewed as 'gold standard' treatment for patients with prostate cancer and will be replicated across the UK.

The prostate cancer telephone follow up provision and support systems for NF1 sufferers are examples of how ideas can become a reality if there is agreement from line managers. In instances where the lead of the organisation supports the vision this can make the operational functioning of a new service easier. However, not all clinical areas are fortunate enough to have such support.

The Occupational Therapist (participant 14), involved in this study had been trying to implement a pro-active falls service targeting people who were potentially at risk of a fall. He made the following statement:

'Most patients that have a fall will spend at least one night in hospital and many do not recover to previous functional level which means a longer stay in hospital. The local ambulance service has reported a 3 fold increase in falls over the past few years.'

His idea was to deliver a comprehensive falls prevention service, utilising a Falls Prevention Champion, to deliver an effective but brief one-stop holistic assessment

and treatment for anyone referred into the project in order to prevent falls in the community. Patients, families or carers, health, partner agencies or adult social care staff, with the patient's consent, would be able to refer to the team for advice, assessment of needs, provision of equipment or walking aids and property risk assessment.

The comment above highlights how a small change to service provision could reduce hospital admissions and provide a holistic approach to care. He spoke of an 80 year old lady who had purchased a walking stick and a shopping trolley independently as she was not aware of any services that could assist her. She subsequently fell because the walking stick was not the correct height for her, resulting in her fracturing her hip and spending five weeks in hospital. Over £25,000 was spent on ambulance service, hospital admission, a five weeks stay, surgery, community follow up, social support, GP visits, daily living equipment and medication. Such lack of vision within organisations has caused frustration amongst the insightful participants who could envisage a more cost effective patient centred service.

However, much of the reason for the 'lack of vision' some participants acknowledged may be due to the NHS processes that are set up to safeguard patients within the NHS. For example, in clinical practice, if an individual makes changes without consulting the senior management team then they are often singled out and their change may frequently be blocked. This may come across to some as bullying as this District Nurse said "the NHS has a culture of work place bullying and blocking of innovation". The 'blame culture' has been referred to by at least seventy percent of the participants and it was the fuel that for some ignited their frustration. In some cases this actually resulted in one of the participants being put on a disciplinary charge and another leaving the service.

"At all times the patient's safety was paramount, but because I acted as an autonomous practitioner and made a decision that was not in line with the organisation's policy I was witch hunted and a disciplinary was made against me and I was suspended on full pay!" (participant 6)

Such actions can be short sighted leaving an already under resourced service to deliver care, potentially resulting in sub-optimal care. This particular organisation went months with one member of staff down impacting on the delivery of the service

and patient experience. It also resulted in a very strong willed practitioner becoming more determined that her decision was correct so a battle of wills commenced. The organisation was not successful in upholding the disciplinary claim and the participant returned to work, but promptly resigned and has now set up a new service in an area where there are more visionary managers. In instances there is evidence of a 'lack of long term thinking by management' and participants stated that in their view there was a 'blame culture' amongst the senior members of their organisation which was something that they could hide behind when confronted with their decision.

## 4.3.3 Frustration – this is how we do things around here

Frustration was frequently mentioned by the participants and this was mainly due to working with people who, in their view, were content to remain working at the pace they were familiar with and not sharing the passion to the extent that the participants did. This is not to say that the service provision was not of an acceptable standard, but opportunities were missed with no consequences or even a desire to revisit these missed opportunities. It is at this time that reflection should be practised, however staff are frequently only focussing on the task in hand and not taking a step back to reflect if things could be done differently. In support of this statement those participants working in the NHS continued to state that they were working in a bureaucratic, top-down culture which constantly frustrated them. A learning disability nurse (participant 19) made the following comment about her organisation saying:

"They are unable to recognise that an add on service could improve patient care dramatically"

She wanted to introduce a befriending service to patients with a learning disability which would increase their social interaction, promote their independence and potentially reduce hospital admissions. Often these patients end up at accident and emergency because they are lonely and want to communicate with someone, not necessarily having a medical ailment.

This is not a new phenomenon in the NHS as accident and emergency units are frequently visited by patients who may be lonely or who have a mental health condition. However no service has been provided to care for them, so in order to

gain attention or to seek help they will turn up at the accident and emergency, which is very often inappropriate, and staff do not always have the expert skills to care for them.

4.3.4 Self- Doubt - do I really understand the problem?

Whilst the majority of participants displayed a substantial amount of confidence and determination a significant minority talked of self-doubt when they observed what changes and innovations were happening around them and seeing what others had achieved. These particular participants stated that they were 'out of their comfort zone'. One of the Health Visitors said:

"I lack experience to expand services and don't think I have the correct skill set to put my ideas into action"

Individuals entering the caring profession do not necessarily commence their career with thoughts of changing service provision, in many instances the underlying reason for training as a nurse or an allied health professional is to care for patients/clients. However, as they progress in their career their confidence grows and for some they have the ability to make changes; others find it difficult and in the words of a Mental Health Nurse:

"My own fears to try something different stops me, I am frequently faced with challenges and I know in my heart what should be done, but I don't always believe in myself"

The majority of the participants were over 45 years and had been trained on average for twenty years. Training in those days was delivered through facts and case studies where critical thinking and enquiry was not encouraged so therefore when they began to critique a service their self-doubt would surface. On the other hand this group of participants had witnessed and worked with health professionals who had been trained to think critically and to ask questions and recognised that they had knowledge of their clinical areas and so they also gradually began to question practice. Working in a caring profession is unlike other careers as these professionals are having a direct influence on patient and client care and a

miscalculation could have life threatening consequences. This may be a further reason for the self-doubt that some participants displayed.

Those participants who displayed greater confidence and self-assurance were very happy to take a calculated risk; this was particularly evident in services where they had pro-active ideas. One example of this was an idea that was developed by a District Nurse who wanted to integrate health, social care and voluntary sector services to reduce the risk of inappropriate hospital admissions. In doing so it is anticipated that this would improve co-ordination of quality care, increase the dignity and well-being of older people and consequently reduce the demand on community services, in particular District Nursing. The idea was supported by the local clinical commissioning group and a local national branch of a charity funded the project. This resulted in some initial challenges around the organisational change with the development of a new integrated role within an established workforce, in particular the District Nursing workforce. However, at the time of the interviews the pilot scheme, which had identified 100 patients who had met the criteria for the service, had been in force for twelve months and the challenges had been overcome with the hospital admission rate for the over 75 year group being reduced by 50%.

This particular example was pertinent to this study as the District Nurse who implemented this pilot had been a student on the programme that was delivered by the researcher. She had commenced the programme at a time when she was at a crossroads in her career and she was frustrated with the service being delivered to the over 75 year group. She also doubted her ability to do anything to change the service. However, attending a programme where she was able to share her frustrations and her vision with other like-minded people added to her belief that something needed to be done to enhance this service delivery. She commented that one of the lectures on the programme posed the question 'What will your workforce look like in nine months' time .... We need to break down the barriers between health and social care to focus upon integration for any changes to be successful'. This statement had resonated with her and she said:

"At that moment I recognised the opportunity I had and what a privilege it would be to make a difference through making a change within my community setting"

This was the turning point for a District Nurse (participant 1) who had started her journey with little confidence and an abundance of self-doubt. She went on to say that:

"On a personal level I have discovered that stripping away all things I have no control over has enabled me to direct all my energy in setting clear examples for senior managers and commissioners of quick win-wins in relation to achieving small goals and positive outcomes with patient care that we can celebrate"

4.4 Theme 3 - Can't do it alone - recognition of the need for further training and support

Lifelong learning is a concept that is encouraged within nursing and allied health professional careers and the use of evidence based practice is evolutionary, therefore it is generally looked upon as an opportunity to advance knowledge in their professional field, which in some instances may also open the door to a career promotion. However, the participants in this study had acquired knowledge and advanced educational training within their specialist field of practice, but without exception, all participants identified the need for even further training of a different kind if they were to get their projects off the ground. The themes identified are: 1) Identifying a lack of skill around finance, 2) How education influences motivation and persistence, 3) Confidence in their own clinical skills, 4) Lack of confidence in knowing how to make changes happen.

#### 4.4.1 Identifying a lack of budgeting/financial skills

For example, to have a greater understanding of financing a new project and the various financial elements that would make it successful was a need recognised by the participants. Economics and finance is not an area that is covered in nursing or allied health professional training so there was an identified gap and recognition for further training and support in this area. As the NHS has become more target driven over recent years the health professionals have had to recognise the significance that finance plays across the roles of all health professionals. Every service being provided has to account for the revenue it is bringing in and the expenditure required in delivering the service. In previous years this was the responsibility of the

manager, but frequently this responsibility is now passed to front-line staff. It is essential, therefore, for anyone developing a project to be able to present a business plan that will reflect the benefits and outcomes of a project. Whilst this caused a certain amount of anxiety it also gave the participants 'ownership' of their idea which fuelled their enthusiasm, along with some anxiety. A school nurse (participant 3) felt that she had some understanding of project finances as she had worked with publishers to write and produce a book, but what she couldn't do was publicise the use of her book effectively and raise the money required.

"I can work with balance sheets and finance I just need to know how to bring the money in and get people interested in my project" (participant 3)

This was a very different aspect of financial acumen which sits alongside marketing an idea and was new to her. It is an illustration of how working in isolation is not the best option when trying to generate an interest in an idea or project. Another example is that of the two specialist nurses (participants 11 & 12) who set up their own social enterprise in order to improve the lives of patients suffering with lymphedema. They provided specialised care to these patients, providing clinical expertise, health promotion and yoga classes to improve the well-being of patients. They understood their market but wanted to know how to generate an interest in their service and make the wider GP population and hospice services aware of their service provision. Their initial starting point was to work with the hospice and Macmillan service to introduce their new service. Their expert provision became widely known and they are now receiving referrals direct from the GPs, releasing space in the hospice for others. Support and understanding from the organisation of the employer, or organisations utilising a service is essential to move any idea or project forward at a reasonable pace.

## 4.4.2 How education influenced motivation and perseverance

These participants, it appears, also developed a keen sense of perseverance over this period helping them learn new business processes which in turn gave them a new perspective and understanding of the importance of being business savvy as well as just having a good idea. The new skills resulted in growing confidence which in turn, many claimed, made them more determined in their aim and willing to take

on new challenges or new aspects to their roles. For example, they now understand the importance and relevance of preparing a business plan before making any changes within service delivery.

"If you have the determination and the will to do something don't let others squash you. Don't let them take ideas from you. If you can't get there using one route use another.... But don't give up" (participant 26, a mental health nurse)

The feeling of being 'done to' rather than being part of the decision making process was felt by a large percentage of the participants and an occupational therapist said:

"Just because somebody says 'No' don't take no for an answer.......find another person who you can influence ....Over the past nine months I have been told 'No' lots of times when I talk about the change I want to implement, but I will look at new ways...I will not give up"

The NHS is going through many changes and challenges at the current time and it is impossible for staff not to be affected by this. These are times of uncertainty and low morale is rife amongst staff yet despite this a learning disability nurse said:

"There is a lot of flatness amongst practitioners and people lose motivation. People are just going through the motions because they feel that there are no opportunities or they don't feel supported. The post graduate programme gave us these tools to sort of take responsibility for our own actions and be accountable, it keeps me motivated. So no matter what hurdles or challenges that you have to face..... I have become more motivated, and happier to persevere and keep pressing on and that helps practice definitely"

Upon reflection of their practice, a number of the respondents recognised that they had courage and tenacity and a resilience and perseverance within themselves that they were not aware of. A dietitian commented:

"The whole idea of seeing ideas becoming a reality rather than just a dream increased my confidence and I realised that the determination to do this was worth it"

At least sixty percent of the participants said that prior to coming onto their chosen programme they would possibly have given up on an idea if it was rejected.

Perseverance is now central to their attitude when they contemplate any changes.

"Everything starts and ends with me now and this has been a big, big improvement in my working life – I take responsibility for any changes and persevere if I am faced with resistance" (Specialist Nurse)

Another respondent said:

"When I go into a boardroom now I am not at all intimidated – I am persistent in my approach and I am respected by the CEO and senior managers – I rock that boardroom now!"(Learning Disability Nurse)

When faced with a negative response about an idea the responses were unanimous, they would persist but they needed to understand the reasoning behind the 'No' response. It is due to their sheer determination and perseverance that this understanding can be found. It takes time to challenge, question and investigate other angles and to come up with potential solutions.

Discussions with the participants showed that they now dealt with perseverance in positive ways and that a much more subtle approach was taken to look at the reasons why an idea was being rejected. The midwife participating in the study said that, if rejected, she would ask:

"Why are they doing that, where is that coming from?"

Using this tactic has prevented her from taking any comments personally and she now tries to see the reasons from the other persons' perspective, particularly from the perspective of her manager, suggesting that they 'might be frightened of where your comments are going or where your thoughts are going'.

At the time that the research was being carried out the participants were open and transparent about their feelings. As the responses began to unfold the concept of perseverance began to emerge. The participants were of the opinion that the programme they had attended had helped them to recognise this inner drive which they had not previously been conscious of.

"I think what the course really made me realise was that there are lots of tactics that people around you will use to try and distract you, or pull you away, or you know that sort of thing. And you really can't be disillusioned and you can't let them get to you. You just have to keep your... true to yourself really" (Community Matron)

#### 4.4.3 Confidence in their own clinical skills

The difficulties mentioned above can be contrasted with the confidence that the participants had in their own clinical skills. Many of the participants were over the age of 45 years so they also brought to the table life skills which are not measured by attending a course or a programme, but draw on the experiences they have had to date which makes them the person they are today.

One participant, who was over the age of 60 years, said that her life experience had provided a 'richness' to what she could contribute to both the programme and to her clinical practice. Another Specialist Nurse said:

"My life skills have enhanced my professional practice and the experiences I have had along the way have made me the person I am and I am richer for this"

While self-belief and confidence was present in most of the participants, it was having had the opportunity to share their thoughts and frustrations with like-minded people that enabled them to pursue their ambitions. Many of the participants made comments about a renewed sense of 'knowing their own potential'. One of the Health Visitor participants said:

"The programme has given credence and confidence to what I do. I now believe in myself as a change merchant, and to recognise that ok, I can be whacky but equally that's actually sometimes how the best ideas start".

Similar comments made in support of growing confidence identified by the participants:

"I am much more focused on goals and aims that I am seeking to achieve. Also, it has made me realise that when you have an idea that ... you really go for that Idea. Actually to follow through can be very hard and quite often things will get kicked into the long grass but you just have to keep on going" (Specialist Nurse)

## 4.4.4 Lack of confidence in knowing how to make changes happen

While there was a general consensus among participants that their ideas were innovative, creative and would save their organisations money, as a result of previous unsuccessful attempts to get a project off the ground there were feelings of uncertainty all round. Participants were unsure who to approach for constructive advice so often went in circles for a while waiting for the right person to talk their ideas through.

"I'm the sort of person that I have ideas I can get to the preliminary stage of things and then I find it quite difficult to move to that next stage of getting people to really take interest in real terms, in a strategic way. People are very nice at a low level way, and are very complimentary and think things have a value to them for patient use, but in my personal experience I'm not good at taking it to that next stag" (Specialist Nurse)

"I can think big, but I cannot act big because of the restrictions to my role". (Occupational Therapist).

One key factor that helped many was the opportunity to network with well-informed others and also to hear prominent speakers at conferences; the participants gained and refined their ideas and were given advice on how to approach senior members of their organisation to get a project noticed. The support enabled them to work through any feelings of inadequacy and any past experiences that were stifling their creativity.

"I felt that I was working in a straightjacket as my ideas were constantly being rejected without any rationale......once I was able to bounce these frustrations onto a colleague I began to see clearly" (District Nurse).

Due to conflict within the workplace, confidence was compromised and feelings of inadequacy were creeping in for some of the participants, but again the support network helped them through.

"Talking to someone who has been through the same experience as I have confirms that I know what I am doing and they know that I always have the patient's welfare at

the core of everything I do and other people do not have the vision that I have, so I say to myself go on do it your way" (District Nurse)

Networking was not an aspect that initially came to mind when looking at how to make change happen, but it ranked highly as an essential part of supporting this change. Sixteen of the participants confirmed that they would contact someone they had met on the programme to seek advice.

"....One of the main areas was meeting up with people in similar situations doing similar things, trying to make a difference...and sometimes if there are queries there or we need to run something past someone we will email each other for support" (Occupational Therapist)

As has already been mentioned, not all the participants had supportive work environments where they could express ideas freely so the programmes provided a venue for that. For those who were able to speak freely without judgement they were able to work through any fears with key work colleagues. Interestingly, after they were some way into the course, some found that once they were honest and opened up about how they felt, they gained support from managers and other professional colleagues. On occasions some claimed to 'take emotional risks' to be heard and the determination and resilience that they saw in others inspired them to keep going.

#### 4.5 Theme 4 – The real me - finding a professional home

A change of perspective with regard to the participants' self-perception was a key finding. Indeed, the emergence of the construct of 'the real me' surfaced for many after they had started their post qualifying programme. The culture of the programmes provided the opportunity to share their experiences in a safe environment with other like-minded people without being judged. This was a sharp contrast to their working life where, if an idea had failed, many of them felt judged and frequently did not attempt to try again. They reported some profound personal changes as they worked through their programme. It is noteworthy that it was not long into their programme before some shifts in perception were evident, suggesting that being amongst people who had similar traits and beliefs had given them the confidence to voice their concerns or misgivings about what they were witnessing in

practice. The sub-themes here are: 1) Permission to think and develop the 'self', 2) Self- belief – beyond work,

# 4.5.1 Permission to think and develop the 'self'

For many working in healthcare it is like living in the fast lane, this is particularly so for staff working in an organisation such as the NHS where all their time was taken up executing their duties leaving little time or energy to think laterally. Therefore, one of the key findings was recognising that while they knew they had the skills and knowledge to carry out an initiative they needed protected time to reflect and to visualise what 'success' might look like. Many said the programmes allowed them to give themselves 'permission to think'. In the words of a dietitian it gave time to 'value myself and accept the past, overcoming my self-doubt'. Other participants said that it also enabled them the opportunity to 'work through fear'.

Arguably it is not common practice to be seen actively 'thinking' as nursing and other allied health professionals are renowned for being 'doers', therefore this change in culture took some time to adjust to without a feeling of guilt for some. A District Nurse said:

"Why give others the power to spend your time – use your time effectively and efficiently"

She was a self-confessed 'deviant' but she had a 'can do' attitude and sought out positive minded people to work with. So with protected space and a safe environment, considerable constructive thinking went on.

A Specialist Nurse said that she puts a 'Do Not Disturb' sign on her door as this is her 'thinking' time. Everyone at the Practice where she works accepts this and when she surfaces they ask her, 'What bright idea have you come up with this time?' The acknowledgement that this is good use of time has given her the opportunity to bring new ideas to the practice and the patients and staff are seeing the benefit of this.

## 4.5.2 Self-belief – beyond work

One common theme that emerged from the data was the strengthening of their selfbelief. Comments were made from both younger and older participants:

"I didn't realise that I was capable of doing this"

"I feel people are seeing the real me for the first time"

One of the younger learning disability nurses said:

"I don't have the fear I had before" (participant 18)

This eight word phrase sums up the difficulties that many of the participants had been faced with but were now able to deal with; this is mainly due to the support of colleagues and the inner confidence that they discovered whilst on their journey. As the midwife in the study claimed:

"There is a lot of change happening for me and I have lots of ideas swimming in my head. Once upon a time I would have thought that my career is over. Now I know that if my job goes I am good at my job and I will be able to get another"

It had been obvious from comments made that many of the participants did not feel they belonged within their organisation as they were frequently referred to as being a 'difficult' member of the team, frequently ostracised when decisions were being made. This resulted for some in a weakening of their self-esteem and a feeling of not having a professional home. However, once on their programme, knowing that they are not alone in their way of thinking gave them the confidence required to continue with their quest to introduce new ideas to their work place.

'Learning to believe in yourself' was mentioned in various ways by several participants as a District Nurse (participant 2) claimed:

"I think I have a rosy future whichever way it goes because I believe in myself" In contrast, another District Nurse (participant 5) said that:

"To believe in yourself is the hardest thing"

She was working in a team of very experienced District Nurses who were uncompromising in how they provided care to patients in their homes, thereby disabling her belief in herself as she frequently suggested alternative ways of providing care. She remained working in the team until she saw an opportunity to work for a more visionary team of heart specialist nurses and she joined their team where she felt she could apply some of her ideas for improving patient satisfaction and care for people with long term conditions.

Many of the participants were finding it difficult to be themselves and fit in with the corporate image and, in many instances bureaucratic systems. However, on occasions when a manager spent time and listened to their ideas, as a learning disability nurse said: "a chink of light appeared" and all was possible.

Some of the participants said that the self-belief that they discovered went beyond their working life into their personal life. For example, one of the School Nurses (participant 25) had been approached to set up a Parish Nurse Service, which would involve the integration of nursing with the beliefs of her religious community. This is a relatively new concept in the UK, originating from the USA. It involves physical and spiritual health (of any denomination) through working with the 'whole' person and the wider community. At the time of being asked about this she did not have much knowledge of parish nursing, but she accepted the offer and said:

"So this is another exciting endeavour and I believe I can do this"

One of the factors that helped participants to recognise their own individual skills was the Myers-Briggs Type Indicator (MBTI) Personality Test that they were asked to participate in during their studies on the Post Graduate Certificate in Social Innovation and also on the NHS Leadership programme. The general consensus was that it helped them to understand themselves, but the added bonus was that it also helped them understand how others reacted and worked. This improved both their working and personal life because they were able to recognise some of the traits that they had learnt about in the psychometric testing. One of the self-confessed shy participants commented:

"Knowing my own characteristic – ISTJ (Introversion, Sensing, thinking, Judgement)..... acknowledging that not only extroverts will get things done,

introverts will go about it in a different way, yet as effective, but they must be seen, therefore relationships must be built"

Another comment made by a Health Visitor (participant 13) suggests that the approach and sensitivity used when working with patients/clients is guided by the values and characteristics displayed by the professional. Frequently she deals with families and children at risk and she reiterates the importance of transparency, sensitivity and professional attitudes when working with this group, she says:

'People don't buy what you do, but why you do something and therefore values are taken into consideration' (participant 13)

# 4.6 Summary of findings of the first stage of the journey

This chapter covers the first part of the participants' journey towards becoming an innovative practitioner within their area of healthcare. A wealth of material was elicited from the interviews showing complex sets of perceptions about the experience; some were what might be expected, others more surprising. The common themes of the observation of sub-optimal care or simply a lack of vision and poor communication acted as drivers for these individuals and were consistent with findings from the literature review which have shown that the support of managers is essential if change is to happen. The literature has also shown that post qualifying education stirs up further questioning and practitioners are more confident to challenge and are ready to face obstacles experienced when suggesting a service improvement where there is a lack of recognition of need, or the admission of a problem with service delivery and, in their view, unwarranted contentment with adequate care. However, some of the personal themes that emerged were unexpected. While appearing confident in their professional skills and knowledge and with many years of experience behind them, it was surprising to hear of the degree of self-doubt and lack of confidence that emerged when they began to organise themselves for action. Interestingly, although very real, the self-doubt did not stop these professionals from pursuing their goals. Some felt irrationally daunted and out of their comfort zone but gradually this feeling disappeared and an inner confidence surfaced as they spent more time on their specific programme. The selfbelief and the knowledge of their specialist areas prevailed, whilst also the crucial

acknowledgment was made that further support and education was required particularly in the field of planning for and managing the financial aspects of their projects.

Two of the less expected themes were that when they started the course a sense of belonging occurred and the feeling that they were discovering the 'real me'. Being able to express their views in a safe environment was a powerful factor in their shifting self-perception, almost as if their ideas had been locked away and not allowed to be released. A linked theme, which was clear from several interviews, was the concern that they had been seen by others in their employing organisation, and to some extent saw themselves, as 'outsiders', and as 'deviant' in some way. However, attendance on their course enabled them to spend time talking with likeminded people and over time these negative perceptions disappeared.

The next chapter continues their journey as it looks at the full impact of the training experience as they gain skills and develop as confident innovative professionals.

## Chapter 5 Findings part two

The findings in the previous chapter looked at the first part of the participants' journey beginning with their recognition of a need for better healthcare delivery in their professional area; the difficulties they encountered during their initial efforts to improve practice; the recognition that they could not succeed unaided and finally, for a significant number, the discovery of a new 'self', a real me. Although the participants had very different working environments, clinical interests and experience, issues emerged from the data so frequently that they became themes. They included varying levels of personal and professional confidence, frustrations with managements' responsiveness and a sometimes belated acknowledgement of the limitations of the knowledge and skills required to successfully become a healthcare innovator with a view to improving care delivery.

# 5.1 The Journey part two: the complex road to becoming an innovator

This chapter will look at the impact of participation in the post graduate course, the NHS leadership course and training days attended by the participants. It will identify the additional skills they developed and how they enhanced their effectiveness when becoming an innovative professional.

This second part of the journey looks at how these courses broadened their knowledge and skills base, reinvigorated their capacity to 'learn' and also how it contributed to their personal development. On joining their respective course the participants were suddenly with like-minded people who were able to support their views but would also challenge them constructively - in contrast to the lack of support many claimed they experienced in their work place. Being with like-minded, innovative and creative people and working in a focused way on their own project had a profound impact on many of the participants. The results are reported here in two sections 1) the 'real me' post education, and 2) the impact of learning

#### 5.2 Theme 5 – The Real Me in Action – Early involvement of others

In the previous chapter the theme of the 'real me' which identified subtle changes in the thinking of the participants, had been influenced by some key factors which helped the participants find themselves: having permission to think creatively, being supported by like-minded people and, most importantly, not being the outsider or 'difficult' member of their team – always wanting to do things differently. This resulted in the strengthening of aspects of their character for many participants. In some instances participants themselves have stated that they did not recognise themselves, never believing that they would be capable of making some of the difficult decisions and organisational changes they did. In making these changes their revised perspective on how they approached their work had a major impact resulting in these changes.

#### Table 7

Key themes incorporating sub themes:

The Real Me in Action - Early involvement of others: No man (or woman) is an island, seeking support from further afield, working differently & change of perspective, increased confidence in own professional judgement, personal development

**Development of the 'Real Me' & The Impact of Learning:** individual strengths, listening to others, skills in finance, business plans, negotiating, working strategically, seizing opportunities – finding new courage, not frightened of failure

#### 5.2.1 No man (or woman) is an Island

Many community health professionals are lone workers, they work independently, often in isolation, which is the nature of working as a community health care professional. However, some have now expressed an interest in working differently in future; in particular through involving others in the early days of their plans for caring for their clients/patients. Though frequently visiting patients and clients on their own, these community health professionals acknowledge that they are part of a bigger community team and they will consider involving other members of their team in the provision of care, as lone working may not now meet their patients more complex needs.

One early finding reported was that participants shifted their view and developed a new understanding of the advantages of making more effort to remain part of a multidisciplinary team, to bring others with them, and not to isolate themselves even in the face of indifference to their plans for better practice.

Recognising that there are people in their organisation who could potentially help them promote their idea prompted several participants to involve selected colleagues. A mental health practitioner (participant 4) stated:

"I would try and get more influential people on board sooner rather than later ....... but it is hard to find people in the NHS with a long term outlook"

A Specialist Nurse (participant 8) had similar views:

"Get buy-in early on. Find an influential person in the organisation who can support and fight your corner"

Interestingly, it was apparent that the rationale for involving others was not necessarily about requiring more money, but about having influential people on their side with their greater and different experience and access to senior figures that several concluded could assist in driving their ideas forward. Alongside these influential people specific departments also needed to be brought on board. The participants recognised that it would be favourable if particular departments or individuals working in these departments expressed an interest in their ideas. Business and finance departments were acknowledged as being significant allies in their drive to push an idea forward as without the support of these areas within an organisation it would be difficult to proceed with any new initiative.

It was not only within their employer organisation that help was sought; some who had successfully set up their own social enterprises already saw the advantage and clearly stated that they would 'enlist the support from other social enterprises/Community Interest Companies to help with the social return on investment' (participant 13).

This particular Health visitor had set up a small specialist service in an Accident and Emergency department and she had been clear from the onset that the expected outcomes of this service was to improve the care and service provision for looked

after children in her geographical area. She recognised that she would have to involve other agencies such as Social Services, together with the General Practitioners, from the onset if this service was to be successful. The early involvement of these agencies did in fact prove to be beneficial in the overall social and clinical success of the service provision.

However, not all the participants had acknowledged the importance of involving other professionals early in implementation of their idea. They had to learn this fact later in the process. For example, a participant working as a specialist wound care nurse (participant 9) across a number of general practices said that she would have brought colleagues on board much sooner had she known how supportive they would turn out to be:

"I would have approached the GPs much sooner as they have funded the pilot with little question,.....they have already referred most of the current caseload"

The only occupational therapist involved in the research stressed that it was important not to try to accomplish the project without the support of their immediate manager. He had initially struggled to get support as his project did not meet the manager's aspirations for the department that year so he adapted his plans to meet part of the manager's plans for the next year.

# 5.2.2 Seeking support from further afield

In addition to eliciting the support of colleagues with an interest in their innovation, seeking support from a coach or a mentor early on in the process was perceived to be beneficial. This was particularly relevant for the health care professionals who were new to promoting their ideas and were aware of how much they lacked confidence or knowledge about the process. Those participants who had access to a mentor or coach from outside their organisation claimed that it gave them the opportunity to discuss their ideas objectively.

"It was so liberating to be able to speak to someone who was not going to judge me by my pecking order within the organisation ....... was able to speak freely without the fear that my job might be in jeopardy" (participant 19) However, for at least one participant, the relationship with the mentor was more important than whether they were part of the organisation or not. A learning disability nurse was mentored by a manager in her organisation who initially was against changing anything within the service.

"Initially I was really concerned because this particular manager was really against me making any changes to the learning disability service and he was initially very obstructive when I went over his head to share my idea with the Chief Executive of the organisation" (participant 18)

Nevertheless, once they were able to remove themselves from the business of the clinical environment and were able to meet over a casual cup of coffee she was able to lay out what the outcomes of her innovation were to be and how they could be achieved with minimum extra input from staff. It was a matter of doing things differently which in the initial stages of conversation the manager had not appreciated. Her anxieties were overcome when she found that she was able to seek support from the manager who was open to listening to her ideas and he was also able to assist her in developing strategies to put her idea into action. This particular learning disability nurse was aiming to improve the end of life care for families with a learning disability.

#### 5.2.3 Working Differently

Alongside bringing others in to their ideas early on in the cycle of their innovation, participants also claimed that they were working differently in other ways. A significant number acknowledged that their perspective had changed in recognising alternative ways of approaching their clinical practice and working across the sector. Prior to attending their course, many of the participants felt that their idea could, and should, only be implemented 'their way'. Several shared that, at times, when their opinion was challenged with an alternative perspective on their identified area of poor practice from another interested professional, they felt that they were being undermined and their confidence floundered. The dietician in the research study provides an example:

"I wanted to review the uptake of prescriptions for supplement drinks by interviewing my patients, but my manager suggested that I review the prescription data from surgery which would give me quantitative data for that surgery" (participant 2)

She went on to say that she felt that this would be too small a sample and it wouldn't give her the complete picture. However, she took the advice and realised that this was a much more effective way of collecting data as she was now able to obtain data from all of the surgeries in her geographical area and it gave her a clearer idea of who was using supplement drinks.

However, this struggle with self-doubt and a feeling of being almost 'got at' was managed and receded in the light of a positive outcome leading from the other person's advice. For many of the participants the recognition that working differently and having the ability to look at ideas and suggestions from another person's point of view actually enhanced their way of working was a big step forward in their development. It enabled them to keep their idea central, but to also take on suggestions from people who have experience in working on projects at a larger scale. Appreciation of these additional skills has been slow at times as some of the participants felt challenged and threatened until recognising that there are different ways of delivering a better service. A Community Matron comments that listening to the views of other professionals has helped her to develop her one stop service for patients in the community:

"I was managing an integrated service and always felt that I had the ability to see things from other professionals' view point, but when we eventually sat down and looked at how I was delegating workload I realised that I was still looking at things from my professional point of view and had not considered the physiotherapist or pharmacist's point of view". (participant 20)

# 5.2.4 Increased Confidence in own professional judgement

The example above of the learning disability nurse who went over the head of the manager and spoke directly to the Chief Executive of the Trust displays how her confidence increased the more obstruction was put in her way. She was determined to share her vision with someone who would initially listen and then hopefully support her:

"I never thought I had it in me to go over the head of my manager, but I had fire in my belly and I had to be heard – I knew my idea would benefit all learning disability patients/clients, whether they are the patient or a family member" (participant 18)

The theme of increased confidence was displayed by the majority of the participants. Comments such as these were common:

"I now have confidence in my instincts ....I no longer think I am not clever enough and I protect my ideas and don't let others think they can do better than me with my ideas" (participant 6)

One of the District Nurse participants said:

"I am not going to beat myself up ... I can do that...Nothing is impossible now – there is potential everywhere and there is not a final 'no" (participant 1)

One of the specialist nurses recognised that her confidence had increased but she still felt stressed about whether she was doing her job properly. Since attending one of the educational programmes on innovation she realises that this does not mean she lacks confidence. She now realises that stress is how she deals with situations:

"Stress can be productive for me but it doesn't mean that I am not confident in what I am doing" (participant 24)

From the quotes above and revisiting the interviews it was apparent that all participants were aware of their expert knowledge in their specific area and, importantly, their clinical expertise was never doubted by others. It is interesting, therefore, to consider why there were so many references to self-doubt within the data. It seemed to manifest itself when the participants were required to move beyond their comfortable, well-practised professional routine with patients, to being required to present their knowledge to an audience of colleagues and senior managers. Not only did they have to show their own capabilities, but to state that they had the insight and courage to identify areas of weak practice within their immediate professional environment, even if it meant implied criticism of close colleagues and line managers, took significant courage. This would test most people's self-confidence so it seemed that it was not only the struggle itself, but the battle with authority that strengthened their resolve.

# 5.2.5 Personal Development

Whilst the participants did not elaborate on how their increased professional confidence had transferred into their personal life, a number said that they now felt more confident in all walks of life as a result of putting their additional education and training into practice. They related incidences of a stronger self being demonstrated in a variety of ways, one being that having a sense of fun has shown a different perspective in dealing with stress at work.

"The sense of fun, working with others and testing things out ... that tremendous sense of fun.... The fun had a profound impact on me both personally and professionally" (participant 20)

This statement was made by a very experienced Community Matron who was one of the older participants and she had made reference to how she did not take herself so seriously anymore and that sharing her experiences with other like-minded people had enabled her to laugh at herself when things went wrong, without feeling a failure.

The following comment was made by a District Nurse who spends much of their time dealing with sensitive issues, such as working with patients at the end of life and patients and families who have life threatening conditions.

"Laughter is a tonic in the right context.......a lot of our work can be quite challenging and demanding but I think a sense of humour and ability publicly to say 'I didn't quite get this right' you know 'I got it wrong this time, but hey-ho let's keep trying'. That sense of being real, and being a person, and being more alongside the patients and the public in the first line I think is what the course has really enabled me to have" (participant 5)

The comment below sums up the feelings mentioned and implied by the majority of the participants:

"My whole life has changed in a way because of my own personal development"

## 5.3 Summary of the 'the real me' in action

In summary, these participants reported several developments within themselves as a result of engaging with the whole social innovation experience; not just the education but the struggle in operationalising what they had learnt. Key lessons learned by these participants has shown that involvement of others supports improved team working and introduces others to the positive impact innovation has on improving service delivery for their patient/client group.

Recognising individual strengths has promoted an increased confidence and growth in personal development enabling these participants to realise that fun is an important element of work, it de- stresses and promotes a culture of openness and honesty resulting in a happier environment to work in.

Acknowledging that other people have a valid point of view and also have the patient's best interest at heart has enabled the participants to view things from other perspectives and also respect other people's opinion. Taking into account the responses from the research it is evident that additional benefits have been gained from attending either the post graduate course on social innovation, the NHS leadership course or training days extending beyond just working on the actual innovative project.

## 5.4 Theme 6 – The Development of the Real Me & the Impact of Learning

This next set of sub themes relates to the time when the participants moved on from their respective course and were actively advancing their project. How the course broadened the participants' knowledge and skills base first in a practical sense, then enabling them to better conceptualise their project's needs, and how they managed significant others with whom they had to work, had all been part of their learning. Themes identified ranged from quite straightforward new knowledge, better planning, taking a strategic view and using the confidence gained with the new knowledge to be bolder in their actions. The theme of confidence, which was discussed in the previous chapter, can be seen here too but is underpinning other factors as opposed to being an actual theme itself. The themes related to the impact of learning were:

Skills gained – finance, business plans, negotiating

- Working strategically
- Seizing opportunities
- Not frightened of failure

# 5.4.1 Skills in finance, business plans, negotiating

Many of the participants stated that prior to attending their course on innovation their knowledge of finance and writing business plans was limited or non-existent. Many were unsure of whom to seek financial assistance from and had been under the impression that their finance department was rather unapproachable. With hindsight, they realised that writing a technically sound and convincing business plan was essential before a financial department would consider a new initiative

Negotiating and seeking funding for projects was a completely new experience for the majority of the participants. They described themselves as being so wrapped up and single minded while focussing on the need for the change in practice and the hoped for outcome for their patients, that they had not realised the importance of producing a plan of action before they could implement their idea. This they later acknowledged to be in part due to their practice of working alone and not sharing with colleagues. This resulted in disappointment for some, with two participants commenting that their inexperience had held them back. One of the Specialist Nurses said:

"Through my inexperience the outcomes for my project were not fully established and I had not put measures in place to implement the change" (participant 7)

However, she did learn from this experience and represented her business plan with clear expected outcomes, costings of the service and expected savings. She realised that what was required was a clear rationale and a solution. She had set up a support service for families of children who were suffering from a rare cancer and she wanted to connect families nationally. Initially she introduced the idea by presenting the quantitative data, but she had not worked through the solution and how this support would be met. When it was first rejected she was disillusioned, but then went on to seek advice from outside of the NHS. In search of technical help, she made contact and negotiated with a dating agency and using their format she

applied the same concept to connecting families who had a child with NF1 (rare childhood cancer). This concept has now spread into Europe and families are using this network as a means of support, thus saving time visiting hospital and releasing time for the consultants and specialist teams to concentrate on new patients.

Another participant stated that her idea of reviewing District Nurses' caseloads was initially rejected because she was unable to articulate what the purpose of her project was. She said:

"If I had clearer aims and objectives ......an audit tool and evaluation of project it would demonstrate the success of such a project and also involve patient feedback" (participant 1)

However, having the project rejected did not dissuade her from her goal. She improved her business plan and presented clear aims and objectives. Although it took two years to get funding for the project, with some stakeholders losing interest along the way, it is now up and running. She claimed that even though she was very disappointed when it was originally rejected, she has learnt from this setback and as a result she has now been approached by the Trust to lead this project throughout her Trust.

A further example is that of one of the Learning Disability Nurses who said that she was disappointed with herself when she realised that she had not really understood the financial incentives that drive organisations to either accept or reject an idea. She was developing a care pathway for patients who had repeat admissions which may have been due to reasons such as falls, chest infections, urinary tract infections, etc. She said:

"I didn't really understand that money was an incentive for the organisation – if I had understood this I would have focussed on just one specific reason for admission and this would have been less daunting and manageable" (participant 19)

Later, knowing what was required of her she condensed her business plan, focussing only on patients who were admitted to hospital with a urinary tract infection and at the time of the interview she was waiting for a response from the trust.

## 5.4.2 Working strategically

While the development of a business plan was acknowledged as key, taking a strategic view was just as important. Without exception, all of the participants commented that when they thought of their idea they were focussing on their individual area of work. Whilst they were confident that their ideas would improve the service provision for patients/clients they worked with, they had not necessarily thought beyond this to the rest of the service.

# One of the Specialist Nurses said:

"My project has now been recognised by NICE (National Institute of Clinical Excellence) as a demonstration of good practice. It has been published and presented widely and is replicated in many areas" (participant 8)

She was looking at care for patients with a neurological condition and identified where there was duplication of service which was distressing for the patient and an ineffective use of resources. She identified the skill mix and expertise required and represented her plan to the Trust. As the Trust was employing a number of the agencies who had the responsibility of visiting these patients it has led to a significant saving by reducing the number of unnecessary visits to these patients. The patient is happier as there is more continuity of care and the Trust can demonstrate cost savings of £300K per annum.

Not all of the projects have had such a national profile, but many of the participants have commented that they now look to see how their idea can be implemented further afield than their 'patch'.

### 5.4.3 Seizing opportunities – finding new courage

Throughout the interviews the participants repeatedly spoke about the realisation of the importance of networking with other professional and patient groups. To have an informed audience to bounce ideas off they said had been invaluable. Yet it was not just the networking that was important, but the opportunity this, at times, gave them to seize new opportunities. Communicating with like-minded people inevitably meant that there could be some synergy in their thinking. This was reported by many participants and the experience of two District Nurses provides an illustration. They

were both working in the Community caring for patients who had lymphedema. On identifying a gap in their patients' care, they decided to put their ideas, energy and joint expertise together into developing a social enterprise in order to develop a lymphedema service for all patients, not just cancer patients, who have lymphedema. They left the community trust they were working in and set up their own social enterprise whereby they were commissioned by the clinical commissioning group (CCG) to provide a lymphedema service. They provide a service across their region and they have since also commenced yoga for patients suffering with lymphedema. This has seen an increase in patient's well-being and is also being commissioned by the CCG. These clinicians are both now highly successful entrepreneurs and would not have taken this leap of faith if they had not had the confidence that their educational programmes had instilled into them.

"Whilst on my course I learnt that if I want change to happen then I have to make it happen and grab every opportunity that comes my way" (participant 12)

Another example of seizing opportunities that required some courage is the example of a Specialist Nurse whose Chief Executive Officer (CEO) was visiting her clinical practice area. When leaving to visit another practice she offered to drive the CEO there. Whilst in the car she seized the opportunity to share her idea with the CEO and they met the following morning and the specialist nurse presented her project idea to her and she was given the funding for the project. This project which was mentioned previously, was addressing duplication of services to patients with a neurological condition.

# 5.4.4 Not frightened of failure

A key theme that was mentioned by many of the participants referred to the importance of learning from mistakes. Learning to cope with failure was seen as a major learning curve for many of the participants. Prior to commencing their relevant programmes many said that if an idea was rejected or was unsuccessful they would have given up. They would not have persevered and may have become disillusioned. The education programmes they attended inspired them to carry on and improve their original idea. Numerous comments were made about coping, or not, with failure.

A District Nurse who, in her own words, always worked within the rules and never did anything that had not been done before said:

"Now I am not afraid to try new things and to turn negative into positive" (participant 5)

Another participant who has had numerous setbacks in her career, commented:

'Since being on the course I now feel alright about failing, previously if my idea was rejected or failed I would have put it to one side and start from scratch with another idea' (participant 2)

A Community Matron said:

"It doesn't matter how out of sync your ideas are – go for it, explore it, test it. It doesn't matter if it fails. Failure is not a problem ....Yes out of three ideas one might work, but if that one works it is worth it, and so it's very much about embracing failure" (participant 20)

One of the Specialist Nurses (participant 10) claimed that: "Failure has served me well" because she learned from it.

These comments have highlighted how embracing failure has enabled the participants to move forward and learn from failure. They see this as a learning opportunity and therefore are more inclined to pursue an activity and wait for the outcome. If it fails they will ask themselves 'why' and 'how can I do it differently next time'. This is a major advancement in the way many of the participants now practice.

5.5 Summary of 'the real me' & the impact of learning

The impact of learning is multi-faceted as can be seen from the range of themes above. One does take a key place however, that of rejecting the solo worker image as their only, or preferred, way of working and involving others in their ideas and plans. Importantly, their learning and the opportunity to put their new knowledge into action has strengthened their inner self and positively impacted on their personal lives. Within their working environment many reported that they now think strategically, seizing opportunities and learning from any setbacks along the way. Key to all the themes that surfaced from the interviews can be seen a spirit of

personal determination that underpins the attitude of the majority of the participants and is reinforced by a comment made by a health visitor: *'Watch and wait'*.

The next chapter presents four case studies, two deemed to be successful and two which failed to materialise. The aim is to bring to life the work of these community healthcare professionals and to illustrate how many of the themes outlined above manifested themselves in real life.

## Chapter 6 Innovation in Action

The last chapter identified key themes seen through the participants' working lives as a result of undertaking training and, most importantly, through putting their learning into action. This chapter first provides an overview of all of the projects that the participants were involved in which are set out as a table (Appendix 1). It provides details of those projects that were successful and those that were not. It details the approximate cost of each project when first set up and the anticipated, or actual outcome, in terms of service improvement and money saved.

## 6.1 Summary of the chart information

The chart (Appendix 1) shows the wide range of diverse projects undertaken. They address the perceived needs in the patient experience across the lifespan, those with chronic illness and those vulnerable to addictive behaviours. When looking at the outcomes of these initiatives what is immediately noticeable is the substantial amount of money saved. Even the more modest ones generated a saving of at least double their start-up cost while several tripled or quadrupled the outlay making substantial, sustainable annual savings for their organisations. This is without considering the clinical and social benefits to the patients. In broad terms, these savings were most often achieved by reducing the need for hospital admission as a result of better care in the community. Timely intervention saved many hundreds of pounds annually. Some projects aimed to improve access to care and so delay or prevent the onset of complications. Others offered a better level of care and a few focused on raising awareness of types of medical conditions or behaviours. It is noteworthy and may be worthy of further investigation at another time, that the initiatives which focused on processes that were already in existence in some format, and offered a better version of an existing service tended to succeed, while those which tried to introduce new ideas or services extra to what was in existence, often fared less well from a point of view of initial funding or management support. These observations are offered by way of context only, this research does not aim to evaluate the projects but it is helpful to get a picture of the real world circumstances the participants worked within and the goals that drove them.

6.2 Four Case Studies - two successful ones and two which did not meet their original goals

Two successful and two unsuccessful projects undertaken by these respondents will be looked at in depth. It will explore each project's aims, structure, process and any other factors that played a role in its success or failure. It is approximately two years at the time of writing since the original interviews took place and all four participants were contacted to ask how the projects had progressing. The names of the participants have changed to maintain anonymity.

6.3 Case One: Patient Support Network (successful initiative)

The first project to look at is one that was developed by a Specialist Nurse (Freya) she worked with children who had been diagnosed with Neurofibromatosis Type 1 (NF1). It is a rare genetic condition which predisposes patients to the development of nerve tumours. It has a range of manifestations: tumours on eye nerves, rare nerve sheath cancers, bones which when broken are difficult to heal, benign tumours which can cause an altered appearance and learning difficulties such as autism. It affects I in 4560 people in the United Kingdom (Evans et al 2010), with approximately 11,267 individuals with NF1 in England today (Ferner 2010).

This condition has well-being implications beyond the medical. People with NF1 also report feelings of isolation and frequently ask their health professionals to help them "meet somebody like me". Korf & Rubenstein (2005) acknowledge that keeping a record of all the medical visits, tests results and clinical decisions can be difficult and frequently results in fragmented care, making continuity of care and implementing support groups difficult. This is recognised as being a potential problem and therefore the project developed has been as a response to requests by attendees at an NF1 public and patient involvement day. Many people with this condition do not know anybody else with NF1 and report feelings of isolation. They wanted to meet somebody with similar manifestations of the condition who shared similar experiences as themselves. They asked to meet others affected by NF1 to provide support at diagnosis, give advice on treatment centres and treatment options, share experiences of navigating the school education system and be a "listening ear."

As this request for peer support appears to be echoed in NF1 clinics across the country on a regular basis this was supporting evidence to proceed with developing a support network for patients and their families. The nationally commissioned complex NF1 service constantly encounters people who have accessed treatments late, have very little understanding on the management and implications of their condition, thereby demonstrating the need for increased education and support. In particular to provide help early on in the individual's journey with this condition.

### 6.3.1 Setting up a patient networking project

To alleviate this need, an "introductory service" was established which encouraged people affected by NF1 to send in details of how they are personally affected. It was developed in three stages and this is one of the reasons that the project was so successful as the Specialist Nurse who developed it wanted to test the hypothesis that people with NF1 want to meet others and share experiences and also to trial the use of the internet as a means of facilitating a support network. In doing this she was also raising awareness of NF1 amongst people not known to the Nationally Commissioned service. She identified an already well used and respected website which is the UKs biggest parenting website offering local information, expert parenting advice and friendly support. It was piloted on this national parenting website the week following the first international NF1 day in May, 2015 when NF1 awareness was at its highest and parents were encouraged to post their experiences of living with NF1. Those accessing the website were encouraged to ask questions to a panel of professionals in two online clinics. The answers could be viewed on line by all members of the parenting organisation as a means of creating an increased understanding of this condition, this site continues to host a NF1 awareness week. It acts as a platform that has raised the awareness of NF1 and in doing so stigma associated with this condition is reduced as people now have some understanding of the condition of NF1.

Responses from participants who accessed the online clinics clearly demonstrated that this initiative was a success and some of the comments made by users of the on line service are demonstrated below:

"Hi, I also have neurofibromatosis type 1. I am so glad this has been posted on here as I have never met anyone else apart from my father (now passed away) who has this condition."

"----I feel we really need advice on how to broach the disorder in an age appropriate and loving way that we can develop and expand as he gets older .... Help!"

".....I just wanted to say hello to any other families with NF1"

These few testimonials are just some of the responses which have all shown that the online network has been a supportive tool for families of NF1 sufferers.

Due to the success of the pilot, it appeared that a web-based approach was an appropriate medium through which to offer support to such a diverse population as the hypotheses proved that people do want to meet others affected by this condition. The access to the on-line clinics also proved that there is a gap in the market for effective on line professional support and the internet was a valid medium for facilitating this.

The second stage of the project was then implemented. A support network was set up which was referred to as the National Friends 1<sup>st</sup> (NF1) support network which facilitated the introduction of people with similar manifestations of NF1. To take this to the next stage the project lead contacted an online dating website to discuss how to set up a similar service for children and families with NF1. Having sought advice and spoken to the IT department within her Trust a national support network website was developed to enable people to register their details and request an introduction to somebody with a similar aspect of the condition enabling the computer to make appropriate "matchings". It had password protected areas, so people could meet online in a safe, secure and private area within the website. It then matched people with similar aspects of the condition and facilitated the introduction. The network could also be accessed by post, telephone and website to ensure equitable access.

For this project to succeed a business case had to be presented to her Trust where she presented the cost of this service to be delivered over a period of 12 months. Freya calculated the cost based on the table below and kept the cost to a minimum in the hope of getting funded for the project:

Table 8

A band 7 nurse for 6 hours per week	£6,500
A band 3 clerical assistant for 16 hours per week	£7,500
Professional & legal advice	£1,000
Marketing & Administrative support	£2,500
Building & Managing a website	£2,500
Total	£20,000

Throughout the development of this project a national renowned NF1 Consultant was a constant source of support. Additional financial support was provided by a national charity and therefore with the agreement of the Trust and the national charity the NF1 support network was implemented and this second stage of the project has continued. It has spread beyond the UK as Freya was invited to speak at a conference in Europe about the internet support service for NF1 patients and families. This initiative was welcomed by European NF1 health providers and this service has now extended to Europe so patients and families are able to communicate with other NF1 sufferers across the UK and Europe also.

The third stage of the project was to find further financial support to develop hand held electronic records. The Department of Health (2012) recognises that one of the biggest challenges facing the National Health Service (NHS) is the increasing prevalence of long term conditions, and they observed that a lack of coordination of care is a major issue for both patients and professionals (Ayling, 2012). Korf & Rubenstein (2005) acknowledged that keeping a record of all the medical visits, test results and clinical decisions can be difficult and they recommended that the coordination of care for people with NF1 is for the patient or a nominated family member to keep all the records in one place, so that they can be taken to each appointment with every professional involved in their care, which would mean that the consulting doctor had all of the up to date information available. To do this a reliable technological system seemed the most likely aid. Freya visited a company that has developed an electronic patient record system facilitating the gathering of information into one secure password protected web-based location. It allows face to face discussions with patients via a secure online video link which is already built into the NHS secure network. Information can be shared with clinicians via any internet-connected computer or hand-held device. This enables those patients and families who live far from specialist centres to be afforded the same opportunities as

those who lived near the specialist clinics. This was funded for a year trial period by the Trust where Freya was employed. Unfortunately the funding for the hand held systems has not yet been approved as the Trust and the NF1 organisation are unable to find a sponsor who will support this. The costings were calculated at approximately £72,000 so it is an investment that will require a secure benefactor.

This project was regarded by her managers as innovative and forward thinking, they recognised Freya's ability to draw parallels with other systems beyond the NHS that could be amended and developed to promote communication via the support network system. The success of this project was due in a major part to her application of her learning, for example the importance of careful planning, having a clear vision of what she wanted to achieve, and provided evidence supporting her observations. Alongside this it was essential to have the ongoing support from senior clinicians who had the vision and belief in the Specialist Nurse to enable the project to be developed and to succeed.

6.4 Case Two: End of Life Care - Learning Disabilities & Families (successful initiative)

The second successful project to be discussed is the development of a bereavement service which focusses on improving the care that people with learning disabilities receive at the end of their lives. The hypothesis that people with learning disabilities are less likely than non-disabled people to be given a choice or to be able to communicate their wishes in relation to end of life care was the driving force for this project. Yet the Department of Health (2008) has recommended that effective end of life care must be delivered in such a way as to ensure that an individual has a 'good death'. This may mean being pain free, dying in familiar surroundings or being in the company of people they know and love. This is particularly significant for people with a learning disability as unfamiliar surroundings frequently cause them more distress. However, the majority of deaths usually take place in hospitals although statistically this is the least preferred place of death (Gomes et al, 2011). Celia had been inspired by a quote made by the famous Dame Cicely Saunders, founder of the modern hospice movement. 'How people die remains in the memory of those who live on'. She wanted to promote this philosophy across her Trust so that all patients and families at the end of their life would benefit from a positive end of life

experience. With this desire to improve the end of life care for her client group Celia met with the Chief Executive to discuss this. It had not been easy to arrange a meeting with the CEO, but Celia's resilience enabled her to find ways to cross boundaries and management structures, eventually meeting with the CEO to present her project.

The National End of Life Care Programme (2011) recommends that deaths should be in a person's usual place of residence. The project lead for this programme (Celia) attempted to find out how many people with learning disability had died in hospital, but the hospital notes and death certificates did not code their learning disability making it difficult to give accurate numbers. Nevertheless she did acquire information which stated that in her borough there was an average of 1%-2% of deaths of people with a learning disability. This translated to approximately 62 deaths per year, of which 60% died in hospital, 20% in care homes and 20% in their own homes (National End of Life Care Intelligence Network, 2012). It was interesting to note that the local hospice had not reported on people with learning disability being admitted or dying in the hospice. This she thought, indicated that there may be inequitable access to hospice care for people with a learning disability.

### 6.4.1 Stages of Development for End of Life Care in Learning Disability Sector

The success of this project was due to the planned stages of development. The development and implementation of the project was devised into five sections. Firstly, Celia carried out scoping exercises and gathered data to support the need for such a service presenting a sound rationale for implementing this. Having acquired this information regarding place of death, Celia went on to interview learning disability staff employed by the trust and the local council. Without exception all staff interviewed claimed that they did not engage with planning end of life care and subsequently access to specialist palliative care services was often patchy and poor. Learning disability staff commented that they were unfamiliar and uncomfortable with providing end of life care, but having been questioned about this many said that they would benefit from additional training in this area of practice.

Secondly, the project lead recognised that there was a significant gap in training for this group of nurses so in order to address this deficit of knowledge she contacted the Macmillan Nursing service. In collaboration with the Macmillan Nursing service a training programme was developed to educate learning disability nurses and allied health professionals in end of life care for adults with learning disabilities. This was a two-pronged training programme as the Macmillan nursing service required an increased awareness of the needs of people with a learning disability and they required training in communicating with this group of people, alongside this the Macmillan staff delivered training in end of life care to the learning disability staff.

The third part of this project fell in line with the Department of Health's (2009) vision to promote the development of care pathways, so a detailed care pathway was designed clearly mapping out the support that can be offered by learning disability services to people with a learning disability. This care pathway began from the point of advancing disease up to the death of a patient and followed with bereavement support. An advance care planning booklet – 'My Future Plan' was also developed which facilitated end of life care discussions between staff and patients and families.

The fourth part of this project involved Celia setting up an End of Life Care task and finish group in the Learning Disability Service. The terms of reference of this group included the development of a consistent approach to quality end of life care support across the Adult Learning Disability Directorate.

Due to the success of the additional training in end of life care to learning disability staff the fifth part of the project resulted in Celia completing a scoping exercise of care homes with learning disability registration with a view to extending the learning disability awareness and end of life care to the staff in the care homes.

The business model presented to the Trust required an identification of the requirements to deliver this project for 12 months. It involved End of Life care trainers, workshops on enhanced communication skills training for 6 nurses, communication development worker, a band 5 nurse one day per week, end of life spiritual care training for 2 nurses, marketing and printing. The costings for the project totalled £10,500 which was fully funded by the Trust. The table below outlines the details of the project:

#### Table 9

A band 5 nurse 1 day per week for 20 weeks £4,000
---

Communication Development Worker	£190
Conference Room Hire	£300.
End of Life Care Trainers	£600.
Refreshments	£200.
Enhanced communication skills training for 6 nurses	£2,874
End of life spiritual care training for 2 nurses	£300.
Printing design & printing –My Future Plan	£2,036
Total	£10,500

Having calculated that this could potentially save up to £68,000 per year by preventing admissions and hospital deaths per year the Trust were willing to fund this for an initial period of 12 months. This project has proved successful so the Trust are willing to support this project beyond the twelve months and is now embedded in the Trust vision. Celia now sits on a steering group within her trust board with a task to roll out an end of life vision for the whole trust.

Not only had Celia identified an inequity of service for people with learning disabilities she has now highlighted this plight so that it is part of the Trust mandate meaning that people with a learning disability will no longer be invisible. The success of this project has been confirmed by staff who have undergone the training and some of their comments are noted below:

"I am far more comfortable supporting people to start talking about death and dying"

"I have gained so much awareness of the needs of people with a learning disability and I can use these additional skills in every aspect of my work and personal life"

In the last financial year, two training sessions with GPs in the area have been funded by a company commissioned to do GP training. In January 2017 a local hospice funded a training day for professionals in palliative care and for professionals working in learning disabilities. Celia co-delivered the training with a palliative care educator. As recently as February 2017 a second hospice funded a training event for Celia to train district nurses, hospice staff & care home staff regarding excellence in End of Life care for people with Learning disabilities. The beneficial outcomes to the local trust has seen that far fewer patients with learning disabilities are being admitted to hospital as they are being cared for in their own homes or care homes by palliative care staff, district nurses and care home staff who have developed communication skills in working with this client group. Actual cost benefits have not yet been calculated, but the initial business case projected £68,000

could be saved in 12 months, an assumption therefore is that significant savings have actually been made as the project continues to be supported by the trust. Celia's learning around a need for sound planning and seeking cooperation with others was operationalised and delivered success.

Both projects mentioned have been led by inspiring and visionary practitioners, whilst the NF1 project had ongoing support from the onset, the End of Life project did not initially have support as Celia was unable to bring it to the attention of influential people within the Trust. Nevertheless, once she had used tactics to get her voice heard she eventually gained support from senior management in her trust and the project is ongoing and is now an ongoing agenda on the Trust agenda, so should Freya leave this particular Trust it is gratifying to know that her legacy will live on.

The next two projects discussed unfortunately were not as successful as the practitioners had hoped. As mentioned in the introduction to this chapter, it is interesting that these are both products rather than a service improvement.

6.5 Case Three: The development of an alcohol consumption awareness game for the over 65s (unsuccessful initiative)

This project was a game board that had been developed for the over 65year population. It was developed by a District Nurse (Evelyn) who was working as a Public Health practitioner and had observed first-hand the way alcohol consumption for some patients crept up on them without their apparent concern or even awareness. She could also see the impact of alcohol dependence both socially and medically. Her aim was to provide a way of informing her patient group of the signs of dependence with the aim of their reducing their alcohol intake. The game provided a light hearted and fun way of highlighting the issues of alcohol consumption in older people and was not developed as a blame tool, but rather a tool that raised awareness of the potential dangers of drinking in the over 65 year age group. Evelyn had compiled a business case outlining the cost of alcohol related admissions to hospital for this age group. Her research had calculated that every minute alcohol related problems cost the UK economy £48,000. Every hour more than 100 people are admitted to hospital in England and Wales with an alcohol related accident or condition and every day 40 people die as a result of alcohol in

England and Wales (IAS, 2015). The percentage of men and women drinking more than the weekly recommended limits has risen by 60% in men and 100% in women (IAS, 2015). Evelyn carried out a scoping exercise with Community Pharmacists. Road Safety Partnership and Drug & Alcohol Services and it affirmed a gap in an alcohol awareness method for older people in her geographical region. Her role as a Public Health practitioner gave her access to day clubs, warden controlled accommodation and other communities where older people meet. She felt confident that she had enough representation of this older population to engage this group in a brief intervention which 'sowed the seed' for them to alter or improve their lifestyles in relation to healthy living. Evelyn's aim was to reduce the actual and potential harm caused by alcohol misuse to individuals and communities whilst ensuring that alcohol could still be enjoyed safely and responsibly. According to the Royal College of Psychiatrists (2011) every single pound spent on treatment in the public sector saves £5. According to a regional public health observatory report (2008) if providing alcohol treatment to the 10% of the UK population who are problem drinkers it could reduce costs between £109million and £156million a year (PHO, 2008). The data did not differentiate between the various age groups so it proved difficult to specify how many of these were over 65 years.

Evelyn's business plan was presented to her Trust and they agreed to fund her £1500 towards the development of the board game.

The prototype game and poster materials cost £260.00. However, Evelyn had not calculated the cost of staff time of those who had incorporated the game into their busy workload and the plan outlined a projected spend of £650. She had also not incorporated the cost of future orders for the game to be used in other Public Health services across her locality. In addition, there was no costing for training other health professionals to use this game as a public health tool. Overall there had been no forward thinking and the business plan was not thought through in a logical format and planned expenditure was not part of the business plan. The funding only enabled the development of one game and one person to deliver it which does not demonstrate sensible business acumen. Even-though Evelyn was very enthusiastic and the game was extremely innovative and fun this was not enough for the project to be successful. The costings totalled £900 which was within the budget of £1,500, this left a surplus of £600. Unfortunately this was not enough to develop additional

games to be utilised across the various sectors, so sadly this business venture did not succeed. Evelyn continues to use the one game originally supplied and uses it when she is delivering a public health awareness session to health professionals.

She was regretful that future planning had not been part of the business plan because those that had used the game found it very useful and some of the comments are made below:

"Information this way sticks, remember old brains need reminding ....."

"Start and stopping places need to be larger for old eyes, but it was very informative"

In Evelyn's own words she felt that she was "in a small boat in a huge unknown sea, rudderless and without direction". This really sums up why this project did not succeed - there was no clear direction. Enthusiasm and creativity was not enough without a solid business case to support it.

Evelyn became very disillusioned, in her words, 'with the patronising attitude that her 'game' received from her immediate Line Manager and subsequent senior clinicians' so she has now gone freelance. She has not gained further funding for the game, however she has adapted the game and has developed it as a question and answer quiz which she presents at her local age concern groups.

6.6 Case Four – The development of sexual health promotion material (unsuccessful initiative)

The project was seeking funding for the implementation of a series of quizzes for adolescents as a way of promoting sexual health promotion. This was being developed alongside the development of a book which had used the same technique of questions and answers relating to sexual health. The project was two pronged, firstly the piloting of a series of quizzes and secondly the support for publishing a book based on the findings from the quizzes. It was anticipated that the quizzes would take place in schools and public health and sexual health clinics in the South of England. This concept began when Jeannie in her role as a School Nurse met a young girl from a local grammar school who had in the past three months experienced unprotected sex with many different men, all of them casual and unknown to her. She had been diagnosed with chlamydia, gonorrhoea and genital

warts, all of which had successfully been treated. She later developed genital herpes and the genital warts returned. This still did not result in her changing her sexual behaviour. She was aware of the implications of sexually transmitted diseases, but she didn't like using protection, so she intended to continue having unprotected sex. The fact that this was a young intelligent girl who had this attitude left Jeannie struggling how to get the message of safe sex across to all the young people in her local schools.

Jeannie decided that she would develop a sex challenge quiz as it combined fun with learning in a non-threatening way which she hoped would create a competitive spirit to excite and engage young people. Jeannie felt that this approach would bring a fresh approach to sexual health learning. The hypothesis for addressing this situation was also supported by the Health protection Agency (2012) who claim that there is evidence of considerable inequality in the distribution of sexually transmitted infections (STIs) and efforts were needed to continue to focus on high risk groups in adults and should also be intensified in those parts of the country with the highest rates of STIs.

Acknowledging that local clubs attracted teenagers and young adults Jeannie developed the quizzes to target a 14-24yr old audience and the logo used for the intended book was graphic in the hope of gaining immediate interest from this group. As more groups participated the objective was to measurably improve the sexual health profiles in those particular areas that had participated.

The business case Jeannie developed was looking for £36,000 to pilot the project for the development of the games and discussions with publishers, which when scaled up would aim to produce a saving of £330,000 per annum or £1million in 3 years in the Southern region where Jeannie worked. This could be done by reducing the amount of STIs by 10% and human immunodeficiency virus (HIV) by 1% as STIs diagnosed amongst young people in the area where Jeannie worked were 50% higher than any other region in England.

The cost of treating 25,000 cases of STIs in her region is £3.3m per annum which is based on the tariff of £132.00 per case. If there was a 10% reduction in treating STIs this would equate to a saving of £330,000 per year as mentioned above. The £36,000 was required to develop, pilot and evaluate the competitions.

This project did not go beyond the stage of being a 'good idea', despite the clear statistical saving that could have been gained. When the business case was presented only three commissioners attended the presentation and they claimed that they did not have enough unallocated underspend to support the business plan. Some of the commissioners did not like the language used in the quiz questions and did not feel it was appropriate to be used in schools. If the language used was going to be transferred into the book they would not allow the book to be part of any school health promotion campaign.

This is an innovative approach to delivering public health and sexual health awareness but Jeannie needed a stronger rationale for using this technique to get the message across. To put this project into action a sample of questions and text from the book should have been sent to managers to gain some idea of how it would have been accepted. Consideration of using 'more acceptable' phrases might have gained some approval from the commissioners. However, this was Jeannie's point exactly, she felt that the usual health promotion messages on the topic of sexual health has always been delivered in 'acceptable language' which is not necessarily the language used by many young people who are participating in unsafe sexual habits, hence using this radical approach as a means of capturing attention. The business plan was also too busy. Jeannie should have just focussed on introducing the quiz to see how this could be used effectively and then produce another business case for the development of a book which would have had clear statistical evidence of why it would have been successful (providing the quizzes had taken place).

Jeannie has since become a social entrepreneur and has published her book abroad where she has gained support from a government initiative in a country where HIV is rife. She no longer works as a School Nurse but spends her time working with young sexually active people and attends sexual health clinics, 'drop in clinics' where she can meet young people and she continues to promote sexual health awareness.

Even though the quiz/book projects were not successful under the auspices of the NHS, they are still being utilised in ways that public health messages can be received by the relevant groups.

# 6.7 Summary

These four examples, whether successful or not, clearly demonstrated the passion that the individual project leads feel for their particular topics. Passion, however, would not seem to be enough, an ability to apply learning from their course would seem to be an essential requisite if innovations are to be supported.

# Chapter 7 Discussion

# 7.1 Discussion of key findings

This chapter will discuss the key findings of the study which examined the personal and professional development of a group of community healthcare professionals,

who, after training in social innovation, instigated and managed their own innovative community healthcare initiatives. Here I will outline the key findings before going on to discuss some of the threads that draw them together and how they can inform recommendations for prospective innovators, their managers and creators of training courses in social innovation for health professionals. As discussed in an earlier chapter, given the paucity of material on the personal characteristics and experiences of healthcare innovators, industry, the arts and education were searched to look for similar studies, and some provided helpful contexts for the findings here.

Through gaining new knowledge about innovation the participants were better informed. This impacted on their professional capacity to work differently and more creatively in their clinical areas in their attempt to improve service delivery.

This study identified and surfaced several characteristics and attitudes that played a significant part in each individual journey mentioned in this study. The key qualities emerging are: having a 'can do' attitude, perseverance and resilience, recognising the need to involve others and work differently, having self-belief in their clinical knowledge and confidence to make changes, not being frightened of failure and learning from it. Personal characteristics also surfaced, with some having an innate confidence from the outset of their project and training and gaining an even deeper understanding of themselves and their strengths, also actively working in an innovative way served to enhance their confidence further. Hence, self-efficacy and confidence in their specialist subject area was evident and a key theme. Other participants only recognised or developed their confidence when they had the chance to demonstrate to themselves and others that the improvement in their clinical practice showed that they were working more effectively, combined with evidence of an increased patient satisfaction.

### 7.2 Exploration of skills required to set up an innovative project

The first set of findings related to the start of the participant's journey and to the first research objective: *To explore and identify the personal qualities, knowledge* 

and skills required of health care professionals when setting up and managing a health and social innovation project.

## Emerging Themes:

7.2.1 Better is possible – a perceived need to raise standards of health care delivery above and beyond current practice / Swimming against the stream – challenges faced when setting out ideas for improvement

At the very start of the process which resulted in their becoming effectively a 'social innovator' most participants had little idea what the term meant and had not consciously aspired to become one. Their starting place was personal and local, frequently as a result of observing poor, but not necessarily bad or dangerous, healthcare. In different locations this comprised of: wasted resources and the delivery of suboptimal care. Their concerns often centred around practical issues such as ineffective time and resource management, the inequitable delivery of good clinical care, poor communication of patient risk, a lack of specialist services and difficulty in maintaining excellence in clinical practice. Most significantly, however, was a factor not visible to the naked eye - a lack of imagination from managers to identify a need for improvement within their organisation. It often took a complaint or crisis to alert managers to the situation, these research participants however sensed disquiet and unease and were keen to avert any concerns through better practice.

Commentators in other fields such as the arts, (Beeftink et al, 2012), industry (Miller & Le Breton-Miller, 2017, Laforet et al, 2006), education (Sarasvathy,2001)) as well as those in healthcare (Adams et al, 2006, Pekkarinen, 2010) acknowledge that even once a problem has been recognised, applying innovation to an established current practice can bring challenges. The findings from this study report that only a portion of the NHS organisations possessed the capability, or desire, to invest in new practices if there was no pressing need. Adams et al (2006) and Cormican & 0'Sullivan (cited in Patterson et al, 2004) claimed that when innovation is applied it is often disjointed, frequently with opposing pressures and so only achieved in part. This was not reflected in this study where an innovation was accepted, a project was either implemented or not, there did not seem to be any middle ground.

Sarasvathy's (2001) principles for effectuation, suggest that to create solutions resources must be available and Sarasvathy's concept of 'effectual reasoning' begins with a given set of means and allows goals to emerge fortuitously over time from the various ideas and diverse aspirations of the founders and the people they interact with. These ideas can witnessed in this study, the participants had to make changes, initially with existing resources, and make good use of their imagination and that of those around them who were supportive. However, some projects did require additional resources such as the befriending service project (project 25, Appendix 1). This required a hired bus to take older people to and from a befriending service at a local community hall as a means of reducing social exclusion.

Unfortunately a bus was not provided so this service was unable to proceed. Knock backs such as this were frequently experienced by the participants, nevertheless this did not prevent them from trying other tactics and so *resilience and perseverance* also became a key theme throughout this study.

These findings are reflected in the work of Beeftink et al (2012) and Laforet et al's (2006) studies focussed on the creative industries and small manufacturing firms where innovation is central to the ever changing demands of creative businesses and small organisations and they discovered that self-efficacy was crucial to survive the challenges in these areas, and found to be an important predictor of performance. This suggests that one's belief in one's own ability to succeed plays a major part in how individuals approach tasks and challenges as well as a measure of their likely hood to succeed. This echoes the findings here, of participants who developed their creative skills to solve elements of suboptimal service provision within their Trusts. It could be argued that these participants had an innovative cognitive style (Shalley et al, 2004, cited in Beeftink et al, 2012) revealing their creative orientation to problem solving, promoting individual performance as well as organisational innovation and effectiveness (Sung et al 2009: 941).

Miller & Le Breton-Miller (2017), in their exploration of sources of entrepreneurial courage and imagination, reveal several characteristics that resonate here. For example seizing opportunities and the willingness to take risks in pursuit of a belief was a key finding. However, Miller & Le Breton-Miller (2017) report that the outcomes in their studies were not always clear in the minds of their subjects at the

onset, whereas in this study all participants had clear ideas of the desired outcomes. This may be the reason why at least two thirds of the projects were successful.

An industry based study by Sullivan and Ford (2013) suggested that individuals require courage to take a risk, even if the outcomes are not known in advance, a view also supported by Miller & Le Breton-Miller (2017). Despite seeing a need for this change some individuals remained afraid to take a risk. The participants of this current study had similar concerns when their Managers were frequently unprepared to take a risk even in the light of limited service provision which could impact on ineffective care. Whilst the NHS and the Department of Health's regulatory bodies have published documents showing how their approach to regulation supports innovation (DH, 2017) incorporating the topic of innovation in its policies, it often fails to manage innovation and innovative people well. This strategic message (encourage innovation) is targeted at all community health services, but according to Teece (2007, cited in Miller & Le Breton-Miller, 2017) even this strategy is not static, it changed and often left one time willing participants confused as to what they could do. Participants in this study were frequently presented with such dilemmas (strategy shift or change of department goals) which added to their frustration.

Pekkarinen et al's (2010) study looked at how public sector in Finland dealt with innovation and highlights some of the challenges which are similar to those experienced by the participants of this study. Minimising the amount of non-productive time, decreasing the costs and providing faster treatment and improved value for their client group, and breaking down 'silos' and encouraging cross-departmental co-operation, are all areas which, it is claimed, can be influenced by innovation. Interestingly Pekkarinen et al (2010: 521) have also claimed that the organisational culture in the public sector is characterised by hierarchy and bureaucracy which can stifle creativity and is also supported by the findings of this study. Moreover, and encouraging is the opinion of Pekkarinen et al (2010) that often these clashes frequently pave the way for innovation. Several of these participants found this to be true.

7.2.2 Can't do it alone – recognition of the need for further training and support, Early involvement of others – No man (or woman) is an island, Seeking support

Findings from this study acknowledged the advantages in seeking out the influential people in an organisation who could not only support an innovation but also persuade others to follow, supported by Bandura's (1997) thoughts on learning from others. Many of the influential people required further persuasion, but through being grounded in the agentic perspective, social-cognitive theory suggests that the participants' self-efficacy and belief in their innovation and evidence of sustainability was sufficient to make things happen. Brooks et al's (2011) study which looked at innovation in mental health services highlighted the importance of the local context as a means of ensuring that the innovation is sustainable. Similar findings were found in this study recognising those that were sustainable had brought their colleagues along with them.

Fitzgerald (2003) looked at how credible evidence of a need for change and proposed benefits influenced innovation in primary healthcare settings and how interorganisational collaboration and multi-professional involvement impacted positively on diffusion of innovation. The findings signified the importance of sharing information and open debate with other professionals prior to adopting a new process. The participants of this study did not always work this way, there was not always evidence of their seeking open debate with their colleagues or informed others, they relied rather on presenting evidence to support their business case, frequently focussing on their specific discipline only. Of importance to these participants are the lessons learnt from Fitzgerald's (2003) study. They show that working in 'silo' is not the most productive way to promote innovation. O'Leary (2011) when discussing innovations in healthcare concurs with this and commented that in order to drive up the quality of care we can all learn from each other. The findings from this study also agreed with this as many of the interviewees commented on the support and knowledge they had gained from listening to, and sharing their experiences with fellow students, but several could do more to work cooperatively and to make that a goal.

DeGraff et al (2002; 140) writes about incubating creativity, promoting motivation, increasing employee satisfaction and developing a strong sense of community. The success of this approach depends on having supportive management to enable those incubators of creativity to have autonomy to develop their ideas. McLean (2005) reports the importance of having management support to progress an idea,

as numerous excellent ideas never see the light of day. Therefore it is imperative that managers are made aware of new initiatives from the onset if support is going to be forthcoming. Findings of this study support this view and one of the participants of this study suggested:

"Get 'buy in' early on. Find an influential person in the organisation that can support and fight your corner"

Opportunities for innovative ideas are often stifled by competing demands and unfamiliarity of the concept; frequent competing demands and poor communication across teams and organisations prevents the progression of an idea (Cooper, 2010). To bring an idea from concept to market, it should be recognised for its potential; it must receive funding in an environment of scarce or at least competing resources; and it must overcome potential obstacles such as technology challenges, competitive pressures, and a variety of other obstacles. For example, Baghi (2015) carried out a case study in Shanghai on the application of mobile healthcare and his findings showed greater patient engagement and compliance as patients were able to monitor their conditions. This could be compared to the specialist nurse here who developed the on-line support system for NF1 sufferers which gave patients, and families, greater autonomy and ownership of their condition, or the community matron who developed the virtual ward, directing care to the appropriate professional, preventing duplication of services.

7.2.3 Change of Perspective in recognising alternative ways of approaching clinical practice and working differently across the sector

The community work-force are pivotal to changes occurring within the NHS relating to the second objective of the study: *To identify the difficulties encountered within individuals themselves and the external factors within their organisations that have impacted on the development of their project*. Many of the changes are outlined in the *'The Five Year Forward View'* (DH, 2014) which plans to meet the challenges of people living longer, frequently with long term conditions and being cared for in their homes or community setting. The Kings Fund (2014) report that community services make up a large part of the NHS service provision from chronic disease management, rehabilitation, health visiting and

midwifery services. It is essential therefore that professionals work together in an integrated way. This causes challenges but reinforces the importance of bringing people on side at an early stage in the change process. Herold & Fedor (2008) also highlight that even if people are supportive of the change from the beginning they must realise that the acquisition of new routines and adapting to these changes may take some time as these are frequently overlooked. Many of the participants of this study were only supported for a set period of time, eg. 3 months, 6 months, 12 months which could lead to difficulties around sustainability. It was therefore essential for them to address the time line in their business case.

Heathfield (2016) commented from a human resource perspective and advised that plans need to be put in place at the onset of a potential change so that teams have a responsibility in building and developing the change and have some ownership of the change. The findings from this study support this view, whereby early involvement had been successful, as in the case of the community matron with her virtual ward project (participant 9) and with others recognising that they should have had buy in earlier, evidenced in the occupational therapist interview (participant 2). This will require focussing on the benefits of the change, cost of service provision, patient experience and outcomes both quantitative and qualitative (Gosselin et al, 2015) and thereby providing a solution and an enhancement of the care provision.

It is virtually impossible for managers currently working within the NHS to be dismissive of innovation as The International Council of Nurses (ICN) commented that if progression was to be made in health services across the world then innovation was fundamental (Joseph, 2015) and this was reinforced again in 2010 when the Prime Minister's Commission on the Future of Nursing and Midwifery in England pledged to promote innovation, stressing that innovation in nursing would become a way of life introducing a cultural change which would affect all levels of nursing (Cooper, 2010). The Kings Fund (2016) also recognises that this will place pressures on the NHS which can only be achieved with the involvement, engagement and commitment of NHS staff. It requires changing cultures both within and across organisations. The participants of this study were constantly trying to change the cultures of their organisations with the aim of driving their innovations forward, but even so they found support erratic.

Even-though many managers may not have the vision they have a responsibility to support their staff in promoting innovation and creativity as this had become a government mandate. In 2009 the NHS produced a survey report on NHS Innovation and Improvement which claimed that there was a 'hunger' for widening knowledge of innovation, but the NHS also said that there was room for improvement through better cross-team integration and communication (NHS, 2007). These views have also been supported by Fitzgerald (2003. The success of any innovation depends on responses from early adopters, realising that individuals will adopt at various stages, so the implementation of an idea may take longer than anticipated.

In a study by Nelson et al (2010) it was the external forms of social support that were associated with the adoption of an innovation. This is relevant to the NHS who frequently rely on external support, whether in the form of external staffing agencies, transport, voluntary services or charity support. However, this was not the case in this study – the staff relied on internal support almost exclusively. The innovations were based on clinical expertise with some administration support only, adding to the value of this research as few external services were required to implement their innovations. Yet unlike commercial companies where they try to introduce new innovations to the market as soon as possible (Dervojeda, K., Schretlen, J., 2013), adopting an innovation in the NHS does not automatically involve their customers (patients). Frequently an innovation is introduced and patients are informed without being an active participant in the change. Once again, this was not evident in this study. Where appropriate, such as the prostate cancer support telephone follow up service (participant 3) and the heart failure service (participant 1), patients were actively involved in the changes recommended to their service provision.

An inductive approach was used in Haggman's study (2009) looking at how innovations began and what were the deciding supportive factors that enabled them to survive. He found that it was essential to recognise a 'champion' of the innovation so that management could be educated in the background of the innovation and the relevance and potential benefits of the planned change would be transparent. This view was supported by Sloan (1875, cited in Miller, Le Breton-Miller, 2017) who said:

"There has to be this pioneer, the individual who has the courage, the ambition to overcome the obstacles that always develop when one tries to do something worthwhile, especially when it is new and different." —Alfred P. Sloan (From Miller & Le Breton-Miller 2017.)

An essential skill of managers taking a new initiative forward and changing a service requires faith, confidence and an ability to respond to challenges. They have a duty to mobilize employee's knowledge and encourage innovation at the same time as maintaining quality. Malloch (2013: 60) reports that leaders within health care organisations are being held responsible for inappropriate variations in care and suboptimal care which could be a reason for not wanting to make changes. However, the participants of this study were not letting this attitude interfere with their projects and continued to proceed with presenting their ideas.

However, in order to alleviate such accusations, solutions to any suboptimal care should be addressed. Yet the pace of implementing a change to improve care, no matter how small, can be slow. Ryan (2017) suggests that on occasions change is not recognised, but gradually change is implemented and individuals begin to react positively to changes quite slowly. Berwick (2003) found in his study outlining that mastering the generation of good changes is not the same as mastering the *use* of good changes a view also supported by Simpson (2009) who recognises that organisations will need time to prepare for change and must consider staff resources, staff training, financial implications and patient satisfaction which can be very challenging. In presenting their business plans to management the participants of the study were able to take account of potential changes and where appropriate included this into their plans for change.

Change is incremental and requires fine tuning and monitoring as it is implemented, reviewing the outcome and perceived benefits on a regular basis (Kaya et al, 2015). Managers who are early adopters frequently have the management skills to support an innovation but do not always have the initial vision, demonstrating they are not risk averse. They work strategically so are influential in gaining support from others who may fall in the early majority category range. Roberts & Machon (2015) recommend that for managers who are supporting change and innovation it is essential that they change themselves first. However, even managers who are early

adopters do not always find working with innovators easy as they challenge current practice, but managers should recognise that these innovators are 'diamonds in the rough' (Roberts & Machon, 2015) and may need to look beyond the individual and consider the innovation that is being presented.

The challenges also encouraged the participants to work differently and so generating change in the workplace. Ovretvreit et al (2012) reviewed twelve organisation and management innovations in Swedish healthcare. The findings showed that the clinicians and not the managers played a significant part in getting innovations implemented. They also found that the internal local context of the organisation was more important than external factors. These findings are not dissimilar to the results of this current study. Interestingly Ovretvreit went on to suggest that innovations may not always be original, but they may be new to the organisation which is when organisations must consider working differently; also suggesting that to work differently frequently leads to quality improvement which was also evident in this study. The specialist nurse who worked with the district nurse to improve services to patients with a neurological condition (participant 23 and the dietitian (participant 10) who identified patients requiring supplement drinks are just two examples of improving quality of care through working differently. One of the challenges that Ovrevreit et al's (2012) study discovered were the various interpretations of the innovation which links to the findings of this study as it was frequently reported that managers did not understand why change needed to happen, often interpreting change as a request for more resources which was not the case.

In the current climate of transformation of community services it is the ideal time to look to new ways of working, engaging stakeholders and other health professionals as changes are made, rather than forcing changes unfairly without consultation. Working through this transformation will require determination, belief, accurate business planning, realistic anticipated outcomes, cost effectiveness and clear communication. Despite this MacFadyen (2015: 4) reiterates that the 'bottom-line principle' of any new idea is the responsibility of the individual and suggests that this may not be easy as they may have to fight for their belief as it may be the only way that change will happen. Linking to this study, the perseverance of the participants displayed resilience to deal with these challenges.

7.3 To look at the impact of education and training on the progress of the participants was the third objective of the study. The educational input had developed in all participants, insights and skills which enabled them to carry out an innovation in their clinical area.

Carr & Clarke's (2010) study demonstrated that nurses struggle with tensions as they aim to improve clinical practice. Approaching their work with a new fountain of knowledge enabling them to work differently they are frequently met with barriers such as entrenched ways of working. These barriers were also experienced by the participants of this study who were frequently confronted with phrases such as "we have always done it this way, so why change now". Statements such as this were challenged by the participants with one of the participants in the study saying: "Just because something has always been done like this it doesn't mean that this is the best way".

It became apparent to all, that to succeed in their endeavour to improve their patients' care they needed to work with others and that their skills and knowledge base required development. The most frequently reported gap in their knowledge was financial management: budgeting and fundraising and how to present a business plan. They looked to formal education to meet these needs but the learning experience gave them so much more. Korukcu & Kukulu (2010) who carried out a study on innovation in nursing education discovered that a number of innovations were being utilised in nurse education resulting in nurses being taught in non-traditional ways, such as using technology, simulation and action learning sets, some of these methods were used by the participants of this study. As mature learners the use of action learning sets supported their experiential learning giving them the ability to draw on their experiences using their new knowledge to form the basis of their discussions. While learning the business skills they were exposed to other ideas and learning experiences such as how to understand white board economics and carry out costing exercises.

Cooley's (2008) study looked at nurses' motivations for studying post-registration nursing programmes and the impact this had on their personal lives. The findings revealed that the reasons were to assist their professional development and to keep up to date with new treatments. They also wanted to gain skills in doing things differently which resonated with the participants of this study. An important finding

from Cooley's study was the positive aspect of working with multi-disciplinary teams which did not necessarily rate high in this study, but did resonate with not working alone.

Stanley (2003) reviewed the lived experience of part-time post registration students who were similar to the participants of this study. An increased confidence was discovered which met with the findings of this study, but additionally Stanley (2003) discovered some of her participants had found the studying stressful and had impacted on their personal life. This was contrary to the findings here as stress was not mentioned at all and the personal development had flowed over into their personal life enhancing their personal development. The demands of their course developed their level of perseverance and raised the motivation to new heights. It made their dreams a possibility.

7.4 The real me – finding a professional home, Increased confidence, Personal development

A finding that was raised by almost every subject and, it could be argued is the single most important one, was the strengthening of the 'self', the discovery of the 'real me' as a result of being in a group of likeminded people. This new manifestation of the 'self' came about through being accepted by others – no longer being the 'difficult' colleague – and by working through the many stages of getting their project off the ground. The struggle had made them stronger. This new self-belief was for some transferred to their personal lives in a positive and sustainable way.

Bandura & Locke (2003) recognised that individuals function at different levels which affect their personal efficacy, stressing that the level of self-efficacy shown by an individual frequently linked to individual perseverance in trying to solve problems. This was evident in this research as many of the participants thrived on trying to solve, what for many would be, unsolvable problems. One such example is the specialist nurse (participant 6) who set up leg ulcer clinics in both areas of her borough delivering an equitable service across her Trust a seemingly simple approach, but it was met with strong opposition.

The individual characteristics of those involved in this research were crucial as to whether a project was successfully implemented or not. One of the key

characteristics found in this study was that of self-efficacy as already mentioned, and research carried out by Amo (2006), Ziyae et al (2015), Mobaraki et al (2012), Beeftink et al (2012), Tierney & Farmer (2011), Ng & Lucianetti (2015), Kumar & Uzkurt (2010), Bandura & Locke (2003) have consistently shown that self-efficacy contributes to attitude, motivation and overall performance.

Sarasvathy (2001) argues that all entrepreneurs begin with three categories of means: (1) Who they are – their traits, tastes and abilities; (2) What they know – their education, training, expertise, and experience; and, (3) Whom they know – their social and professional networks. The findings of the current study also highlight that the personal traits of the professionals are key to their success, education is important and developing professional networks were key to the success of several projects. And as Sarasvathy (2001) says 'Using these means the entrepreneurs begin to imagine and implement possible effects that can be created with them. Most often, they start very small with the means that are closest at hand, and move almost directly into action without elaborate planning.' Our healthcare innovators started with what they had at hand, funds, existing policies, the limited support of colleagues but used them to create a new way of working.

These survival tactics fall in line with Robinson's (2014) traits for a successful entrepreneur, whilst these are not research based the participants of this study displayed these seven traits: tenacity, passion, tolerance of ambiguity, vision, self-belief, flexibility and rule breaking. Paterson et al's(2009) review of characteristics and behaviours of individuals support these views but have also added that 'Intelligence is a necessary but not sufficient condition for innovation' claiming that cognitive ability alone does not account for much of the variance observed in individual innovative performance, and that domain specific knowledge is a key resource for innovation in organisations. This was apparent in all of the projects that were presented; each individual had a vested interest in the project because their whole self was emerged in their specific innovation due to their intense knowledge of the subject. Sung & Choi (2009) also identified five personality factors in support of innovation, these being; extraversion, agreeableness, conscientiousness, emotional stability, openness to experience. Other than extraversion the participants of this study would fall in line with these characteristics also. Reflecting on one interview

the participant said that she felt that she did not have to have an extrovert personality to be a successful innovator.

The majority of the participants in this study identified a resilience which was driven by their increased confidence and self-belief. As such there are many factors that contribute to the use of resilience in the workplace and for those practitioners working in the field of community health there is a greater need for resilience now more than ever. Cusack et al's (2016) study which looked at factors in nursing workplaces that promote psychological resilience identified that environment as well as self-efficacy, coping and mindfulness were important factors in promoting resilience. This is particularly pertinent to the community health professionals who frequently work in non-purpose built community buildings and also patient's private homes. This adds to the significance of their coping abilities and resilience as they are often working in sub-optimal environments.

A study by Cope et al (2016) claim that nurses are survivors and transformers of their own reality which is why resilient nurses are the ones that remain in the workforce despite the adversity. This is exemplified in the data gathered from the participants of this study who developed their resilience as the belief in their innovation grew. Even though some of the participants of the study did not remain in their place of work, they all transferred to an environment where they were able to utilise their resilience to develop their innovation, supporting Cusack et al's (2016) work on the relevance of having a supportive environment.

A report carried out by the Royal College of Nursing in 2016 found that nurses are under more pressure now than ever before, often resulting in burn out, a view supported by The Kings Fund (2015) claiming that the UK suffers from the highest burn out in Europe amongst NHS staff. These indicators have also been seen by Turner et al, (2015) and Rushton et al (2017). Therefore, to be able to work effectively amidst the chaos, changing landscape and demands of the workforce, resilience is recognised as a beneficial attribute enabling healthcare professionals to flourish and thrive amid challenging work environments (Hunter & Warren, 2013).

Masten's (2001:228) research on resilience in children looks at the systemic nature of resilience, for example the risks, assets and protective factors and suggests that: "resilience refers to a class of phenomena characterized by good outcomes in spite

of serious threats to adaptation or development". Similar research on adults is scarce even though organisations offer training on resilience as a means of one could argue that this does not deal with the issue of stress within organisations. This is an area that could be researched further. Nevertheless this concept is transferable to the participants of this study who were constantly threatened with negative views from managers and colleagues but had positive outcomes nevertheless.

Whilst there is considerable overlap between resilience and other psychological constructs (Cusack et al, 2016), with most definitions based around two core concepts: adversity and positive adaptation. Although resilience has been a major construct in positive psychology (Winnie et al, 2011: 610) frequently strengthening the capacity of an individual to respond to stressors in a way that balance and harmony is maintained (Pines et al. 2012) and is a necessary quality for surviving negative effects of workforce challenge and stress (Tusaie & Dyer, 2004). As such there are many factors that contribute to the use of resilience in the workplace and individuals react to difficult times in a variety of ways, some becoming overcome with stress but others strengthening their determination and energy resulting in a resilient approach which gives individuals the strength to break cycles of hostility and negativity frequently displayed by managers (Jones, 2012). Having an optimistic attitude and being able to contextualise and visualise the anticipated outcomes of a project is an additional benefit in following a project through from idea to implementation which corroborates the actions shown by the participants of this study.

The ability to cognitively process thoughts and ideas is instrumental in fuelling resilience which frequently results in the formal solution-focused therapy approach (Walsh, 2010). This model originated in family therapy and has since proceeded to be used in organisational change, incorporating strength-based values. It was not apparent that participants were consciously using a specific model, but the fact they began with the solution relates to Walsh's theory through the utilisation of their individual strengths.

The personal development experienced by the participants of this study was grounded by their inner confidence which gradually developed as they progressed through the programmes they studied, but also as they were able to see their

innovations becoming a reality. Research carried out by Gardner et al (2008) on nurse practitioners in Australia and New Zealand suggest that confidence increased as knowledge and competence improved. Gardner et al's (2008) study collaborates the numerous findings of this study: working well with others, being creative, taking risks, open communication, being a knowledgeable doer, self-efficacy in managing unfamiliar situations and having a vision. Gardner's (2008) work also supports the notion that education and training are a significant factor in nurses' confidence. Even though Gardner et al's (2008) study was not looking at innovation the traits found in their participants were similar to those found in this study. Gerrish et al (2011) also looked at the role of the advanced practice nurse and concurs that knowledge is power, instilling confidence in nurses.

#### 7.5 Recommendations

The fourth objective: To inform potential innovators about what they can expect to encounter as they develop an idea and wish to develop their own project, will be met when the following recommendation are met.

The findings of this study established that providing the participants with relevant skills to formulate a business plan with sufficient detail to present to a business board suggests that these are skills that all health professionals should aspire to. Therefore it is recommended that innovation should be a subject covered within all health professional curricula, particularly post registration programmes. As a result of carrying out this research and discovering that the participants of the study were making a significant difference to community service provision, the researcher was influential in changing the Specialist Community Public Health Nursing and Specialist Community District Nursing programmes which are delivered at the researcher's university. These programmes now have a module which looks specifically at innovation in the community providing students with an introduction to economics to enable them to prepare business cases demonstrating clear costings, savings for the organisation and expected outcomes and benefits for patients. Commissioners will be invited to speak to students outlining what their expectations of business proposals are and established innovative practitioners will also be invited to speak to students to discuss how they achieved the support of their organisation in making

changes. The importance of sustainability and engaging key influential people in the organisation are essential skills to develop, additional topics such as developing personal resilience, dealing with failure, swallowing pride and taking advice from others and listening to others within the workplace are all be essential elements of the module. The overall aim of the module is to produce practitioners who are more entrepreneurial and business savvy than previous community health professionals. Community health professionals will be expected to present business cases in a factual way identifying clear anticipated outcomes and potential savings to the organisations whilst putting aside any emotional factors if they want their business case to be viewed seriously. They have to understand that taking an innovation forward will require tenacity and viewing their role as one of immense importance in getting changes implemented in their workplaces. In times of transformation of community services health professionals must contemplate working differently demonstrating an increase in shared working, utilising the expertise of others and reducing silo working to provide high quality care.

The researcher is a member of a national body of community health educators and the findings of this study and the introduction of the module on innovation will be shared with this national group as a means of influencing future curriculum developments across the country. All members of this group provide post qualifying community health programmes and in introducing this module to future community health professionals it will enhance practice knowledge and result in a strengthened national community workforce. The community workforce has previously been seen as a Cinderella service compared to secondary care provision. However if there was a national drive to enhance the skill set of community practitioners demonstrating that they are business minded and can promote innovation within community services it is anticipated that local organisations across the country will see an increase in efficiency and patient satisfaction and the government may give community services the recognition it deserves.

The findings of the research stress the importance of not only having the right environment but having the support of the organisation to support an innovation. A poorly structured organisation without the ability to support staff will frequently result

in failure. The research discovered that some managers were prepared to take on the ideas of others but had not always recognised the suggestions as a potential innovation. If Managers were given the tools to recognise an innovation and a skill set to provide a platform for innovators to be listened to it would improve the possibilities of innovations being driven forward. The tools to promote innovation could take the form in various guises:

- A handbook identifying key characteristics of innovative people The overall aim of the handbook for managers is to have a resource that will give the managers skills in supporting innovative practitioners.
- Suggestions to enable staff to have 'thinking time' to experiment with new ideas
- Space and opportunities for staff to discuss evidence based practice and share best practice with like-minded people
- Carry out a meaningful appraisal with employees enabling them to speak openly about new ideas and initiatives
- Provide opportunities for employees to attend an education/training in innovation giving the employees the tools to promote and implement their innovation
- Develop study days for managers as a 'taster' to prepare them for working with innovator practitioners.

A Community Health Care Trust has asked for the findings of this study to be presented to senior managers in their organisation. This is an ideal opportunity to disseminate the results and engage local health providers and commissioners to work with their front line staff in providing evidence based innovative practice to patients in their region. The presentation and a list of key factors in recognising innovation will be provided. Work will commence on developing the handbook for managers with the aim of having a finished professional product by Spring 2018.

Finally, the contribution to practice knowledge will be enhanced through publication of this study. The researcher initially aims to publish the findings in peer reviewed journals such as the Journal of Nursing Management, Journal of Community Health and Health and Social Care in the Community journal and also the British Journal of

Educational Psychology. These journals are targeted at managers, educationalists and community health professionals and will address the skills required to be an innovator, but most importantly, will stress how enabling innovation to occur has opened up opportunities for the participants identifying 'the real self'. The discovery of the 'real me' was the most profound finding of this study and this should be shared with others, increasing understanding of the complex learning and development that innovative practitioners experience.

#### 7.6 Limitations of the study

A wider sample: The sample was guided by the opportunity to include individuals who had undertaken formal education in innovation or had presented their own social care projects at conferences, two factors are noteworthy: 1) the imbalance between the genders (more women than men) and 2) the limited ethnic diversity among the sample. It would be worthwhile investigating both issues. Are there in reality as many men working as innovators within the NHS as women but pursue their goals without feeling the need for structured education? Likewise, are there as many healthcare professionals from a diverse ethnic pool working innovatively but again do not seek out training or are there barriers to some individuals considering undertaking a plan to improve clinical practice in their area. These biases, may well have influenced the responses reported here, this research is therefore, only a snapshot of a growing phenomenon which must involve a broad range of staff profiles.

A longer study: The researcher was in contact with participants for just two years which is a relatively short period of time considering the complexity of setting up a new project and seeing an impact, let alone get any idea of the sustainability of the projects. Revisiting these same projects and their innovators at the five year point would be instructive.

Theory: This study is a straightforward, descriptive, inductive study which has not been informed by any theoretical perspective from the disciplines of management or psychology. As so little was known about the topic this was considered a legitimate place to start, letting the participants tell their story without any externally imposed

framework. However, given the complex and personal nature of the themes that emerged, it may be of value if any future study worked within an established psychological framework to give a deeper understanding of the lives of these innovators.

#### 7.7 Conclusion

This study has produced some important findings in relation to the personal and professional journey of healthcare social innovators and has provided information for others taking the same road in future. This is the first time that the process of innovation has been researched from the perspective of the health professional. It became clear during the interviews that behind each innovative project there was a dedicated, indeed passionate, person who cared very much about the quality of service their patients and clients received. It was their observations of suboptimal care in their professional area that ignited their interest: money being wasted, time misused, poor use of existing resources, poor communication of risk to patients were all observed in various forms. A secondary, but important motivator was the apparent lack of concern by many colleagues and managers responsible. It was the lack of attention given to patients and sometimes their families, to deliver care that was excellent not just adequate that slowly pressured the innovators to take action which may have been due a lack of vision, or even ambition. Nevertheless this could not be ignored and the participants of the study felt the need to intervene.

Some interviewees had been in a state of limbo for many months, frustrated at a lack of encouragement from their managers before they discovered and signed on to their particular course promoting challenging the status quo and supporting innovative practice. Many reported feeling like the 'difficult' colleague and that their idea for improving practice was seen as criticism of current work and colleagues. However, once in the company of like-minded individuals in the safety of the classroom they blossomed and talked of becoming the 'real me' for the first time at work. The combination of their learning and the practical experience of setting up a project led to some significant developments. Their readiness to identify a solution rather than focussing on the problem was an essential element to the success of their projects.

Many barriers and challenges were presented along the way, but the experience of failure gave them an added incentive to try harder; being knocked back developed an inner resilience that they had not readily acknowledged or been aware of previously.

Much has been written about the operationalising of a project, but no literature has been sourced on the impact of the learning experience on the individual. As such this study is a contribution towards research in this field and questions whether education on social innovation has improved clinical practice and enhanced the personal and professional development of individuals. The study has shown that in instances unlearning traditional practices and learning new ways of working has improved patient outcomes. Acknowledging that frequently innovation begins at the face of delivery and not necessarily from managers was significant, however it was essential that gaining support from the organisation was key in enabling an innovation to move forward. The ability to take advantage of opportunities and the passion, resilience, competence, confidence and clinical expertise were seen to be essential drivers in providing creative and innovative solutions to delivering a quality, effective service provision to patients in the community. Their learning was considerable and spanned their professional and personal lives. It should not to be forgotten that the outcome of their work was, in many cases, a significant improvement of the quality of care delivered to the patients.

#### Chapter 8 Reflection

8.1 The researcher's personal journey

This is my personal journey through the Professional Doctorate programme and my evolution from being a novice researcher to becoming a competent researcher. It

has been challenging and enjoyable in equal measure – it has defined me as a person and as a professional. It has made me look at research in a whole new way – taking account of the bigger picture and how my research can influence the future of community health professionals through the inclusion of innovation in programmes that I deliver to community health students. It has demonstrated to me that I can perform research at doctoral level as previously I was like many students – the word research filled me with dread. I now have a much deeper understanding of the various research approaches that can be used for social science research.

I sometimes thought that the pressure of working on the professional doctorate programme and working full time in a highly pressurised job at another University would break me. My working life is very stressful and the past five years have presented me with many exasperating challenges at work. Whilst this was difficult, it was also the catalyst that made me continue with my studies.

I began my Professional Doctorate whilst in my final year of the MA in Leadership and Management so from the commencement of the programme I was juggling academic studies with work. I was a student on the first cohort to enrol onto the Professional Doctorate at Westminster University. The first year I studied a reflective module and a professional practice module so I was fairly comfortable with both of these topics. As a qualified nurse reflection is a practice that is encouraged throughout all aspects of our work and as I had only just completed my MA in Leadership and Management I was also relatively confident with the content of the professional practice module.

#### 8.2. Year One

During the first year we were asked to think about our topic for the research project. Since qualifying as a nurse my main interest has been in care of the older person and during my years working as a District Nurse I had become increasingly concerned with the number of staff, both qualified and unqualified, that were visiting vulnerable people in their homes so my initial focus was to look at the quality of service provision for older people in their own homes. I presented my research idea to the research teaching team and they suggested that it might be difficult for me to gain ethical approval to speak to either patients or carers as I was no longer working

as a practitioner and did not have direct access to patients. This threw me slightly so I had to go back to the beginning to look for another research topic that would interest me. At the time I felt very disillusioned and considered whether I should discontinue the course; if I couldn't research an area that I was interested in what was the point in carrying on? However, my resilience and determination to never give up kept me going. I reflected on this incident and how I had felt, and using Gibbs Reflective Cycle (1988) I asked myself if I had to present my research topic again, what would I do. I would take the advice of the teaching team and look at another topic where I could still research an element of quality of care and service provision. I also needed to consider how I could do this so that I would be able to gain ethical approval from both universities.

#### 8.3 Year Two

The second year of the programme concentrated on research and the various research methodologies. It was during this period that I began to recognise and understand the various research methods that I could potentially use. So I spent about six months looking at how I could still carry out a meaningful piece of research that would be able to influence care in the community. I was fortunate that at that time I was recruiting for an exciting programme on social innovation. As the social innovation course evolved and students were coming back to the classroom with ideas and ways of improving the service that was being delivered to patients in the community I had a light bulb moment. I could use students on this programme as the core participants of the study and I would still be able to use my findings to influence care in the community. I spoke to my supervisors about this idea and the view was that this would be suitable for my research. This interruption had delayed my progression but it had been worth it as I was now looking at care in the community across the cross sectorial service provision in the community. Once the ethics had been approved from both universities I was able to commence the data collection. Reflecting on this period of deciding which theoretical research approach to use once again delayed the actual commencement of data collection. As the methods chapter (chapter 3) has suggested, I had contemplated using a number of different research approaches and I think that it was during this period that my greatest learning took place. Recognising that grounded theory, which is commonly

used in health research, would not be appropriate for the length of time that I had to gain the data opened my eyes to how time consuming this approach would be and how it would benefit a longitudinal study. Why then couldn't I use a narrative analysis approach as I wanted to understand the journey that the participants had travelled during their studies? This wouldn't work as I required an element of structure to my interview questions. Surely an ethnographic research approach would be suitable as I had been a District Nurse so I could have 'insider' knowledge to many of the responses. I realised I could not use this approach as I was not observing the participants in practice. Prior to studying for the Professional Doctorate I would not have appreciated the subtle differences between these approaches. Deciding on thematic analysis has once again opened my eyes to an approach to research that, in my opinion, enables the researcher to gently direct questions, but at the same time allows the participants to share their experiences and tell their story in a format that then enabled codes and themes to be identified.

#### 8.4 Professional Development

What have I done with this learning now? I have taught on a research module which I would never have dreamed I would be confident in my knowledge to do so. I have supported colleagues who have wanted to carry out research and have shared my knowledge with them. I realise that when I previously sat on the ethics committee at my university I did not have enough knowledge of the different research approaches that could have impacted on the ethics of an application. I am going to apply to sit on the ethics panel once again as I am sure that I will have an increased ability to offer constructive comments in relation to the research approaches presented.

As I mentioned at the onset of this reflection I have not only grown professionally but also personally. Professionally as I have progressed with the data collection, literature review, data analysis and the writing up of my findings I have become less stressed about the end result. I have learnt so much along my research journey that I am able to take constructive criticism, comments and suggestions from supervisors without feeling a failure. I am in awe of their expertise in research and would aspire to being able to support students in the same way at some stage.

#### 8.5 Personal Development

From a personal point of view, life has thrown many personal challenges my way during this past six years. Even though I have been immersed in reading articles and collecting data these challenges have highlighted what is important in life to me, which is why at times I have had to put my studies for the professional doctorate aside to deal with what I see as my priorities. This has assisted me in 'keeping sane' throughout this period of my life. Recognising that I have had to do more 'juggling' than I have ever had to do before has enabled me to compartmentalise sections of my life with the aim of trying to achieve a work life balance. I am not saying that this has been easy, but my motto in life is that this is not going to beat me but it is not going to make me ill either – from a nursing background I have the philosophy that there is never an educational emergency so I refuse to be a casualty of one through stress. I think I have achieved this over the past six years. However, even though I have said that I have maintained my work life balance there is always a section of my brain that says 'why aren't you working on your professional doctorate?' I am looking forward to no longer hearing that voice in my head.

#### References

Adams, R., Bessant, J., Phelps, R., (2006). Innovation management measurement: A review. *International Journal of Management Reviews.* **Vol.8,** 1, 21-47.

Adams, R., Tranfield, D., Denyer, D. (2011). How can toast be radical? Perceptions of innovations in healthcare, *The International Journal of Clinical Leadership*, 17, 37-48.

Al-Ubaydli, M., (2011). *Personal health records: A guide for clinicians*. Wiley Publishers, New Jersey.

Allen, L., Hetherington, E., Manyama, M., Hatfield, J., Marle, G., (2010). Using the soial entrepreneurship approach to generate innovative and sustainable malaria diagnosis interventions in Tanzania: a case study. *Malaria Journal.* **9:** 42. http://www.malariajournal.com/content/9/1/42 (accessed 2nd June, 2015).

Amo, B., (2006). Employee innovation behaviour in health care: the influence from management and colleagues. *International Nursing Review.* **53**, 231-237.

Andrews, V. Tonkin, E., Lancastle, D., Kirk, M., (2013). Using the Diffusion of Innovations theory to understand the uptake of genetics in nursing practice: identifying the characteristics of genetic nurse adopters. *Journal of Advanced Nursing*, **70** (4), 878 – 893.

Anquillano, F., (2013). Perseverance the Backbone of Nursing. *Research Journal of Social Science & Management*, <a href="www.https://theinternationaljournal.org">www.https://theinternationaljournal.org</a>. (accessed 20<sup>th</sup> August 2017).

Apekey, T., McSorley, G., Tilling, M., Sinwardena, A., (2011). Room for improvement? Leadership, innovation culture and uptake of quality improvement methods in general practice. *Journal Evaluating Clinical Practice*. **17**, 2, 311-318. www.ncbi.nlm.nih.gov (accessed 20<sup>th</sup> February 2015).

Aronson, J., (1994). A Pragmatic view of Thematic Analysis, *The Qualitative Report.* **Vol. 2**, No.1, Spring 1994. (accessed 22 April 2016).

Asselin, M. (2003). Insider research: issues to consider when doing qualitative research in your own setting. *Journal of Nurses Staff Development*. March-April, **19**, (2): 99 – 103.

Ayling, M., (2012). *Mental Health & Chronic Illness*. National Institute for Health Research. Association for Young People's Health. www.https://youngpeopleshealth.org.uk (accessed 20<sup>th</sup> September 2016).

Bacigalupe, G., Askari, S., (2013). E-Health Innovations, Collaboration, and Healthcare Disparities: Developing Criteria for Culturally Competent Evaluation. *Families, Systems & Health, American Psychological Association.***Vol.31,** No.3: 248 - 263.

Baghi, R., (2015). Medical Treatment InThe Digital Age. http://www.studenthtesis.cbs.dk (accessed 15<sup>th</sup> May 2016).

Baldacchino, D.R., (2011). Teaching on Spiritual Care: The perceived impact on qualified nurses. *Nurse Education in Practice*, Vol.11 (1): 47-53.

Bandura, A., (2001). Social Cognitive Theory: An Agentic Perspective. *Annual Review. Psychology*.52: 1-26.

Bandura, A., Locke, E., (2003). Negative Self-Efficacy and Goal Effects Revisited. *Journal of Applied Psychology.* **Vol.88,** 1, 87-99.

Bargh, J., McKenna, A., Fitzsimons, A., (2002). "Can you see the real me? Activation and expression of the 'true self' on the Internet", *Journal of Social Issues.* **58** (1): 33-48.

Beeftink, F., Van Eerde, W., Rutte, C., Bertrand, J., (2012). Being Successful in a Creative Profession: The Role of Innovative Cognitive Style, Self-Regulation, and Self-Efficacy. *Journal of Business Psychology*. 27, 71-81.

Berg, B., (2007). *Qualitative Research Methods for the Social Sciences (6<sup>TH</sup> ed).* London, Pearson.

Berwick, D., (2003). Disseminating Innovations in Health Care, *JAMA*, April 16; 289 (15): 1969 – 1975 http://www.jamanetwork.com (accessed 15<sup>th</sup> May 2016).

Beshears, J. & Gino,F., (2015). Leaders as Decision Architects: Structure Your Organisation's Work to Encourage Wise choices. *Harvard Business Review 93*. (May 2015): 52 – 62. http://www.hbs.edu.faculty (accessed 15<sup>th</sup> May 2016).

Bessant, J., Kunne, C., Moslein, K., (2012). *Opening up healthcare innovation: Innovation solutions for a 21<sup>st</sup> century healthcare system*. Advanced Institute of Management Research. London.

Bettencourt, L., Lobo, J., Strumsky, D., West, G., (2010). Urban Scaling and its Deviations: Revealing the Structure of Wealth, Innovation and Crime across Cities. https://doi.org/10.1371/journal.pone.0013541 (accessed 10th May 2015).

Bhaskar, R., (2008). *A Realist Theory of Science with a New Introduction*. Routledge, London.

Bird, J., Roddick, G., (1991). *The Big Issue*, <u>www.https://thebigissue</u> (accessed 20<sup>th</sup> August 2014).

Blackwell, W., (2014). What inspired you to become a nurse? www.carolinashealthcare.org.body (accessed 20<sup>th</sup> May 2017).

Blaikie, N., (2011). Designing Social Research 2<sup>nd</sup> Edition, Cambridge, Polity Press.

Blaut, J., (1993). The Colonizer's Model of the World: Geographical Diffusionism and Eurocentric History. The Guilford Press, New York. Cited in McMaster, T., Wastell, D., (2005). Diffusion – or delusion? Challenging an IS research tradition. *Information Technology & People.* Vol.18, No.6, 383-404.

Block, D., (2013). Disruptive Innovation: Contributing to a Value-based Health Care System. *Innovation* Primo: 20<sup>th</sup> October, 2014.

Borrelli, B., (2010). Smoking Cessation: Next Steps for Special Populations Research and Innovative Treatments. *American Psychological Association*, **Vol. 78**, No. 1: 1-12.

Bornstein, D., (2007). How to change the world: Social Entrepreneurs and the power of new ideas, Oxford University Press, Oxford.

Boyatzis, R. E., (1998). *Transforming qualitative information: Thematic analysis and code development.* Thousand Oaks, CA: Sage.

Braun, V., & Clarke, V., (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3 (2): 77-101. www.tandfonline.com.doi.abs (accessed 16<sup>TH</sup> May 2016).

Braun, V. & Clarke, V., (2013). Successful qualitative research: A practical guide for beginners. London: Sage.

Braun, V., & Clark, V., (2014). What can "thematic analysis" offer health and wellbeing researchers? *International Journal of Health & Well- being Studies*, **Vol 9**, 10. <a href="http://www.3402/qhw.v9.26152">http://www.3402/qhw.v9.26152</a> (accessed 16<sup>th</sup> May 2015).

British Medical Association, (2016). *Poor working conditions affect patient care, suggest report.* https://.www.bma.org.uk (accessed 8<sup>th</sup> September 2016).

Brooks, H., Pilgrim, D., Rogers, A., (2011) Innovation in mental health services: what are the key components of success? *Implementation Science*.**6:** 120. http://implementationscience.com/content/6/1/120 (accessed 19<sup>th</sup> October 2015).

Burke, E., (1790). *Reflections on the Revolution in France,* cited in Godin, B., (2013). Innovation after the French Revolution, or, Innovation Transformed; From Word to Concept. *Project on the Intellectual History of Innovation.* Working Paper No. 14. <a href="https://cslic.ca">https://cslic.ca</a> (accessed 10<sup>th</sup> May 2015).

Burke, W., (2002). *Organisational Change: Theory and Practice,* Sage Publications, Thousand Oaks, California.

Butler, P., (2011). Great Ormond Street hospital issues apology to Baby P Whistleblower. *The Guardian*, 14<sup>th</sup> June, 2011. <a href="https://theguardian.com">www.https://theguardian.com</a> (accessed 20<sup>th</sup> September 2015).

Butterworth, T., (2013). Our Nurse-Led Approach to Productivity Must Live On. *Nursing Standard*, Vol 27, No30, 14-15.

Carr, S., Clarke, C., (2010). The manager's role in mobilising and nurturing development: entrenched and engaged approaches to change. *Journal of Nursing Management.* **18:** 332-338.

Cash, K., (2000). Do we need managers? *Nurse Education Today,* Harcourt Publishers, 2**0**, 257-258.

Chaudoir, S. Dugan, A. Barr, C., (2013). Measuring factors affecting implementation of health innovations: a systematic review of structural, organisational, provider, patient, and innovation levels. <a href="http://www.implementationscience.com/content/81/22">http://www.implementationscience.com/content/81/22</a> (accessed 15<sup>TH</sup> May 2016).

Christie & Co., (2015). The UK Nursing Workforce Crisis or Opportunity? *Market Insight Report*. https://www.christie.com (accessed 13<sup>th</sup> May 2016).

Clark, M., (2009). Social Business, Department of health, London.

Cohen, L., Mannion, L., Morrison K., (2007). *Research Methods in Education, 6<sup>th</sup> Edition*, London, Routledge Falmer.

Cooley, M., (2008). Nurses' motivations for studying third level post-registration nursing programmes and the effects of studying on their personal and work lives. *Nurse Education Today*, **28**, 588-594.

Cooper, J., (2010). Promoting innovation. Editorial. *Gastrointestinal Nursing*. **Vol.8**, 4, 5.

Cope, V., Jones, B., Hendricks, J., (2016). Resilience as resistance to the new managerialism: portraits that reframe nursing through quotes from the field. *Journal of Nursing Management*. **Vol.24**, 1, 115-122.

Corbin Dwyer, S., Buckle, J., (2009). The Space Between: On Being an Insider-Outsider in Qualitative Research. *International Journal of Qualitative Methods.* **8,** 1, 54-63.

Cormican, K., O'Sullivan, D., (2004). Auditing best practice for effective product innovation management, 'Technovation' **24**, 819-829, cited in Paterson, F., Kerrin, M., Roissard, G., (2009). Characteristics & Behaviours of Innovative People in Organisations: Literature Review, *NESTA Policy and Research Unit*, City University, London.

Cornwall, J., (2014). The future of primary healthcare in Australia: Where to from here? *The Australasian medical journal*. <a href="https://ncbi.nlm.nih.gov">https://ncbi.nlm.nih.gov</a> (accessed 24<sup>th</sup> May 2016).

Cresswell J., (2007). Qualitative inquiry and research design; choosing among five approaches Second Edition, Thousand Oaks, CA: Sage.

Cresswell, J., Clark, V., (2011). *Designing and Conducting Mixed Methods Research, Second Edition,* Thousand Oaks, CA: Sage.

Creasey, T., (2007) *Defining Change Management*. <a href="http://www.change-management.com/prosci-defining-change-management">http://www.change-management</a>. (accessed 10th January 2016).

Cusack, L., Smith, M., Hegney, D., Rees, C., Breen, L., Witt, R., Rogers, C., Williams, A., Cross, W., Cheung, K., (2016). Exploring Environmental Factors in Nursing Workplaces That Promote Psychological Resilience: Constructing a Unified Theoretical Model. *Frontiers in Psychology.* 7, 600-609.

Darzi, A., (2008). *High Quality Care for All – Year of Care* <a href="https://www.yearofcare.co.uk">https://www.yearofcare.co.uk</a> (accessed 12<sup>th</sup> May 2015).

Davis, C., (2011). Turning on nursing innovation to improve care and efficiency, *Nursing Management*, Vol 18, No.3, 6-7.

Deakin, H., Wakefield, K., (2013). Skype interviewing: reflections of two PhD researchers, *Qualitative Research Journal*, **0** (0): 1-14.

Dean, E., (2011). Standardised care and innovation helps slash pressure ulcer rates, *Nursing Standard*, **Vol. 26**, No.5: 8.

DeGraff, J., Lawrence, K., (2002). *Creativity at Work: Developing the Right Practices to Make Innovation Happen.* Jossey-Bass, San Fransisco.

Denis, J., Hebert, Y., Langley, A., Lozeau, D., Trottier, L., (2002). Explaining Diffusion Patterns for Complex Health Care Innovations. *Health Care Management Review*, **27** (3) 60-73.

Department of Health, (2006). *Our Health, Our Care, Our Say: a new direction for community services,* London, Department of Health.

Department of Health, (2009). Working to put people first: The strategy for the adult social care force in England, Department of Health, London.

Department of Health, (2010a). *High Quality Care for All, Next Stage Review,* Darzi Report, London, Department of Health.

Department of Health, (2010b). *Equity and Excellence; Liberating the NHS (White Paper)*, Crown, London.

Department of Health, (2010c). Leading the way through social enterprise; the Social Enterprise Pathfinder Programme evaluation, Crown, London.

Department of Health, (2011a). A Guide to the Right to Provide, Crown, London.

Department of Health, (2011b). *Innovation, Health and Wealth: Accelerating Adoption and Diffusion in the NHS*. Department of Health, London.

Department of Health, (2011c). *Transforming Community Services Demonstrating and Measuring Achievement: Community Indicators for Quality Improvement,*Department of Health, Leeds.

Department of Health, (2011d). *Innovation Health and Wealth, Accelerating Adoption and Diffusion In The NHS*. Department of Health, NHS Improvement & Efficiency Directorate, Innovation and Service Improvement, Leeds.

Department of Health, (2011e). *National End of Life Care Programme*, www.https://webarchive.nationalarchives.gov.uk (accessed 20<sup>th</sup> May 2016).

Department of Health, (2011f). *Creating Change – Innovation Health & Wealth One Year On.* Department of Health, Leeds.

Department of Health, (2012a). *Sir David Nicholson sets out next steps for transition*. http://www.gov.uk (accessed 2<sup>nd</sup> April 2015).

DH (2012b) NHS Chief Executive's Review of Innovation in the NHS Summary of the responses to the Call for Evidence and Ideas, <a href="http://www.dh.gov\_(accessed 10th">http://www.dh.gov\_(accessed 10th)</a> May, 2015).

Department of Health, (2014). *The Five Year Forward View,* Department of Health, Leeds.

Department of Health, (2012c). Social enterprises to receive £19million investment over next year announced, www.gov.uk (accessed 10<sup>th</sup> May 2015).

Department of Health, (2015a) 2010-2015 Government policy; research and innovation in health and social care, www.gov.uk (accessed 11th May, 2015).

Department of Health, (2015b). *Framework for Commissioning Community Nursing – NHS England*, Department of Health, Leeds.

Department of Health, (2017). Innovation in regulation. Department of Health, Leeds.

Department for Innovation, Universities and Skills, (2008). Department for Innovation, Universities and Skills departmental report. <a href="https://www.gov.uk">https://www.gov.uk</a> (accessed on 20<sup>th</sup> August 2014).

Dervojeda, K., Schretlen, J., (2013). Innovation How to convert research into commercial success story? *European Commission*. Price Waterhouse Coopers EU Services EESV, Belgium. https://ec.europa.eu (accessed 17<sup>th</sup> August 2017).

Dopson, S., Fitzgerald, L., Ferlie, E., Gabbay, J., Locock, L., (2010). No magic targets! Changing clinical practice to become more evidence based. *Health Care Management Review*, **27** (3) 35-47.

Dorning, H., Bardsley, M., (2014). Focus on: Allied Health Professionals – can we measure quality of care? *Quality Watch*, The Health Foundation, Nuffield Trust, <a href="https://www.qualitywatch.org.uk">www.qualitywatch.org.uk</a> (accessed May 9<sup>th</sup> 2016).

Douglas, M. Overstreet, R., Hazen, B., (2016). Art of the possible or fool's errand? Diffusion of large-scale management innovation. *Business Horizons, Science Direct* <a href="http://dx.doi.org/10.1016/j.bushor.2016.03.002">http://dx.doi.org/10.1016/j.bushor.2016.03.002</a> (accessed January 10<sup>th</sup> 2017).

Doyle, C., Howe, C., Woodcock, T., Myron, R., Phekoo, K., McNicholas, C., Saffer, J., Bell, D., (2013). *Making change last: applying the NHS institute.*<a href="https://implementationscience.biomedcentral.com">https://implementationscience.biomedcentral.com</a> (accessed 8<sup>th</sup> June 2016).

Doyle, C., (2013) Applying the NHS for innovation and improvement sustainability model to healthcare improvement. *Implementation Science*, 8: 127. <a href="https://www.researchgate.net">https://www.researchgate.net</a> (accessed 10<sup>th</sup> May 2016).

Doyle, G., Garrett, B., Currie, L., (2014). Integrating mobile devices into nursing curricula: Opportunities for implementation using Rogers' Diffusion of Innovation model. *Nurse Education Today.* 34, 775-782.

Driedger, S., Kothari, A., Graham, I., Cooper, E., Crighton, E., Zahab, M., Morrison, Sawada, M., (2010). If you build it, they still may not come: outcomes and process of implementing a community-based integrated knowledge translation mapping innovation. *Implementation Science*. **5.** 47: 1-13.

Drucker, P., (1957). America's Next Twenty Years. Harper & Brothers, New York.

Edward, R., Holland, J., (2013). *What is Qualitative Interviewing?* University of Edinburgh, *www.https://eprints.ncrm.ac.uk* (accessed 8<sup>th</sup> September 2015).

Ellison, N., Heino, R., Gibbs, J. (2006). Managing impressions online: self-presentation processes in the online dating environment. *Journal of computer-mediated communication*. 11, 415-441.

Evans, K., George, N., White, K., (2010). *Ensuring that all children and young people make sustained progress and remain fully engaged through all transitions between key stages.* National Foundation for Educational Research. <u>www.https://nfer.ac.uk</u> (accessed 16<sup>th</sup> May 2016).

Fanning, A., (2013). Social Innovation and Enterprise, cited in *Community and Public Health Nursing* (2013). Sines, D., Aldridge - Bent, S., Fanning, A., Farrelly, P., Potter, K., Wright, Wiley Blackwell, Oxford.

Feig, B., (2006). *Hot Button Marketing*. Online Research Library: Questia. <a href="https://questia.com">www.https://questia.com</a> (accessed 2<sup>nd</sup> June, 2016).

Ferner,R., (2010). The Neurofibromatoses, *Practical Neurology*, www.https://group.bmj.com (accessed 20<sup>th</sup> May 2016).

Fiedler, R., Giddens, J., North, S., (2014) Faculty Experience of a Technological Innovation in Nursing Education. *Nursing Education Perspectives*. **Vol. 35**, No. 6: 387 – 391.

Findlay, L., (2012). Number of social enterprises could be half government estimate, *Third sector*, Haymarket Business Media, <u>www.thirdsector.co.uk</u> (accessed 2<sup>nd</sup> June, 2016).

Fitzgerald, L., Ferlie, E., Hawkins, C., (2003). Innovation in healthcare: how does credible evidence influence professionals? *Health and Social Care in the Community*. **11**, (3): 219 – 228.

Fletcher, A., (2016). Applying critical realism in qualitative research: methodology meets method. *International Journal of Social Research Methodology.* Vol 20, No.2. Taylor & Francis online http://tandfonline.com (accessed 29<sup>th</sup> October 2016).

Frankel, A., (2008). What leadership styles should senior nurses develop? *Nursing Times*, 104, 35, 23-24.

Frymere, B., (2006). The search for talent why it's getting harder to find, *The Economist*, 7<sup>th</sup> October, 2006. <a href="https://www.economist.com/">https://www.economist.com/</a> (accessed 18<sup>th</sup> June 2016).

Gallagher, L., (2006). Continuing Education in Nursing: A concept analysis. *Nurse Education Today*, **27**, 466-473.

Gardener, A., Hase, S., Gardner, G., Dunn, S., Carryer, J., (2008). From competence to capability: a study of nurse practitioners in clinical practice. *Journal of Clinical Nursing*. **Vol.17**, 2, 250-258.

Gardener, C., Acharya, T., Yach, D. (2007). Technological and social innovation: a unifying new paradigm for global health. *Health Affairs*.**26**, 4, 1052-1061.

Gbrich, C. (1999). Qualitative Research in Health: An Introduction, Allen & Unwin, Australia.

Geibert, R., (2006). Using Diffusion of Innovation, Concepts to Enhance Implementation of an Electronic Health Record to Support Evidence-based Practice. *Nursing Administration Quarterly*, July-September 2006, **Vol. 30**, No.3. 203-210.

Gerrish, K., McDonnell, A., Nolan, M., Guillaume, L., Kirshbaum, M., Tod, A., (2011). The role of advanced practice nurses in knowledge brokering as a means of promoting evidence-based practice among clinical nurses. *Journal of Advanced Nursing*. **Vol.67**, No.9. 2004-2014.

Gibbs, G.,(1988) Learning by Doing: A Guide to Teaching and Learning Methods. Oxford Further Education Unit, Oxford.

Gibson, R., Kelly, B., (2010). Stoking Your Innovation Bonfire: A Road Map to a Sustainable Culture as Ingenuity and Purpose. *Innovation in Nursing Practices*, **9**, 3, 1217-1228 <a href="https://researchgate.net">www.https://researchgate.net</a> (accessed 16<sup>th</sup> May 2016).

Glaser, B. G., & Strauss, A. L. (1965). Awareness of dying. Chicago Illinois: Aldine.

Gliddon, P., Powell, M., (1952). *Called to Serve,* Hodder & Stoughton, London. Cited in Harris, R., Bennett, J. Ross, F., (2013). Leadership and innovation in nursing seen through a historical lens, *Journal of Advanced Nursing,* November, 2013, 1-10.

Godin, B., (2013). Innovation after the French Revolution, or, Innovation Transformed; From Word to Concept. *Project on the Intellectual History of Innovation*. Working Paper No. 14. https://cslic.ca (accessed 10<sup>th</sup> May 2015).

Godin, B., (2015). *Innovation Contested: The Idea of Innovation over the Centuries,* Routledge, Political Science, London.

Godin, S., (2005). *Purple Cow: Transform Your Business by Being Remarkable,* Portfolio Hardcover, Penguin, London.

Goldenberg, M., Kamoji, W., Orton, L., Williamson, M., (2009). *Social Innovation in Canada: An Update,* Canadian Policy Research Networks, Ontario.

Goldsmith, S., (2010). The Power of Social Innovation: How civic entrepreneurs ignite community networks for good, Jossey-Bass, San Fransisco.

Gomes, B., Calanzani, N., Higginson, J., (2011). *Local preferences and place of death in regions within England 2010.* National End of Life Care Intelligence Network. www.https://endoflifecare-intelligence.org.uk (accessed 20<sup>th</sup> May 2016).

Goodwin, N., Smith, J., Davies, A., Perry, C., Rosen, R., Dixon, A., Dixon, J., Ham, C., (2013). *Integrated care for patients and populations: Improving outcomes by working together,* The Kings Fund & Nuffield Trust, London.

Gopalasamy, G., (2010).Innovation study – Report, Evidence for Innovation as Evolution model, *Centre for Innovation in Health Management*.

<a href="http://www.cihm.leeds.ac.uk/new/international-network/social-enterprise-and-health/">http://www.cihm.leeds.ac.uk/new/international-network/social-enterprise-and-health/</a>. (accessed 6th June 2015).

Goss, K., (1979). Consequences of Diffusion of Innovations, *Rural Sociology.* **44**, 4, 734-772.

Gosselin, T., Ireland, A., Newton, S., O'Leary, C., (2015). Practice Innovations, Change Management, and Resilience in Oncology Care Settings. *Oncology Nursing Forum*. www./https://ncbi.nlm.nih.gov (accessed 18<sup>th</sup> August 2017).

Government Agency, (2010). *The Big Society*, London. <a href="https://www.gov.uk">https://www.gov.uk</a> (accessed on line 10<sup>th</sup> May, 2015).

Green, J., Thorogood, N., (2004). *Qualitative Methods for Health Research,* London: Sage.

Guest, G., MacQueen, K. M., & Namey, E. E., (2012). *Applied thematic analysis*. Thousand Oaks, CA: Sage.

Haider, M., Kreps, G., (2004). Forty Years of Diffusion of Innovations: Utility and Value in Public Health. *Journal of Health Communication*, **Vol 9**, 3-11.

Hails, C., Steadman, J., Walls, J., (2010). *PACE: Pushing the Boundaries of Community Care – An Evaluation of Bromley Post- Acute Care Enablement Service July 2009 – April 2010*, Bromley Community Provider Unit, NHS Bromley, Kent.

Haggman, S., (2009). Functional actors and perceptions of innovation attributes: influence on innovation adoption *European Journal of Innovation Management*, **Vol.12**, 3, 327-348.

Ham, C., Baird, B., Gregory, S., Jabbal, J., Alderwick, H., (2015). *The NHS under the coalition government, Part one: NHS reform,* The Kings Fund, London.

Hammersley, M., (2012). What is Qualitative Research? Bloomsbury Academic, London.

Hanna, P., (2012). Using internet technologies (such as Skype) as a research medium: a research note. *Qualitative Research* **12** (2): 239-242.

Harper, E., Thompson, A., (2012). Qualitative Research methods in Mental health and Psychotherapy, Wiley, Blackwell, New Jersey.

Harris, R., Bennett, J., Ross, F., (2013). Leadership and innovation in nursing seen through a historical lens. *Journal of Advanced Nursing*.**00**, 1-10.

Harris, M., Weisberger, E., Silver, D., Dadwal, V., Macinko, J., (2016). That's not how the learning works – the paradox of Reverse Innovation: a qualitative study. *Globalisation and Health.* 12. 36. <a href="https://doi.org/10.1186/s12992-016-0175-7">https://doi.org/10.1186/s12992-016-0175-7</a> (accessed 7<sup>th</sup> August 2016).

Hayes, K., Eljiz, K., Dadich, A., Fitzgerald, J., Sloan, T., (2014). Trialability, observability and risk reduction accelerating individual innovation adoption decisions. *Journal of Health Organisation and Management*, Vol.29, No.2, 271-294.

Heathfield, S., (2016). You Can Manage Resistance to Change. *Communication at Work*. <a href="https://susanheathfield.com">https://susanheathfield.com</a> (accessed 31st August 2017).

Hellstrom, T., Hellstrtom, C., Berglund, H., (2002). The innovating self: exploring self among a group of technological innovators. *Journal of Managerial Psychology*, **Vol.17**, 4, 267-286.

Herold, D., Fedor, D., (2008). *Leading Change Management: Leadership Strategies That Really Work.* Kogan Page, London.

Holland, K., Lauder, W., (2012). A review of evidence for the practice learning environment: Enhancing the context for nursing and midwifery care in Scotland. *Nurse Education in Practice*, 12, 60-64.

Holland, M., Tucker, I., Mark, M., Kelly, A., Honigsbaum, O., (2012). Africa innovations: 15 ideas helping to transform a continent. *The Guardian*, 26<sup>th</sup> August, 2012. www.https://theguardian.com (accessed 15<sup>th</sup> September 2015).

Hubert, A., (2010). *Empowering People, driving change: Social Innovation in the European Union, innovations in primary care*. <a href="www.https://oecd.org">www.https://oecd.org</a> (accessed 22<sup>nd</sup> August 2014).

Hughes, F., (2006). Nurses at the forefront of innovation. *International Nursing Review.* **53:** 94 – 101.

Hunter, B., Warren, L., (2013). *Midwives' experiences of workplace resilience*. www.https://midwiferyjournal.com (accessed 2<sup>nd</sup> June 2017).

International Council of Nurses, (2009). *Delivering Quality, Serving Communities, Nurses Leading Care Innovations*. International Council of Nurses, Geneva.

Ionescu, C., (2015). About the conceptualization of social innovation. *Theoretical and Applied Economics.* **Vol. 22,** No.3: 53-62.

Jahanmir, S., Lages, L. (2015). The late-adopter scale: A measure of late adopters of technological innovations. *Journal of Business Research*, http://dx.doi.org/10.1016/j.jbusres.2015.10.041 (accessed 18th May 2016).

Janghorban, R., Roudsari, R., Taghipour, A., (2014). Skype interviewing: the new generation of online synchronous interview in qualitative research, *International Journal of Qualitative Studies on Health and Well-being*, *9:* 1-9.

Jiwa, M., (2013). Innovating for General Practice. *The Australasian Medical Journal*. **6,** 1, 36-40.

Jiwa, M., McManus, A., Dadich, A., White, J., Rieck, A., Razmi, S., (2013). Harnessing information technology to innovate in primary care. *Quality in Primary Care*.21, 43-49.

Joffe, H., (2011). Thematic analysis. In D. Harper & A. R. Thompson (Eds.), Qualitative methods in mental health and psychotherapy: A guide for students and practitioners: 209 – 223, Chichester: Wiley.

Jordan, A., Huitema, D., (2014). Innovation in climate policy: the politics of invention, diffusion, and evaluation. *Environmental Politics*, **Vol 23**, No.5, 715-734.

Jones, J., (2012). Manage workplace chaos by building your resilience. <u>www.journals.rcni.com</u> (accessed 29<sup>th</sup> April 2017).

Joseph, M. (2015). Organisational Culture and Climate for Promoting Innovativeness. *The Journal of Nursing Administration*, Vol.45, No. 3, p.172-178.

Kaya, N., Turan, N., Aydin, G., (2015). A Concept Analysis of Innovation in Nursing. *Social and Behavioral Sciences*. 195, 1674-1678.

Kendall-Raynor, P., (2013). Innovation institute leave legacy of a more efficient health service, *Nursing Standard*, **Vol 27**, No.30, 12-13.

Keogh, B., (2015). Review of elective waiting time targets, *Health Service Journal*, 3 June, 2015.

Knight, J., (2010). Promoting Innovation, *Nursing Standard*, Vol. 25, No. 8, p 10-11.

Korf, B., Rubenstein, A., (2005). *Neurofibromatosis: A Handbook for Patients, Families, and Health Care Professionals (2<sup>nd</sup> edition),* Thieme Medical Publishers, New York.

Korukcu, O., Kukulu, K., (2010). Innovation in nursing education. *Procedia Social and Behavioral Science*, 9, 369-372.

Krueger, R., Casey, M., (2009). Focus Groups: A Practical Guide for Applied Research, 4<sup>th</sup> Edition, Sage, London.

Kumar R., Uzkurt, C., (2010). Investigating the effects of self- efficacy on innovativeness and the moderating impact of cultural dimensions. *Journal of International Business and Cultural Studies*. 1-15.

Laforet, S., Tann, J., (2006). Innovative characteristics of small manufacturing firms. *Journal of Small Business and Enterprise Development*, **Vol.13**, No.3, 363-380.

Lambrou, P., Kontodimopoulos, N., Niakas, D., (2010). Motivation and job satisfaction among medicine and nursing staff in a Cyprus public general hospital, *Human Resources for Health*, 8, 1-26.

Lansley, A., (2012). *NHS Reforms Live blog – Monday 13<sup>th</sup> February, 2012.* <a href="https://www.theguardian.com">https://www.theguardian.com</a> (accessed 20<sup>th</sup> February 2015).

Leggott, K., Martin, M. Sklar, D., Hellitzer, D.,Rosett, R., Crandall, C., Vagh, F., Mercer, D., (2015). Transformation of anesthesia for ambulatory orthopedic surgery: A mixed-methods study of a diffusion of innovation in healthcare. *Healthcare*. 1-7. <a href="https://www.elsevier.com/locate/hjdsl">www.elsevier.com/locate/hjdsl</a> (accessed 12<sup>th</sup> May 2015).

Lennan, E., McPhelim, J., (2012). A snapshot of innovation: chemotherapy delivery in the UK. *British Journal of Nursing*. <a href="http://dx.doi.org/10.12968/bjn.2012.21.Sup4.S12">http://dx.doi.org/10.12968/bjn.2012.21.Sup4.S12</a> (accessed 29<sup>th</sup> April 2017).

Leonard, L., McCutcheon, K., Rogers, K., (2015). In touch to teach: Do nurse educators need to maintain or possess recent clinical practice to facilitate student learning? *Nurse Education in Practice*. 16, 148-151.

Lin, C., Blake, J., Keppel, G., Dobie, S., Osborn, J. Cole, A., Baldwin, L., (2016). Using the diffusion of innovations theory to assess socio-technical factors in planning the implementation of an electronic health record alert across multiple primary care clinics. *Journal of Innovation in Health Informatics*, **Vol 23**, No.1, 1-9.

Lindsey, B., (2008). Looking at Positive Behavior Interventions and Supports through the Lens of Innovations Diffusion. *The Innovation Journal: The Public Sector Innovation Journal*. **Volume 13**, 2, 1-18.

Litalien, D., Morin, A., Gagne, m., Vallerand, R., Losier, G., Richard, R., (2017). Evidence of a continuum structure of academic self- determination: A two-study test using a bi-factor-ESM representation of academic motivation, *Contemporary Educational Psychology.* Vol. 51, 67-82.

MacFadyen, J., (2015).Innovation: An Open Access, Equal Opportunity Invitation for Diversity. *Innovation and Holistic Care.* **29,** 1, 3-5.

Madge, C., O'Connor, H. (2004). Online methods in geography educational research. *Journal of Geography in Higher Education*, 28 (1): 143-152

Malloch, K., Melnyk, B., (2013). Developing High-Level Change and Innovation Agents: Competencies and Challenges for Executive Leadership. *Nursing Administration Quarterly*. **Vol.37**, 1, 60-66.

Mann, S., (2011) A critical Review of Qualitative Interviews in Applied Linguistics, Oxford University Press, Oxford.

Martin, R., Osberg, S., (2015). *Getting Beyond Better: How Social Entrepreneurship Works*. Goodreads. www.https://goodreads.com (accessed 3<sup>rd</sup> June 2016).

Masten, A., (2001). Ordinary Magic: Resilience Processes in Development. *American Psychologist.* **Vol 56,** 3, 227- 238.

McEwan, A., Tsey, K., McCalman, J., Travers, H., (2010). Empowerment and change management in Aboriginal organisations: a case study. *Australian Health Review*, **34**, 360-367.

McLean, L., (2005). Organizational Culture's Influence on Creativity and Innovation: Review of the Literature and Implications for Human Resource Development.

Advances in Developing Human Resources, Vol.7, 2, 226-246.

McMaster, T., Wastell, D., (2005). Diffusion – or delusion? Challenging an IS research tradition. *Information Technology & People.* **Vol.18**, No.6, 383-404.

McSherry, R. & Douglas, M., (2011). Innovation in nursing practice: a means to tackling the global challenges facing nurses, midwifes and nurse leaders and managers in the future. *Journal of nursing management*, 19 (2), 165-169.

Merrium Webster Online Dictionary. <a href="http://www.merriam-webster.com">http://www.merriam-webster.com</a> Accessed August 10<sup>th</sup> 2017.

Miller, D., Le-Breton-Miller, I., (2017). Sources of Entrepreneurial Courage and Imagination: Three Perspectives, Three Contexts. *Entrepreneurship Theory and Practice*. **Vol 41,** 5, 667-675.

Miller J., Glassner, B. in Silverman, D., (2011). *Qualitative Research 3<sup>rd</sup> Edition,* Sage, London.

Mobaraki, M., Mohageri, A., Karami, Y., (2012). Identifying the Factors That Affect the Success of Self-Efficiency and Self-Employment Plans among the Clients of Imam Knomeini's Relief Committee through the Lens of Social Entrepreneurship. *International Journal of Business and Management*, **Vol.7**, No.11, 43-55.

Mullahy, C., Boling, J., (2008). Issues and Interventions. *Professional Case Management*. **Vol 13**, No.5: 286-289.

Mulgan, G., (2007). Social Innovation: what it is, why it matters and how it can be accelerated. *Skoll Centre for Social Entrepreneurship*, Said Business School, University of Oxford.

Murray, R., Caulier-Grice, J., Mulgan, G., (2010). *The Open book of social innovation,* The Young Foundation and The National Endowment for Science, Technology and the Arts, London.

National end of Life Care Intelligence Network, (2012). www.https://endoflifecare-intelligence.org.uk (accessed 20<sup>TH</sup> May 2016).

National Health Service, (2009a). *NHS Institute for Innovation and Improvement – Annual Report and Accounts for the period 1 April – 31 March, 2009.* https://www.gov.uk (accessed 10<sup>th</sup> January 2015).

National Health Service, (2009b). The NHS Constitution. <a href="https://www.webarchive.nationalarchives.gov.uk">https://www.webarchive.nationalarchives.gov.uk</a> (accessed 25<sup>th</sup> May 2015).

National Health Service Confederation, (2009c). Making sense of the new innovation landscape, *Health Services Research Network,* The NHS Confederation, London.

National Health Service, (2011). The NHS Institute for Innovation and the Improvement and National Innovation Centre, London.

National Health Service Leadership Academy, (2013). Healthcare Leadership Model, The nine dimensions of leadership behaviour. <a href="www.leadershipacademy.nhs.uk">www.leadershipacademy.nhs.uk</a> (accessed 2<sup>nd</sup> June 2016).

National Health Service, (2014). *Creating Change Innovation Health & Wealth One Year On.* NHS, London.

National Health Service, (2015). *Integrated Clinical Entrepreneur Training – an overview*, NHS England, Leeds.

National Health Service, (2015a). NHS Innovation Accelerator, <a href="https://www.england.nhs.uk">https://www.england.nhs.uk</a> (accessed 6<sup>th</sup> January 2017).

National Health Service (2015) *Statistics on Alcohol – NHS Digital.* www.https://digital.nhs.uk (accessed 21st May 2016).

National Health Service, (2017) Monthly newsletter, <a href="https://www.ourhealthiersel.nhs.uk">https://www.ourhealthiersel.nhs.uk</a> (accessed 2<sup>nd</sup> June 2017).

National Patient Safety Agency, (2010). <a href="www.npsa.nhs.uk">www.npsa.nhs.uk</a> (accessed 12<sup>th</sup> May 2015).

Nelson, M., Brice, J., Gunby, N., (2010). Innovation Adoption Decisions: The Effect of Problem Solving Styles and Social Support. *Journal of Organizational Culture, Communications and Conflict*, **Vol. 14,** 1, 73-88.

Nesbitt, J., (2013). Journal clubs: A two-site case study of nurses' continuing professional development. *Nurse Education Today.* **Vol 33** (8): 896-900.

Newton, N., (2010). *The Use of semi-structured interviews in qualitative research:* strengths and weaknesses. <a href="https://www.scienceopen.com">https://www.scienceopen.com</a>. (accessed 26<sup>th</sup> May 2016).

Ng.T., Lucianetti, L., (2015). Within-Individual Increases in Innovative Behaviour and Creative, Persuasion, and Change Self-Efficacy over time: A Social-Cognitive Theory Perspective. *Journal of Applied Psychology*. http://dx.doi.org/10.1037/ap10000029 (accessed 10th January 2016).

Nieboer, A., (2012). Innovative culture in long-term care settings: the influence of organizational characteristics. *Health Care Management Review,*<a href="http://www.researchgate.net/publication/51458713">http://www.researchgate.net/publication/51458713</a> (accessed 19th June 2015).

Obholzer, A., Roberts, V., (1994). *The Unconscious at Work: Individual and Organizational Stress in the Human Services*. Taylor and Francis, <a href="http://www.amazon.co.uk/exec/obidos/ASIN/0415102065/qid=1086790397/sr=1-1/ref=sr\_1\_2\_1/026-3276884-8882052">http://www.amazon.co.uk/exec/obidos/ASIN/0415102065/qid=1086790397/sr=1-1/ref=sr\_1\_2\_1/026-3276884-8882052</a> (accessed 2nd June 2016).

Oborn, E., Barrett, M., Darzi, A., (2011). Robots and service innovation in health care. *Journal of Health Services Research & Policy.* **Vol.16,** 1, 46-50.

O'Connor, H., Madge, C., Shaw, R., (2008). Internet-based interviewing. In: Fielding, N., Lee, N. and Blank, G. (eds). *The Sage Handbook of Online Research Methods*. London: Sage, 271-289.

OECD, (2013). *Entrepreneurship at a Glance*. <a href="https://www.oecd-library.org-entrepreneurship">https://www.oecd-library.org-entrepreneurship</a> (accessed 28<sup>th</sup> May 2016).

OECD, (2015). The Innovation Imperative in the Public Sector: Setting an Agenda for Action, *Executive Summary* <a href="https://wwwoecd.org.publications">https://wwwoecd.org.publications</a> (accessed 28<sup>th</sup> May 2016)

O'Leary, K., (2011). Innovation by example, Nursing Standard, Vol. 25, no.37: 22-23.

Ovretveit, J., Andreen-Sachs, A., Carlsson, J., Gustafsson, H., Hansson, J., Keller, C., Lofgren, S., Mazzocato, P., Tolf, S. Brommels, M., (2012). Implementing organisations and management innovations in Swedish healthcare: Lessons from a comparison of 12 cases. *Journal of Health Organisation and Management*, Vol. 26, 2: 237-257.

Oxford Dictionary, (2011). *The Oxford Mini English Dictionary*, Oxford University Press, Oxford.

Parliamentary and Health Service Ombudsman, (2016). *Vulnerable patients and their families suffering harrowing ordeals due to poor hospital discharge,* May 2016, https://.www.ombudsman.org.uk (accessed 9<sup>th</sup> September, 2016).

Patterson, F., Kerrin, M. Gatto-Roissard, G., (2009). Characteristics and Behaviours of Innovative Behaviour in the Workplace, <a href="https://www.nesta.org.uk">https://www.nesta.org.uk</a> (accessed 28<sup>th</sup> May, 2016).

Patton, M. (1990). *Qualitative Evaluation and Research Methods, Second Edition,* Sage, London.

Patton, M., (2011). Developmental Evaluation: Applying Complex-ity Concepts to Enhance Innovation and Use. *The Canadian Journal of Program Evaluation*. **Vol.26**, No.2; 108 – 110.

Pawson, R., (2003). *Critical appraisal: assessing research quality*, Social Care Institute for Excellence, London. http://.www.scie.org.uk. (accessed 10<sup>th</sup> May 2016)

Pekkarinen, S., Hennala, L., Harmaakorpi, V., Tura, T., (2011). Clashes as potential for innovation in public service sector reform. *International Journal of Public Sector Management*. **Vol.24**, No.6, 507-532.

Pesut, B., (2013). If time is relative ..... then why don't I have enough of it? *Nursing Philosophy.* **Vol.14,** 2, 75-77.

Pines, E., Rauschhuber, M., Norgan, G., Cook, J., Canchola, L., Richardson, C., Jones, M., (2012). Stress resiliency, psychological empowerment and conflict management styles among baccalaureate nursing students. *Journal of Advanced Nursing*. **68,** 7, 1482-1493. <a href="https://ncbi.nlm.nih.gov">https://ncbi.nlm.nih.gov</a>. (accessed 18<sup>th</sup> August, 2017)

Pool, I., Poell, R., Berings, M., Cate, O., (2015). Strategies for continuing professional development among younger, middle-aged, and older nurses: A biographical approach. *International Journal of Nursing Studies*. 52, 939-950.

Porter, S. (2007). Validity, trustworthiness and rigour: reasserting realism in qualitative research, *Journal of Advanced Nursing*, 60 (1): 79-86.

Price, D., Buch, C., Hagerty, B., (2015). Measuring Confidence in Nursing Graduates within the Framework of the AACN Essentials. *Nursing Education Perspective*. Vol **36** (2) March/April: 116-117.

Prime Minister's Commission for Nursing and Midwifery in England, (2010) *Front Line Care; The Future of Nursing and Midwifery in England*, <a href="http://tinyurl.com/yjtcjrz">http://tinyurl.com/yjtcjrz</a> (accessed May 9<sup>th</sup> 2016).

Public Health England, (2008). *Public Health Observatory Report* www.https://data.gov.uk (accessed 20<sup>th</sup> May 2015).

Queen's Nursing Institute, (2010). *The Fund for Innovation & Leadership Impact Report 2008-2010.* QNI, London.

Queen's Nursing Institute, (2012). *The Fund for Innovation & Leadership Impact Report 2010-2012*. QNI, London.

Queen's Nursing Institute, (2014). *The Fund for Innovation & Leadership Impact Report 2013-2014.* QNI, London.

Raelin, A., (2010). *The Leaderful Fieldbook: Strategies and Activities for Developing Leadership in Everyone*. Nicholas Brealey Publishing, Boston.

Rao, M., (2014). Inequality rife among black and minority ethnic staff in the NHS, *The Guardian*, August 2014.

Redfern, S., Harris, R., (2011). The impact of nursing innovations in the context of governance and incentives. Journal of Research in Nursing, **16**, 3, 274-294.

Richards, L., (2009). *Handling Qualitative Data A Practical Guide*, 2<sup>nd</sup> Edition, Sage, London.

Riessman, C. K., (2007). *Narrative methods for the human sciences*. Thousand Oaks, CA: Sage.

Ritchie, J., (2001). Case series research: a case for using qualitative research in assembling evidence for evidence-based practice, *Physiotherapy Theory and Practice*. **Vol.17**: 27-135.

Ritchie, J., Lewis, J., (2009). *Qualitative Research Practice – A Guide for Social Science Students and Researchers*. Sage, London.

Ritchie, J., Lewis, J., McNaughton Nicholls, C., Ormston, R. (2013). *Qualitative Research Practice A Guide for Social Science Students and Researchers*, Sage, London.

Roberts, G., Machon, A., (2015). *Appreciative Healthcare Practice – a guide to compassionate person-centred care*. <a href="https://researchgate.net">https://researchgate.net</a> (accessed 20<sup>th</sup> August 2017).

Robinson, J., (2014). The 7 Traits of Successful Entrepreneurs, *Entrepreneur*, January 10, 2014.

Robson, C., (2004). Real World Research (2<sup>nd</sup> edition), Wiley, Sussex.

Robson, C., (2011). Real World Research (3rd edition) Sussex, Wiley.

Rogers, E., (1976). New Product Adoption and Diffusion. *Journal of Consumer Research.* **Vol.2**, 336-344.

Rogers, E., (1983). *Diffusion of Innovations:* Third Edition, Macmillan, New York, London.

Rogers, E., (1995). *Diffusion of Innovations* (4th edition), Free Press, New York.

Rogers, E., (1995). Forty Years of Diffusion of Innovations: Utility and Value in Public Health, in Haider & Krep, 2004, *Journal of Health Communications*, **Vol. 9**, 3-11.

Rogers, E., (2003). *Diffusion of Innovations (5<sup>th</sup> edition),* Free Press, New York.

Roller, M., Lavrakas, P., (2015). *Applied Qualitative Research Design: A Total Quality Framework Approach*, The Guildford Press, New York.

Romero, J., Varela, L., Ricoy, C., (2013). Diffusion of innovations in social interaction systems. An agent-based model for the introduction of new drugs in markets. *European Journal of Economics*, 14: 443-455.

Rosenblum, M., Rothberb, A., (1994).cited in Kenny G. (2012). An introduction to Moustakas's heuristic method. *Nurse Researcher:* **19**, 3: 6-11.

Ross, F., Redfern, S., Harris, R., Christian, S., (2011). The impact of nursing innovations in the context of governance and incentives. *Journal of Research in Nursing*, **16**, 3, 274-294.

Rossman, G., Rallis, S., (2012). Learning in the Field 3<sup>rd</sup> Edition, Sage, London.

Royal College of Psychiatry (2011). <a href="www.https://rcplh.ac.uk">www.https://rcplh.ac.uk</a> (accessed 20<sup>th</sup> August 2017).

Ryan, B., Gross, N., (1950). Acceptance and Diffusion of Hybrid Corn Seed in Two Iowa Communities, *Research Bulletin, Economics & Sociology Section.* www.http://lib.dr.lastate.edu ...No.372. (accessed 8<sup>th</sup> January 2015).

Ryan, F., Coughlan, M., Croin, P., (2007). Step-by-step guide to critiquing research. Part 2: qualitative research *British Journal of Nursing*, 2007, **Vol 16**, No.12: 738-744.

Sarasvathy, S., (2001). What makes entrepreneurs entrepreneurial? *University of Washington School of Business Administration*. 2-9 <u>www.prawfsblawg.blogs.com</u> (accessed 1<sup>st</sup> September 2017).

Sarhan, F., (2014). An Evaluation of Telehealth to support patients with long-term conditions (COPD and CHF) in Buckinghamshire. *Centre of Excellence for Telehealth and Assisted Living (CETAL) www.http://bucks.ac.uk* (accessed 15<sup>th</sup> May 2016).

Schleien, S., Miller, K. (2010). Diffusion of Innovation: A Roadmap for Inclusive Community Recreation Services. *Research & Practice for Persons with Severe Disabilities*. **Vol 35**, No.3-4: 93-101.

Seale, C., (2004). Qualitative Research Practice. Thousand Oaks, California.

Sedgwick, M., Spiers, J. (2009). The Use of Videoconferencing as a Medium for the Qualitative Interview, *International Journal of Qualitative Methods*, **Vol. 8**, No. 1, 1-11, <a href="https://ejournals.library.ualberta.ca/index.php/IJQM/index">https://ejournals.library.ualberta.ca/index.php/IJQM/index</a> (accessed 26<sup>th</sup> May 2016).

Seitz, S., (2015). 'Pixilated partnerships, overcoming obstacles in qualitative interviews via Skype: a research note'. *Qualitative Research*, early online publication, https://doi: 10.1177/1468794115577011.

Shamsuzzoha, B., Syed, Dadwal, V. Martin, G., (2013) Reverse innovation in global health systems: towards global innovation flow, *Globalisation and Health*, 2013, **9**, 36 http://www.globalisationandhealth.biomedcentral.com. (accessed 27<sup>th</sup> May, 2016).

Silverman, D. (2011) Doing Qualitative Research Third Edition, Sage, Los Angeles.

Simpson, D., (2009, Organizational Readiness for Stage-Based Dynamics of Innovation Implementation. *Research on Social Work Practice*. **Vol.19**, 5, 541-551.

Singh, R., Mathiassen, L., Stachura, M., Astapova, E., (2010). Sustainable Rural Telehealth Innovation: A Public Health Case Study. *Health Services Research*. 985-1083.

Sines, D., Saunders, M., Forbes-Burford, J., (2009) *Community Health Care Nursing* (4<sup>th</sup> ed) Wiley- Blackwell, Oxford.

Skinner, B., (2010). Web alert: enabling and stimulating innovation in the delivery of health care. *Knowledge share*, Royal Sussex County Hospital, Brighton.

Sloan, A.,(1875) cited in Miller, D., Le-Breton-Miller, I, (2017). Sources of Entrepreneurial Courage and Imagination: Three Perspectives, Three Contexts. *Entrepreneurship Theory and Practice.* **Vol 41**, 5, 667-675.

Slovic, P., (2000) cited in Nicholson, N. Soane, E. Fenton-O-Creevy, Willman, P. (2005) Personality and domain-specific risk taking, *Journal of Risk Research*, **8**,(2): 157-156.

Slovic, P., (2000). The Perception of Risk, Earthscan, London.

Smit, E., De Vries, H., Hoving, C., (2013). Determinants of practice nurses' intention to implement a new smoking cessation intervention: the importance of attitude and innovation characteristics. *Journal of Advanced Nursing*, **Vol.69**, No.12: 2665 – 2673.

Sobh, R., Perry, C., (2006). Research design and data analysis in realism research, *European Journal of Marketing*, **Vol. 40**, No.11/12: 1194-1209.

Stanley, H., (2003). The journey to becoming a graduate nurse: a study of the lived experience of part-time post-registration students. *Nurse Education in Practice.* **3**, 62-71.

Stefancyk, A., Hancock, B., Meadows, M., (2013). The Nurse Manager Change Agent, Change Coach? *Nurse Administration Quarterly*, **Vol. 37**, No. 1: 13-17.

Strating, M., Nieboer, A., (2010). Norms for creativity and implementation in healthcare teams: testing the group innovation inventory. *International Journal for Quality in Health Care*, **Vol.22**, No.4, 275-282.

Suhonen, M., Paasivaara, L. (2011). Nurse Manager's challenges in project management *Journal of Nursing Management*, 19, 1028-1036.

Sullivan, D., Ford, C., (2013). How Entrepreneurs Use Networks to Address Changing Resource Requirements During Early Venture Development. *Entrepreneurship Theory and Practice.* **Vol.38,** 3, 551-574.

Sullivan, J. (2012). Skype: An Appropriate Method of Data Collection for Qualitative Interviews? *The Hilltop Review*, **Vol.6**; Iss.1, Article 10.

Sung, S., Choi, J., (2009). Do Big Five Personality Factors Affect Individual Creativity? The Moderating Role of Extrinsic Motivation. *Social Behaviour and Personality*, **37.** No.7, 941-956.

Social Innovation, (2012). <u>www.socialenterprise.org.uk</u> (accessed 10<sup>th</sup> January 2015).

Takasa, M., (2013). The relationship between the levels of nurses' competence and the length of their clinical experience: a tentative model for nursing competence development, *Journal of Clinical Nursing.* **22** (9–10): 1400-1410.

Teece. (2007). cited in Miller, D., Le-Breton-Miller, I, (2017). Sources of Entrepreneurial Courage and Imagination: Three Perspectives, Three Contexts. *Entrepreneurship Theory and Practice.* **Vol 41,** 5, 667-675.

TEPSIE, (2014). Social Innovation Theory and Research: A Guide for Researchers, 'The Theoretical, empirical and policy foundations for building social innovation in Europe', European Commission -7<sup>th</sup> Framework Programme, Brussels: European Commission, DG Research.

Tidd, J., Bessant, J. (2013). *Managing Innovation: Integrating Technological, Market and Organisational Change (5<sup>th</sup> edition),* Wiley, London.

Tierney, P., Farmer, S., (2011). Creative Self-Efficacy Development and Creative Performance OverTime. *Journal of Applied Psychology*, **Vol.96**, 2, 277-293.

Trueland, J., (2014). All in a day's work. Nursing Standard. Vol. 28, No. 26, 24-25.

The Guardian, (2012). Africa innovations: 15 ideas helping to transform a continent, *World News, The Guardian,* 26<sup>th</sup> August, 2012, <u>www.theguardian.com-World-Africa</u>, (accessed 5<sup>th</sup> May 2016).

The Kings Fund, (2012). What can hospital staff do to improve the quality of care for patient? The Kings Fund, London.

The Kings Fund, (2014). *Transforming community services: learning from previous mistakes.* www.https//kings.fund.org.uk (accessed 17<sup>th</sup> August, 2017).

The Kings Fund, (2016) *Deficits in the NHS 2016.* www.https//kings.fund.org.uk (accessed 17<sup>th</sup> August 2017).

The Young Foundation, (2012). Social Innovation Overview: A deliverable of the project: "The theoretical, empirical and policy foundations for building social

*innovation in Europe*" (TEPSIE), European Commission – 7<sup>th</sup> Framework Programme, Brussels: European Commission, DG Research.

Torri, M., Laplante, J. (2009). Enhancing innovation between scientific and indigenous knowledge: pioneer NGOs in India. *Journal of Ethnobiology and Ethnomedicine*. 5, 29. <a href="http://ethnobiomed.com/content.5/1/29">http://ethnobiomed.com/content.5/1/29</a> (accessed 10th April 2017).

Turner, S., Kaylor, S., (2015). Neuman Systems Model as a Conceptual Framework for Nurse Resilience. *Nursing Science Quarterly*. **Vol.28**, 3, 213-217.

Tusaie, K., Dyer, J., (2004). Resilience: a historical review of the construct. *Continuing Education* https://researchgate.net (accessed 10<sup>th</sup> August, 2017).

UNISON, (2015). *Red Alert Unsafe Staffing Levels Rising*, Unison the public service union, London.

Unison (2015). *National Joint Council Pay Claim 2016-2017* <a href="http://.www/.unison.org.uk">http://.www/.unison.org.uk</a> (accessed 16th May 2016).

Vedel, I., Ghadi, V., De Stampa, M., Routelous, C., Bergman, H., Ankri, J., Lapointe, L., (2013). Diffusion of a collaborative care model in primary care: a longitudinal qualitative study. *BioMed Central Family Practice*, 14:3. http://www.biomedcentral.com/1471 (accessed June 2<sup>nd</sup> 2016).

Vichensky, T., Beanlands, H., (2004). Qualitative Research, *Nephrology Nursing Journal*, **31** (2): 234- 238.

Vygotsky, L., (1962). *Thought and Language.* Cambridge, MA: MIT Press.

Walsh, T., (2010). *The Solution-Focused Helper: Ethics and Practice in Health and Social Care.* Open University Press, Berkshire.

Walsh, T., Barr, P., Thompson, R., Ozanne, E., O'Neill, C., Ellwyn G., (2014). Undetermined impact of patient decision support interventions on healthcare costs and savings: systematic review. British Medical Journal: 346 doi: <a href="https://doi.org/10.1136/bmj.g188">https://doi.org/10.1136/bmj.g188</a> (accessed 4th April 2016).

Weglicki, R., Reynolds, J., Rivers, P. (2015). Continuing professional development needs of nursing and allied health professionals with responsibility for prescribing. *Nurse Education Today.* **Vol.35** (1), 227-231.

Welch, R., Mooney, J., Shindul, J. (2008). The university/community partnership: transdisciplinary course development. *Journal of Interdisciplinary Care*, **22**, 5, 461-474.

Weller, S., (2015). The potentials and pitfalls of using Skype for qualitative (longitudinal) interviews. *National Centre for Research Methods Working Paper 4/15.* 

West, M., Eckert, R., Armit, K., Loewenthal, L. West, T., Lee, A. (2015). *Leadership in Health Care: A Summary of the Evidence Base*, The Kings Fund, London.

World Health Organisation (2012). *The World Health Statistics 2012,* www.https://who.int (accessed 20<sup>th</sup> August, 2014).

Ziyae, B., Mobaraki, M., Saeediyoun, M., (2015). The Effect of Psychological Capital on Innovation in Information Technology. *Journal of Global Entrepreneurship Research.* **5,** 8, 1-20.

Appendix 1 Details of Projects

Overview of projects research participants initiated 1 - 17 were deemed to be successful projects, 18 - 26 did not succeed as planned.

	Aim of Project	Project	Financial	Anticipated cost	Outcomes	Progress	
			Expenditure	savings			
1	To reduce	Heart Failure Self-	£7,550 – Develop	£14,270 over a 6	Reduction of	First 6 week period	
	emergency	Management	self - assessment	week period for 36	hospital admissions	successful additional	
	admissions for	Programme	toolkit and	patients with heart	and bed days.	costs submitted to	
	patients diagnosed		additional hours	failure	Patients could	carry out an	
	with heart failure		for band 6 nurse		monitor their quality	evaluation of	
					of life and take	programme	

					ownership of their		
2	Paduas bessital	Falla Provention	C20,000 +>	£115 000 souther !f	actions Reduction of	6 month nilat	
2	Reduce hospital admissions due to falls	Falls Prevention Service	£20,000 – to employ a band 4 falls prevention champion and minimal marketing costs	£115,000 saving if 15 referrals per week are prevented over six months.	Reduction of hospital admissions and community team referrals	6 month pilot successful – plans to employ an additional band 4	
3	To provide a cheaper alternative to hospital appointment follow-up	Nurse-led telephone follow up for men with prostate cancer	£35,000 – to employ a part time band 6 community nurse & admin costs	£82,600 if 800 patients are telephoned per year (each hospital visit costs £147.00)	Improved patient satisfaction - they are able to fit telephone consultations around their work/personal life	Successful – after 3 years a full evaluation of this service to be carried out	
4	To provide 3 x weekly clinics for mobile patients with venous leg ulcers	Nurse-led venous leg ulcer clinic in General Practice	£35,000 – 1 community nurse band 6 and one band 4 for 15 hours over 6months	£102,500 saving over a 6 month period estimated on 2 visits per patient per week	Improved patient & staff satisfaction due to faster healing rates & reduced social isolation	Successful – have now employed a community nurse lead for the clinics Commission services to a wider audience	
5	To develop a pathway to identify Learning Disability patients at high risk and for carers to track early warning signs in order to access health care sooner	Reducing Avoidable Hospital Admissions Pathway through inter-professional working	£12,000 to develop pathway	£19,000 per year	Prevention of hospital admissions, reducing anxiety in this client group	Successful – application to employ a discreet fixed term 'Reducing Hospital Admissions" nursing role	
6	Reduce the inequality of leg ulcer care across a geographical region	Silver Standard Leg Ulcer Service	£15,000 – 4hr x 52 weeks for band 6 time + £7,800 dressings	£375,000 per year by reducing chronic leg wounds by 50%	Reduction of chronic leg ulcers	Successful after 1 year.	
7	geographical region  To provide training and tools required to facilitate quality end of life care for adults with learning disabilities  End of Life Care awareness training for care home staff and learning disability awareness training for Macmillan staff		£10,500 – training, printing, Band 5 x 1 day per week for 20 weeks	£68,000 per year by reducing current hospital admission by half	Raised profile of end of life care for people with learning disabilities and 50% of staff have had training, reducing hospital admissions and improved end of life care for patients and families	This project is now on the Trust annual agenda and will continue for the foreseeable future	
8	To improve the quality of nursing care to all patients requiring care from District Nursing teams, through the use of a standardised caseload management tool.	Development of a pro-active caseload management tool	£11,550. For 3 month pilot Band 6 project lead, administrative support, Training & Development materials	£90,000 per District Nursing team over a 3 month period by preventing 2 hospital admissions per month	District Nurse time is freed up to visit more patients by reducing caseload	The District Nurse profile within Trust has been raised and she is now taking on more projects. This caseload tool is now used to inform commissioning, workforce planning	
9	To keep patients with long term conditions at home through extending out of hours community health services	Virtual Ward out of hours service	£15,000 to pilot the programme for 3 months – £7,200 Band 5 community nurse and £3,200 Band 3 health care assistant	Estimated saving of £48,000 for 1 night stay for 10 patients per month over 3 months. Reduced hospital admissions, particularly for patients with long term conditions.	Access to service 24 hours per day resulting in improved quality of life & continuity of care.	Virtual Ward has been commissioned and is ongoing.	
10	Financial saving from a reduction in oral nutritional supplement prescribing	Reduction of inappropriate prescriptions for oral nutritional supplements	£51,000 to employ a prescribing support dietitian and part time dietetic assistant for an initial 12 month period	Anticipated to save £150,000 by reducing prescription for oral nutritional supplements.	Improve the quality of care delivered to patients by ensuring that malnutrition is identified and managed appropriately.	Savings have been made and a dietetic assistant is still employed	

11	Raise awareness of NF1	To improve the lives of NF1 patients and families	£20,000 for Band 7 nurse and administrative support	Reduction of hospital emergency admissions due to additional available expert support	Improved quality of life for patients and families of NF1 sufferers	NF1 support network continues	
12	Prevent long waiting lists for patients requiring lymphedema provision	Develop a lymphedema service for patients within a general practice region	Set up social enterprise with colleague	Creating a value for local people whilst utilising expert skills	On- site service provision reducing time spent on hospital outpatient clinics	Ongoing successful lymphedema service provision for local people	
13	As above		Set up social enterprise				
14	provision for communities in a rural region of services, develop a training schedule for community & primary care nurses		Joint training sessions for District Nurses and General Practice Nurses	Prevent duplication of service provision	Seamless care provided for patients, promoting continuity of care and increased patient satisfaction	Work is continuing with Clinical Commissioning Groups & Trust	
15	Promote infection control in the workplace and reduce hospital admissions	I in the programme for care home staff		To enhance care home staff knowledge of risks associated with infection related conditions	A 70% reduction of infection related admissions to hospital	Posters are displayed in care homes outlining key risks associated with infection related conditions	
16	To assess vulnerable patients in the community predicting those at risk of adverse outcomes	Develop a prediction tool to assess vulnerable at risk patients	Time to introduce this tool to all District Nurses at monthly meetings	Individualised care pathways benefitting individual patients	Carers, health professional & patients benefitting as a result of early identification of risk	Steering group set up to support this project	
17	Improve the physical activity levels of females  Develop a community cardiac rehabilitation women's walking group		Plan cardiac education into District Nurse & General Practice Nurse visits & clinics for female patients	Improved well- being of female participants	The women's quality of life improved and blood pressure was maintained at optimal level	The project is ongoing	
18	To encourage young people to learn about and test their comprehension of a broad range of contraception and sexual health topics.  Games/Quizzes to decipher the knowledge and understanding of public health issues related to sexual health practices		£36,000 to develop quiz and publish findings in the format of a book on sexual health	Anticipated saving of £330,000 per year through reduction in unplanned pregnancies and reduction in sexually transmitted diseases	No improved health outcomes to report	Was unable to seek funding for the project	
19	To reduce the number of patients being admitted inappropriately from between 50-70 per day down to 20-25 per day	Accident & Emergency Health Visitor Service	£46,750 – whole time equivalent full time HV band 6 + administrative costs	£144,000 anticipated saving of 100 patients per month	Reduction of 100 admissions per month	Pilot service explored but not implemented - HV since left and set up a Social Enterprise	
20	To reduce the actual and potential harm caused by alcohol misuse to individuals and communities  Produce a game linking to the Public Health work plan for Healthy Ageing professio facilitating game and producing games habeen fact		£1500 – time to develop game, health professionals time facilitating the game and producing ongoing games had not been factored into the costing	Prevention of alcohol related admission of older people to A&E had not been calculated	Those that used the game did say that it raised their awareness to the dangers of excess alcohol, but no statistical evidence available	This project did not proceed as future planning had not been accounted for in the original costing	

21	To increase capacity and delivery, cut agency spend by 50% within Health Visiting teams	Development of a transformational leadership programme that incorporates coaching, supervision, goal setting & motivation mapping	£10,000 to deliver over a 9 month period	£182,000 approximately over 9 months	Greater autonomy for Health Visitors, increased morale and increased job satisfaction and productivity within Health Visiting teams, reduced agency staff	Unable to obtain funding from Trust or various charities to support this transformational programme. The Health Visitor has since left and has set up her own company
22	To provide an equitable maternity service for all pregnant women in a rural area and to enhance staff skills and knowledge to provide continuity of care	Implementation of local scanning procedures and purchase a birthing pool	£35,000 Train 2 midwife sonographers, purchase relevant stationery. Purchase & install birthing pool Staff training to lead unit and backfill for staff	Improved choice for pregnant mothers and an increase in local maternity services	Anticipated outcomes were to reduce travelling time for expectant mothers, care closer to home and provide greater choice of method of delivery	Unable to obtain funding to purchase equipment. Midwife has now left this post and taken another position where she has more autonomy
23	Reduce unplanned admissions for patients suffering with multiple sclerosis	dmissions for atients suffering with multiple Admissions		Anticipated reduction of 100 admissions to hospital per year saving approx. £146,000	Increased patient choice, care closer to home	Unable to supply funding beyond pilot stage. Participant has now left this employment
24	Improve mental health services for outpatient clinics	Develop a weekly mental health drop in clinic	Band 6 nurse to facilitate sessions for 3 hours per week	Reduction of A&E visits	Improved quality of life for patients through ongoing care in the community	Trust provider changed so project not taken forward as business case not produced in detail to support this
25	Prevent social exclusion	Arrange a bus befriending service for the older socially excluded	Bus hire and hire of venue for group meetings	Reduction of hospital admissions caused by depression and mental health conditions	Improved well-being of patients	Interest in project from the mental health services, but no funding to support bus service
26	Promote family support for patients with mental health conditions	Reduce hospital admissions	Provide training for mental health practitioners and support workers	Anticipated cost saving would be reduced hospital admissions and inappropriate emergency call outs	Ongoing family support, continuity of care	This project did not continue due to changes of commissioned services

### Appendix 2 Information Sheet for Participants of the Study

<u>Title of Study:</u> A qualitative study of UK community health practitioners' perceptions of their personal and professional development after training for and practicing as health and social innovators

I would like to invite you to participate in this research study. You are not obliged to participate and if you choose not to take part you will not be disadvantaged. Before you decide to take part, it is important for you to understand why the research is being carried out and what your participation will involve.

What is the project about? Innovation is not a new concept and this study aims to determine the expectations of students undertaking a programme where this is actively promoted and encouraged. A number of students will be interviewed before and after completing the programme. It will also determine the impact of such a programme on students and their practice, exploring whether such a programme facilitates theory to practice knowledge transfer and if this impacts on the quality of care given to service users by having social innovators as part of the community health workforce. This research will also be looking at community practitioners who have not had any formal training on innovation but who have been successful in setting up an innovative project in their work place.

**Selection Process**: You have been invited to participate as you have expressed an interest in innovation either through partaking in a programme on social innovation or by being an innovative community practitioner. For those of you who have attended a social innovation educational programme I would like to find out your views about the programme and how this has impacted on your clinical practice. For other participants I would like to find out what have been the drivers to enable those of you who have not had formal training on innovation but have been successful in implementing an innovative project in your clinical practice.

**Who is involved in the research study**? The study will be led by a member of staff from Buckinghamshire New University who is undertaking a Professional Doctorate at the University of Westminster. The staff member has a special interest in utilising innovation to enhance clinical practice

**Do I have to take part?** No, participation is entirely voluntary. If you decide to take part you will be given this information sheet to keep and will be asked to sign a consent form agreeing to take part. Should you decide not to partake in the study you do not have to give a reason. If you decide to withdraw during the study, any of the information you have provided can also be withdrawn if you request this.

What is involved? If you agree to take part you will be amongst other participants who have agreed to take part in this study. I will arrange to meet with you at an agreed time prior to commencement of your period of study. If you are participating in focus group interviews it will last between 1-2 hours and there will be a number of key questions that I would like to ask you. The interview will be audio taped, subject to your permission to allow all of your responses to be captured accurately. If you are having a telephone interview it is anticipated that it will last between 10 – 30mins. The telephone interview will also be recorded for accuracy. Individual face –to- face individual interviews will last no longer than 60 minutes.

**What happens to the information?** All of the information is confidential and your information will be coded not using names so it will be totally anonymous. The information will be downloaded onto the researcher's personal computer, which is password protected and the data will only be accessible to the researcher.

What will happen to the results of the project? The results will be collated and presented as part of the researcher's final thesis. Some data will be used in publications and conference presentations, but these will be anonymous, unless permission to name individual projects has been given by any of the participants. A copy of the final report can be sent to you if you request this.

Thank you for taking the time to read this information sheet.

Appendix 3	Consent Form for Participants
Please state yes or no	
I have read and understood the p	roject information sheet
I have been given the opportunity	y to ask questions about the project

I agree to take part in the project. Taking part in the project will include being interviewed on an individual basis of part of a focus group and will involve the focus group discussion being audio taped	r as
I understand that my taking part is voluntary; I can withdraw from the study at any time and I will not be asked questions about why I no longer want to take part	
I understand my personal details such as phone number, email or address will not be revealed to people outside of this project.	ıf
I understand that my words may be quoted in publications, reports, web pages, and other research outputs but my name will not be used	/
I agree for the data to be held by the Key Researcher for the period of time that the study is being carried out and understand that it will be destroyed once the final thesis has been written up	
I understand that no other researcher will have access to the data.	
On this basis I am happy to participate in the research study looking at the impact of innovation on the quality and effectiveness of service delivery in a clinical community setting.	I
Name of Participant	
Name of ResearcherAgnes Fanning SignatureAgnes Fanning Date	
If you have any queries or concerns, please contact: [	
Agnes Fanning	
Buckinghamshire New University,	
Queen Alexandra Road,	
High Wycombe,	
HP11 2JZ	
Agnes.fanning@bucks.ac.uk	
01494 522141 ext. 5765	
One copy to be kept by the participant, one to be kept by the researcher	
Adapted from the UKDA Model consent form	
Appendix 4 Questions for participants	
Question 1: Why did you enrol onto the programme?	
Question 2: Did you have any expectations from the programme you studied?	

Question 3: What skill set did you bring to the programme?

Question 4: Having been on the programme has it changed your practice?

Question 5: Do you feel the programme has enhanced your practice?

Question 6: Has your organisation noticed any change in the way you work?

Question 7: If you learnt one key thing from the programme you attended what would it be?

Question 8: Looking back on Day 1 do you feel you are working in a different way than before the programme?

Question 9: What made you want to look differently at how you, or others in your organisation were delivering care? Was there a problem in the way they were working?

Question 10: What impact did that level/style of care have on the client group?

Appendix 5 Additional Questions

Question 11: Are there any areas of disappointment for you in how your project is progressing/has been able to develop?

Question 12: What would you do differently (if anything) if you could start your project again?

Question 13: What advice would you give to anyone who is planning to introduce innovation into their organisation or profession?

Question 14: Thinking of your current project, how has working in a different way changed the service provided? If it has, In particular, how are these three key areas impacted:

- The impact on the quality of care delivered to your client group?
- Any difference in the service delivery process?
- Evidence of value for money for your organisation
- Any other observations you may have about the impact

## Appendix 6 Consent Form for Transcriber

One copy to be kept by the transcriber, one to be kept by the researcher

Please state Agree or Do Not Agree
I have read and understood the project information sheet
I have been given the opportunity to ask questions about the project
I agree to take part in the project as a transcriber for the project. The project will include interviews on an individual basis or as part of a focus group and will involve the focus group discussion being audio taped
Transcripts will be sent to the transcriber at intervals agreed between the transcriber and the researcher
All information remains confidential and must not be shared with anyone other than the researcher
I agree for the data to be held by the Key Researcher for the period of time that the study is being carried out and understand that it will be destroyed once the final thesis has been written up
I agree to destroy all transcriptions once they have been sent to the researcher and the content has been confirmed by the participant and the researcher
I understand that no other researcher will have access to the data
Name of Transcriber Date
If you have any queries or concerns, please contact: [
Agnes Fanning
Buckinghamshire New University,
Queen Alexandra Road,
High Wycombe,
HP11 2JZ
Agnes.fanning@bucks.ac.uk
01494 522141 ext. 5765

#### Appendix 7 Ethics Approval from Bucks New University



22<sup>nd</sup> April, 2013

Mrs Agnes Fanning

Primary Care & Public Health,

Buckinghamshire New University,

Queen Alexandra Road,

High Wycombe, HP11 2JZ

Dear Agnes,

I am writing to confirm that Ethical approval was granted by the Society & Health Ethics committee of Buckinghamshire New University on 26<sup>th</sup> February 2013 for your project titled:

"A qualitative investigation of the impact of a Social Innovation educational programme on Community Health Professionals and their client group"

I hope that your research project goes well.

Yours sincerely,

Mel Nakisa

Dr. M. Nakisa

Secretary to the Ethics Committee

Research Unit

Academic Quality Directorate

Appendix 8 Ethics Approval from University of Westminster

# UNIVERSITY OF WESTMINSTER#

Ethics Committee-database										
Comments				Date Closed Final Outo		come				
Approved by Chair's Action	on			29-Jul-1	9-Jul-13 Approved					
Application No	Date Received	Surname	Forename	Level		School/Faculty		Supervisor (Where applicable)		
12-13-28		Fanning	.6	PhD Student	Life So	Sciences P		Patricia Maitland		
Project Title		Course (student)	Date of Meeting	Outcome	Discu	ssion	Conditions	Amendments Deadline	Comments	Date Closed
A qualitative investigation of the impact of a social innovation programme on community health professionals and their client group		Mphil/Phd	26-Jun-13	Resubmit			Yes		Approved by Chair's Action	29-Jul- 13